Perceptions of Early Childhood Teachers About How Trauma Impacts Young Children in the Classroom and the Adequacy of Their Preparation to Teach Young Children Who Have Experienced Trauma

> Michele L. Hodapp B.A., University of Wisconsin-La Crosse, 1974 M.S., University of Kansas, 1990

Submitted to the Graduate Department and Faculty of the School of Education of Baker University in partial fulfillment of the requirements for the degree of Doctor of Education in Leadership in Higher Education

Tes Mehríng

Tes Mehring, Ph.D. Major Advisor

Tracy Cooper

Tracy Cooper, Ph.D.

Harold Frye

Harold Frye, Ed.D.

Date Defended: August 30, 2022

Copyright 2022 by Michele L. Hodapp

## Abstract

Nearly every teacher will have contact with a child or family who has been impacted by trauma, and research has indicated that the intervention and support a young child receives can make a critical difference (Erdman, Colker, & Winter, 2020). Trauma in young children is a public health crisis (Erdman et al., 2020; Van der Kolk, 2014). While there is compelling research about the development of young children and the quality of early childhood education experiences, few studies have focused on early childhood care and education (ECCE) teachers' perceptions of how trauma impacts young children in center-based programs, the information and skills needed to meet the challenges of young children who have experienced trauma, professional development opportunities that are available to provide the knowledge and skills needed to work with young children who have experienced trauma, or the challenges related to accessing professional development. Ten ECCE teachers working directly with young children who may have experienced trauma participated in the study. All participants had two or more years of experience and were employed in a center-based ECCE classroom in Johnson County, Kansas, from January 2020 through July 2021. Four themes were identified from the data analysis: (1) teachers' perceptions of trauma in early care and education; (2) teachers' professional development needs working with young children who may have experienced trauma; (3) the resources and professional development opportunities available to ECCE teachers; and (4) the challenges of working with young children who may have experienced trauma. The need for collaboration among ECCE professionals and infant-toddler mental health practitioners and organizations to provide

ii

relevant professional development for ECCE teachers and providers who work with young children and their families who have experienced trauma was identified in the current study.

# Dedication

This dissertation is dedicated to my grandson, Bear, and all young children who have experienced trauma and who are triggered without warning and struggle with caregivers and teachers who have little or no awareness of the impact of trauma or have effective strategies to support them in self-regulation and healing. You are my inspiration.

These efforts are in remembrance of my late husband, William K. Hodapp, my mom, Virginia Northrop Ryan, and my dad, Charles "Chuck" Ryan. Each was generous with their time and talents and willing to share them with children of all ages. In addition, each demonstrated a love of learning and valued life-long education. I was aware of their presence and have had them in my thoughts and heart throughout this doctoral journey.

#### Acknowledgments

I am blessed as I have danced syncopated rhythms between life's fragile moments and the pursuit of a dream. So many people, family, friends, and colleagues have supported me on this doctoral journey. I am grateful and amazed that you have hung in there with me through these many years.

- My sons and daughters-by-love, Ryan, Patrick and Lindsay, and Tommy and Alyssa. You always had faith in me and believed I could "get'er" done. I appreciate your gifts as teachers by vocation and avocation. I take pride in your abilities as parents and your abundant love for your children by birth and children by choice. Joy fills my heart.
- My dear friend and "Baker Buddy" Dr. Pamala Asher. We have shared the trials, tribulations, and celebrations since the first weeks in class. We persevered and demonstrated grit, never giving up through all the personal and professional detours we encountered. Thank you.
- My "therapon" friends, Nancy and Dana, and my scouting family, the Wood Badge staffs of C5-307 20/1 and C5-307-17-2, the "Bobwhite Chicks," the Eagle Moms, the scouts and leaders of troop 6282. You have often inquired, "How's it going?" Cheered me on with each small achievement and offered a note or a nod of encouragement or a smile that kept me going. It has meant a lot.
- Many faculty, colleagues, co-workers, and students in education. Dr. Harold Frye presented this opportunity through a casual conversation and a business card, and then the cosmic tumblers fell into place. Thank you for continuing to connect and serve on my committee. Dr. Tes Mehring, my advisor and member of my

v

committee, thank you for being a consummate cheerleader and reminding me to take "one bite of the elephant" at a time. Dr. Tracy Cooper, a member of my committee. Thank you for challenging me to digest information from experts and speak in my words. Dr. Peg Waterman, research analyst. Thank you for your guidance through methods and data analysis.

- Dr. Jonathon Long, formerly Kansas City Community College, and Dr. Ryan Meador, Metropolitan Community Colleges, for providing Directed Field Experiences sites, and Dr. Fred Grogan, Baker University. Thank you all for your willingness to serve as mentors.
- The faculty and students at Benedictine College in Atchison, KS. They permitted me to adopt them as my school of higher education to complete early assignments and projects through coursework.
- Co-workers, colleagues, and friends in early care and education programs and organizations across the Kansas metropolitan area. I share your struggles in working with young children. I appreciate your time and efforts in providing the best experiences you can with the knowledge and skills you have to offer children.

vi

Abstract ii
Dedication iv
Acknowledgementsv
Table of Contents
Chapter 1: Introduction1
Background5
Statement of the Problem10
Purpose of the Study10
Significance of the Study11
Delimitations12
Assumptions12
Research Questions
Definition of Terms13
Organization of the Study16
Chapter 2: Review of the Literature
Definition of ECCE19
History and Development of ECCE
Description of ECCE Workforce and Programs
Definition of High Quality ECCE
ECCE Teacher Professional Development
Young Children's Social and Emotional Development
Trauma Experiences in Young Children40

# **Table of Contents**

History and Development of Mental Health Services	42
Mental Health Services for Young Children	45
Summary	51
Chapter 3: Methods	52
Research Design	52
Setting	53
Sampling Procedures	54
Instrument	54
Data Collection Procedures	57
Data Analysis and Synthesis	61
Reliability and Trustworthiness	65
Researcher's Role	67
Limitations	69
Summary	70
Chapter 4: Results	71
Descriptive Demographics and Participant Background	71
Teachers' Perceptions About Trauma in Early Care and Education	73
Teachers' definition of trauma	74
Behaviors of children who have experienced trauma	75
Responses of other children to a child who has experienced trauma	a79
Teachers' responses in the classroom to the behavior of a child wh	io has
experienced trauma	82

Professional D	evelopment Needs of ECCE Teachers	86
Inform	ation needed by ECCE teachers	87
Skills r	needed by ECCE teachers	
Attribu	tes desired for ECCE teachers	92
Resources and	Professional Development Available to ECCE Teacher	rs95
Resour	ces needed by ECCE teachers	95
Profess	ional development available to ECCE teachers	96
Networ	ks and organizations accessed by ECCE teachers	
Challenges Wo	orking with Young Children Who Have Experienced Tr	auma102
Lack of	f awareness of trauma	
Needs	of children who may have experienced trauma	
Lack of	f preparedness of ECCE teachers	
Lack of	f relevant raining opportunities for ECC Teachers	
Lack of	f research regarding young children who may have exp	erienced
1	trauma	113
Attitud	es toward ECCE and trauma	114
Summary		115
Chapter 5: Interpretati	on and Recommendations	116
Study Summar	у	116
Overvi	ew of the problem	116
Purpos	e statement and research questions	117
Review	v of the methodology	118
Major	findings	120

Findings Related to the Literature	124
Conclusions	129
Implications for action	130
Recommendations for future research	132
Concluding remarks	134
References	137
Appendices	159
Appendix A: K.A.R. 28-4-428a. Education and Training Requirements	160
Appendix B: Baker University IRB Approval	164
Appendix C: Facebook and Professional Messaging Board Posting	166
Appendix D: Invitation to Participate	168
Appendix E: Confirmation to Participate	170
Appendix F: Consent Form	172

#### Chapter 1

# Introduction

Parents, caregivers, and teachers desire the best early childhood experiences for young children. Laughlin (2013) indicated that in 2011, over 20.4 million children under five years of age participated in early care and educational services. The workforce providing early care and education to our communities' youngest members is as diverse as the families they serve (Nicholson, Perez, & Kurtz, 2019; Rhodes & Huston, 2012). According to Concordia University St. Paul (2019), scores of women and men have aspirations to work in the education field. However, it demands an exceptional caliber of educator to work with young children during their first five years of life. These teachers are charged with fostering fundamental physical, social, behavioral, and cognitive growth and development.

The National Academies of Sciences, Engineering, and Medicine (2018) noted that well qualified ECCE staff are crucial in the delivery of exceptional ECCE services. Researchers have indicated that care providers' and teachers' education and training are critical factors in attaining and maintaining quality early childhood programs and practices. There is growing evidence that the caregivers and teachers who provide specific knowledge and skills to their position are the most highly qualified to deliver high-quality programs and resources to young children and their families (National Association for the Education of Young Children [NAEYC], 1993).

The Childcare Resource and Research Unit (2020) offered no single definition of quality in child care but identified several overall elements of child care critical to children's well-being. Several common human factors are associated with early care and instruction. These include effective leadership; knowledgeable, competent, trained and nurturing staff; small group size and low staff-child ratios; positive interactions and relationships with adults; inclusion of children with special needs; diversity; positive relationships; effective parent communication; and parent involvement (Childcare Resource and Research Unit, 2020; Early Childhood Education, 2020; NAEYC, 2020; Sim, Belanger, Stancel-Piqtak, & Karoly, 2019). In addition, aspects of the early learning environment and curriculum encompass a safe, clean, and well-maintained environment; include developmentally appropriate practices and learning activities; address areas of child development (large and fine motor, social-emotional, language, and cognitive skills); provide a balanced schedule including opportunities for active play, quiet activities, rest, outdoor time, nutritious meals and snacks, and self-care; and engage young children in a variety of activities provided individually in a large group and small group (Child Care Resource and Research Unit, 2020).

McLoyd, Aikens, and Burton (2006) contended that the quality of children's ECCE experiences matters for long-term intellectual development, socio-emotional wellbeing, and health. A young child's social-emotional development is dependent on a secure, trusting relationship with a designated caregiver (Zero to Three, 2005). The early childhood teacher spends a significant amount of time with a child during waking hours. The Learning Tree Institute (2021) in *The 2020 Kansas Child Care Market Analysis Final Repor*t noted children in full-time care are with family child care providers or in child care centers 50 -55 hours per week

Lieberman, Chu, Horn, and Harris (2011) argued that children under age 5 are at exceptionally high risk for exposure to potentially traumatic events due to their

dependence on parents and caregivers. The American Psychological Association Presidential Task Force on Posttraumatic Stress Disorder and Trauma in Children and Adolescents (2008) defined a traumatic event as "one that threatens injury, death, or the physical integrity of self or others and also causes horror, terror, or helplessness at the time it occurs" (p. 2). Traumatic events include a serious illness that requires significant medical intervention; the loss or life-threatening illness of a close family member; witnessing natural disasters; domestic or community violence or threats of terrorism; homelessness; and neglect or physical or sexual abuse (National Child Traumatic Stress Network Schools Committee, 2008). Unfortunately, exposure to trauma is a widespread experience for our youngest children (Ippen, Harris, Van Horn, & Lieberman, 2011). Numerous researchers, Briggs-Gowan, Ford, Fraleigh, McCarthy, and Carter (2010) and Shahinfar, Fox, and Leavitt (2000) indicated that a significant number of children in the United States have been vulnerable to trauma, and the likelihood of experiencing trauma is highest for children in the early childhood years. Statman-Weil (2015) noted that 26 % of children in the United States would have witnessed or experienced trauma before age 4.

Reactions to traumatic events can be displayed in various behaviors and vary from child to child. The National Child Traumatic Stress Network Schools Committee (2008) reported, "Many behaviors seen in children who have experienced trauma are nearly identical to those of children with developmental delays, attention deficit hyperactivity disorder (ADHD), and other mental health conditions" (p. 7). Because of the types of trauma young children may have experienced, ECCE teachers may observe increased separation anxiety, increased irritability and frustration, difficulty playing and working with adults and other children, tantrums, aggressive behaviors, lack of interest or ability to concentrate, detached, hyper-alert responses, or limited communication (American Psychological Association Presidential Task Force on Posttraumatic Stress Disorder and Trauma in Children and Adolescents, 2008; Buss, Warren, & Horton, 2015; Erdman, Colker, & Winter, 2020; Nicholson et al., 2019; Stone & Bray, 2015; Zero to Three, 2020).

Trauma experiences can have short-term and long-term impacts. According to the National Child Traumatic Stress Network Schools Committee (2008), trauma experienced in early childhood may result in learning or behavior challenges, poor school performance, and reoccurring physical and emotional distress. Researchers (Cleveland Clinic Pediatrics, 2020; Monnat & Chandler, 2015; Prevent Child Abuse America, 2020; Substance Abuse and Mental Health Services Administration, 2020) have indicated that long-term implications for multiple or repeated adverse childhood experiences may increase the risk of physical and emotional health problems in adult life. Childhood trauma may increase an individual's risk of anxiety, asthma, cancer, depression, diabetes, heart disease, obesity, post-traumatic stress disorder (PTSD), stroke, substance abuse, and suicide (Monnat & Chandler, 2015; Prevent Child Abuse America, 2020).

Van der Kolk (2005) stated, "Childhood trauma, including abuse and neglect, is probably the single most important public health challenge in the United States, a challenge that has the potential to be largely resolved by appropriate prevention and intervention" (p. 401). Teachers in early care and education programs can help a young child who has experienced a traumatic event. The National Child Traumatic Stress Network (2010) encouraged educators to report suspected abuse, work with parents and colleagues to address noticeable concerns, make a referral, and share *Trauma Facts for*  *Educators* (NCTSN, 2008) with other early childhood professionals. The American Psychological Association Presidential Task Force on Posttraumatic Stress Disorder and Trauma in Children and Adolescents (2008) noted,

Conveying information about common reactions to trauma can often be helpful to the child and the people around him or her, including parents, teachers, coaches, clergy, and community leaders. Knowing what to expect and what reactions are most common can relieve adults' worries that the child will not recover or will be damaged forever. If the individuals in a child's support system understand his or her behavior and distress as normal reactions to abnormal events, they can better support the child during the recovery period. (p. 4)

Nearly every educator will interact with a child or family who has been affected by trauma, and research has shown that the support and intervention a child receives when young can make a critical difference (Erdman et al., 2020).

## Background

Historically, child care was intended to promote parental employment, and early childhood education was designed to facilitate child development. However, in practice, this distinction is artificial. Many early education programs enable parents to work and many child care settings are designed to promote learning and development (Rhodes & Huston, 2012).

Young children develop social-emotional competencies in the context of the relationships that exist between a small group of consistent caregivers and the child. Caregivers may include parents, child care providers, foster parents, grandparents, and other family members (Cohen, Onunaku, Clothier, & Poppe, 2005). Rhodes and Huston (2012) reported that by the age of 5, most children had spent significant time in the care of adults other than their parents. Second only to the family, more than two million individuals in early care and education play a central role in children's development (Rhodes & Huston, 2012). The early years of childhood experiences build the foundation for physical and emotional well-being, intellectual development, and social competence (National Scientific Council on the Developing Child, 2007; Zero to Three, 2020).

The Institute of Medicine and National Research Council (2015) identified competencies for early childhood educators in its publication *Shared Knowledge and Competencies for Educators of Children Birth through Age 8.* The competencies related to young children's social-emotional development include:

- Ability to establish relationships and interactions with children that are consistent, nurturing, and use positive language.
- Ability to promote positive social development and self-regulation while mitigating challenging behaviors in ways that reflect an understanding of the multiple biological and environmental factors that affect behavior.
- Ability to recognize the effects of factors from outside the practice setting (e.g., poverty, trauma, parental depression, experience of violence in the home or community) that affect children's learning and development and to adjust practice to help children experiencing those effects. (p. 3)

From birth to age 5, young children are especially vulnerable to adverse effects of trauma due to rapid developmental growth, dependence on caregivers, and limited coping skills (Schonfeld, Demaria, & Kumar, 2020). Despite decades of statistical data, mental health professionals generally have limited knowledge about the impact of traumatic

events on younger children compared to older children and adolescents (De Young, Kenardy, & Cobham, 2011). "Children younger than five years of age typically experience rapid developmental changes that often are misinterpreted or not fully accounted for, which hinders proper diagnosis and intervention" (Buss et al., 2015, p. 266). Explanations for this disparity in knowledge include a historical resistance to the notion that early childhood mental health is critical. As a result, researchers have spent little time and effort investigating the effects of trauma exposure in early childhood. In addition, a widely held misconception has been that infants and young children lack the perception, cognition, and social maturity to remember or understand traumatic events (Buss et al., 2015; Cohen et al., 2005; Zeanah & Zeanah, 2019; Zero to Three, 2005).

Buss et al. (2015) indicated that early childhood mental health research had developed rapidly over the past 20 years. Practitioners and researchers who work with this population have contributed to understanding trauma and early childhood mental health. However, the information gained by mental health professionals has been slow to move into the field of ECCE, in part, because of the differences between the ECCE systems. Phillips and Lowenstein (2011) reported that current ECCE programs in the United States emerged from two major policy streams with different historical origins, goals, and funding sources.

The movement for early education began in the 1800s in Europe with Friedrich Froebel's work (1782 – 1852) in Germany and Maria Montessori's (1870 – 1952) in Italy (Kamerman, 2006). Child care in the United States began with women moving into the workforce during WWII. Rhodes and Huston (2012) identified landmarks in the early care and education movement. Early education programs were initiated in the 1960s as part of President Lyndon Johnson's War on Poverty. Several programs were created aimed at educational support and interventions for young children and their families living in poverty. They included: the Head Start Economic Opportunity Act of 1964, Sesame Street Workshop<sup>®</sup> (1969), and early childhood-focused research projects, among them the Perry Preschool Project (1962) in Ypsilanti, Michigan, and the Abecedarian Project (1972) in Chapel Hill, North Carolina.

It is recognized that the first three years of life are critical to optimizing brain growth and development and that this period of the lifespan has a lasting influence on an individual's life (Centers for Disease Control and Prevention, 2020b; Center on the Developing Child, 2007; National Research Council and Institute of Medicine, 2000). The use of imaging technology in neurobiology provided the opportunity to examine the developing brain. According to Halfon, Schulman, and Hochstein (2001), the years from 1990 to 2000 have been labeled the 'decade of the brain.' In this time frame, the scientific understanding of brain development accelerated aggressively.

The 1990s brought a resurgence in the focus on early care and education. The Head Start Expansion and Quality Improvement Act was passed in 1990. In 1995, Early Head Start, an expansion of Head Start, was created to support families with infants, toddlers, and pregnant women (Rhodes & Houston, 2012). In addition, state prekindergarten programs were initiated to prepare at-risk children for formal schooling (Barnett, 2011). Maxwell, Lim, and Early (2006) indicated that in 2002 President Bush sponsored the federal government passage of the early childhood initiative *Good Start, Grow Smart*, a companion to Public Law 107 – 110 No Child Left Behind (NCLB) Act of 2001. "Both sets of legislation emphasized quality and focused national attention on

the importance of 'highly qualified teachers in children's educational success" (Maxwell et al., 2006, p. 1).

Researchers have shown that a well-trained and knowledgeable workforce is a critical quality component of any ECCE program (Martinez-Beck, & Zaslow, 2006). Friedman-Krauss et al. (2019) found a wide diversity in the educational requirements for early childhood teachers. According to a joint statement by the NAEYC and the National Association of Child Care Resource and Referral Agencies (NACCRER, 2011), the professional development of the early childhood workforce varies considerably. Some are graduates of technical programs, while others take college courses or enroll in degree programs. Some teachers and administrators have college degrees in early education or related fields. Still, others have no prior related education. The majority of the early childhood workforce participates in yearly training (NACCRER, 2011).

Gerrity and Folcarelli (2008) reported that childhood trauma is a painfully common problem, both domestically and internationally. Although some children are at greater risk of experiencing trauma than others, traumatic events happen to children of all ages, socioeconomic groups, racial and ethnic groups, and geographic regions in the United States. According to Erdman et al. (2020) almost all teachers will work with a child or family who has been impacted by trauma, and research corroborates that the intervention and support a young child receives will make a-significant difference. Most educators, however, have not been educated in relevant strategies for managing traumatic reactions in young children (Early et al., 2007). Providers of professional development must respond to early childhood teacher needs. Limited research has focused on the content and quality of training that improves ECCE teacher competencies (Early et al., 2007).

# **Statement of the Problem**

Because early care and education teachers play a critical role in young children's social and emotional development, they are often on the front lines of observing children's behaviors and their interactions with peers, other adults, and the environment. ECCE personnel often recognize or identify a need, a challenge, or a red flag in a young child's development. The American Psychological Association Presidential Task Force on Posttraumatic Stress Disorder and Trauma in Children and Adolescents (2008) indicated that most children and adolescents with traumatic exposure or trauma-related psychological symptoms are not identified and do not receive any help. While there is significant research about the development of young children and early childhood learning options, there is limited research focusing on how well early childhood teachers are prepared to work with students who have experienced trauma. Zero to Three (2021) reported that until recently, the efforts and funding to provide mental health training were focused on mental health professionals and clinicians, with limited training provided to ECCE providers.

# **Purpose of the Study**

This qualitative study explored ECCE teachers' perceptions about the adequacy of the training they received to work with young children who have experienced trauma. Four purposes guided the current study. The first purpose was to investigate ECCE teachers' perceptions of how trauma has impacted young children in early childhood center-based programs. A second purpose was to explore ECCE teachers' perceptions about the information and skills needed to meet the needs of young children who have experienced trauma. The third purpose was to identify ECCE teachers' perceptions about professional development opportunities available to them that provide preparation to meet the needs of young children who have experienced trauma. Finally, the fourth purpose was to examine the perceptions of ECCE teachers about the challenges they face in accessing professional development to aid them in working with young children who have experienced trauma.

## Significance of the Study

There is limited research about the preparation of ECCE teachers related to working with young children who have experienced trauma. This study added to the body of literature on the professional development of ECCE teachers who work with young children who have experienced trauma. ECCE teachers may be interested in the current study results as they seek training and support for the challenges they experience when acquiring training and support for working with children who have experienced trauma. The study results may guide early childhood administrators to emphasize and provide professional development for ECCE educators that focuses on the impact of trauma on early childhood instruction. The study results may guide parents of young children to form partnerships with early childhood teachers to meet the needs of their children who have experienced trauma. Individuals and organizations who provide professional development to ECCE providers may be interested in the current study results as they design professional development opportunities for ECCE teachers of young children who have experienced trauma. Individuals and organizations that provide mental health services for children and families who have experienced trauma may be

interested in the current study results. State and national officials and policymakers who determine funding and policies for the well-being of young children and their families, quality early childhood care and education, including professional development, and infant-toddler mental health services, may be interested in the results of the current study. Individuals and organizations who advocate and support young children and their families, the early education workforce, and mental health professionals may also be interested in the current study results.

## Delimitations

Lunenburg and Irby (2008) defined research delimitations as the self-imposed boundaries that the researcher determines. The sample for the current study was limited to ECCE teachers employed in early childhood center-based programs in Johnson County, Kansas, from January 2020 through December 2020. The sample was limited to ECCE teachers with two or more years of experience working directly with young children in an ECCE environment. Only teachers who had contact with young children who have experienced trauma were included in the study. The study sample included ECCE teachers who voluntarily chose to participate in personal interviews.

#### Assumptions

"Assumptions are postulates, premises, and propositions that are accepted as operational for purposes of the research" (Lunenburg & Irby, 2008, p. 135). The following assumptions were made regarding this study:

1. The sample was representative of ECCE teachers employed in center-based child care programs in Johnson County, Kansas.

- 2. The ECCE education teachers participating in personal interviews understood the questions and answered honestly.
- 3. All participants in the current study had contact with at least one child who had experienced trauma during their employment in an early learning center-based program between January 2020 and December 2021.

# **Research Questions**

Creswell (2014) stated that research questions "narrow the purpose statement and become major signposts for the readers" (p. 151). Four research questions guided the current study:

**RQ1.** What are the perceptions of ECCE teachers about how trauma has impacted young children in early childhood center-based programs?

**RQ2.** What information and skills do ECCE teachers perceive they need to meet the challenges and needs of young children who have experienced trauma?

**RQ3.** What professional development opportunities do ECCE teachers perceive are available to them that provide knowledge and skills to work with young children who have experienced trauma?

**RQ4.** What challenges do ECCE teachers perceive they face in accessing professional development to aid them in working with young children who have experienced trauma?

# **Definition of Terms**

Several definitions are provided in the following section to provide a common understanding of words and terms used throughout the current study. "Consistent terminology and definitions have emerged as a critical issue for the early education field" (The NAEYC & NACCRRA, 2011, p. 3).

**Center based.** Research Connections (2020) used the term center-based to refer to child care provided in nonresidential group programs, such as public or private schools, churches, preschools, daycare centers, or nursery schools.

Early childhood education (ECE). Gwynedd Mercy University (2020) defined ECE as "the period of learning and care that takes place from birth to 8 years old" (para. 2).

**Family child care.** Rhodes and Huston (2012) and Research Connections (2020) used the term family child care to identify child care provided in a home setting. Family child care homes provide care for one or more children in the caregiver's home. Family child care may be paid or unpaid, regulated or unregulated, listed or unlisted.

**In-service training.** Research Connections (2020) referred to in-service training as professional development outside of a credentialing program that early childhood teachers attend to strengthen their skills and update knowledge and practices in the field. A specified number of in-service training hours are generally required for early childhood professionals to maintain program licensure or certification.

**Professional development.** Research Connections (2020) referred to professional development as a continuum of learning and support activities designed to prepare individuals for work with, and on behalf of, young children and their families and ongoing experiences to enhance this work. "Professional development encompasses education, training, and Technical Assistance (TA), which leads to improvements in the knowledge, skills, practices, and dispositions of early education professionals" (para.

139). Professional development may include workshops, conferences, onsite technical training and assistance, professional learning communities, online coursework, books, and professional journals (LeeKeenan & Chin Ponate, 2018).

**Trauma.** Bartlett and Sacks (2019) described trauma as one possible outcome of exposure to adversity. According to Bartlett and Sacks, trauma occurs when a person perceives an event or set of circumstances as extremely frightening, harmful, or threatening emotionally, physically, or both.

**Traumatic event or experience.** Buss et al. (2015) cited the National Library of Medicine (2013) when they described a traumatic event or experience as "one that threatens injury, death, or the physical integrity of self or others and also causes horror, terror, or helplessness at the time it occurs" (p. 2). In addition, according to the National Child Traumatic Stress Network Schools Committee (2008), situations that can be traumatic include:

- Physical or sexual abuse
- Abandonment, betrayal of trust (such as abuse by a caregiver), or neglect
- Neglect
- The death or loss of a loved one
- Life-threatening illness in a caregiver
- Witnessing domestic violence
- Automobile accidents or other serious accidents
- Bullying
- Life-threatening health situations or painful medical procedures

- Witnessing or experiencing community violence (e.g., shootings, stabbings, robbery, or fighting at home, in the neighborhood, or at school)
- Witnessing police activity or having a close relative incarcerated
- Life-threatening natural disasters
- Acts or threats of terrorism (viewed in person or on television)
- Living in chronically chaotic environments in which housing and financial resources are not consistently available. (p. 7)

Young children. IGI Global (2020) provided three definitions of young children:

- Children birth to eight years.
- The age range of children who are typically age four to eight and are enrolled in pre-kindergarten through third grade in school.
- Children aged from birth to five years of age (p. 1).

# **Organization of the Study**

This dissertation is organized into five chapters. Chapter 1 provided an introduction, including the background, statement of the problem, the purpose of the study, the significance of the study, delimitations, assumptions, the research questions, and definition of terms. Chapter 2 provides a review of the literature as it applies to the definition of ECCE; the history and development of ECCE; the description of the ECCE workforce and programs; the description of quality ECCE; the professional development of ECCE teachers; young children's social and emotional development; description of trauma experiences in young children; the history and development of mental health services; and mental health services for young children. Chapter 3 describes the methods used to conduct the study, including the research design, setting, sampling procedures,

instrument, data collection procedures, data analysis and synthesis, reliability and trustworthiness, researcher's role, and limitations. Chapter 4 presents the results of the data analysis. Chapter 5 provides a study summary, findings related to the literature, and conclusions.

#### Chapter 2

## **Review of the Literature**

The purpose of this qualitative study was to examine ECCE teachers' perceptions of how trauma impacts young children in ECCE center-based programs and the training ECCE teachers receive to work with young children who have experienced trauma. Four purposes guided the current study. The first purpose was to investigate ECCE teachers' perceptions about how trauma has impacted young children in early childhood centerbased programs. The second purpose was to explore ECCE teachers' perceptions about the information and skills needed to meet the needs of young children who have experienced trauma. The third purpose was to identify ECCE teachers' perceptions about professional development opportunities available to them that provide preparation to meet the needs of young children who have experienced trauma. The final purpose was to examine the perceptions of ECCE teachers about the challenges they face in accessing professional development to aid them in working with young children who have experienced trauma.

Prior to the development of this study, an extensive review of the literature was conducted. The summary of the literature begins with a definition of ECCE, the history and development of ECCE; the description of the ECCE workforce and programs; and the description of quality ECCE. Next, an explanation of the professional development of ECCE teachers is provided. This section is followed by an overview of young children's social-emotional development and a description of trauma experiences in young children. Finally, the literature review concludes with the history and development of mental health services and mental health services for young children.

# **Definition of ECCE**

According to White (1975), the care and education of young children have been the role and responsibility of families since the beginning of human time. Bornstein (2012) stated, "Some demands on parents are universal. For example, parents in all societies must nurture and protect their young" (p. 5). However, not all children have been nurtured and protected throughout history.

Cultures, historical events, scientific discoveries, and financial status determine who or what institutions assumed or were assigned the responsibilities associated with early child care. Heywood (2001) stated that any child's care and education were the fate of an era's religious, economic, political, and scientific beliefs. Phillips and Lowenstein (2011) and Michel (2011) indicated that families' socio-economic status has determined who will provide care for young children, the goals, and the quality of early care and education services throughout the centuries.

In the 21<sup>st</sup> century, the phrases *child care* and *early childhood education* have been used to describe young children's experiences in settings they were placed in while their parent or guardian managed work, school, or civic responsibilities (NAEYC, 2020). Rhodes and Huston (2012) stated that many child care environments are intended to promote development and learning, and many early education programs support parents while employed. Learning results from each one of a child's experiences. The profession has advocated defining such experiences for young children with comprehensive descriptions such as 'early learning' and 'early childhood care and education (NAEYC, 2020).

### **History and Development of ECCE**

Heywood (2001) indicated that for centuries, the early years of development were viewed as the preparation for adulthood and only as a means to an end, becoming a productive member of society. In many cultures, there was little value placed on children before the age of 18. Demause (2002) stated that for hundreds of years infanticide was practiced due to economic burden, perceived lack of value, or religious sacrifices. Obladen (2016) stated that in the 300s, the emperors Constantine and Valentinian moved to outlaw infanticide. For hundreds of years, regardless of socio-economic status in the western world, a solid Christian belief was held that children were inherently evil. Beating and torture were implemented daily to purge wickedness or force submission (Grille, 2006; Heywood, 2001).

According to Heywood (2001), before the 19<sup>th</sup> century, the years prior to 17 years of age were given little attention. Children were viewed as miniature adults, biding time until the young human could be a productive member of the community. "The moment children could survive without the care and attention of their mothers or nannies, somewhere between the ages of 5 and 7, they were launched into the great community of men" (Heywood, 2001, p. 11). Children joined adults in their leisure activities and trades by living and working with fully trained people. By today's standards, the care and education of children prior to the 19<sup>th</sup> century are viewed as abusive and intolerable (Demause, 2002). As humankind's understanding of the world broadened, the status of children slowly evolved (Grille, 2006).

Before the 1500s, the care and education of children depended on the knowledge and wisdom of the elders' and wives' tales (Mechling, 2021). The term wives' tale (2021) was defined as "a belief, usually superstitious or erroneous, passed on by word of mouth as a piece of traditional wisdom" (para. 4). As cultures emerged across the world and communities were established, the care and education of children continued to rely on elders' wisdom, family traditions, and old wives' tales. The reliance on experts occurred with the scientific revolution of the 16<sup>th</sup> century, the creation of universities, and the mass publication of the written word (Mechling. 2021). In addition, the artists, storytellers, and poets of the late 18<sup>th</sup> and early 19th centuries offered a romantic view of childhood that was an unrealistic portrayal of the status of children (Heywood, 2001).

Throughout most of human history, children have been apprentices and indentured servants in agriculture and the trades (Child Labor, 2020). Children were taught the skills and trades of their parents or caretakers. As the U.S. was being colonized, children were essential to the labor force. The founders of the 13 colonies believed in a Puritan work ethic, and they valued hard work over idleness. This mindset applied to children as well. As a result, child labor prevailed throughout history, reaching its peak during the Industrial Revolution (Child Labor, 2020).

Child care manuals flourished in the 19th century due to an upsurge in male pediatricians and obstetricians. At no time in antiquity had men of medicine held keen regard for the care of ordinary women and children. Instead, they produced books to prove themselves and justify being superior to the lowly midwife (Oneill, 2013).

ECCE programs in the United States developed with different goals, unique historical origins, and varied funding sources. "ECCE programs in the U.S. evolved out of diverse historical streams including child protection, early childhood education services for children with special needs, and services to facilitate mothers' labor force participation" (Kamerman & Gatenio-Gabel, 2007, p. 26). According to Michel, 2011), one stream provided interventions to enhance children's well-being and development. In the 19<sup>th</sup> century, the Industrial Age, early care emerged as the working classes needed places for their children to be housed while employed. Child care in America included an array of formal and informal arrangements commonly connected with immigrants, the poor, and minorities and was classified as custodial and charitable (Michel, 2011). According to Cahan (1989), the first efforts in early care were nurseries. These were charitable inventions to save or rehabilitate indigent children (Cahan, 1989). The first early education programs were nursery schools founded in Massachusetts in the 1830s. Later, kindergarten programs were established based on Froebel's work (Kamerman, 2006). Gradually, early education programs increased to offer an enhanced experience to middle-class families (Child Care Lounge, 2018; Kamerman & Gatenio-Gabel, 2007; Michel, 2011).

By the mid-1800s, the long-term consequences of child labor on children's education and health provoked reformers, labor groups, teachers, and churches (Paul, 2017). The University of Iowa Labor Center (2004) reported that child labor reform and union organizing were often intertwined. Their shared initiatives were undertaken by organizations directed by middle-class consumers and working women. During the Great Depression, the deep-rooted conventions of child labor were disrupted because Americans desperately needed jobs (Child Labor, 2020). The National Child Labor Committee's efforts to end child labor was integrated with endeavors to provide no cost required education for all children. This was accomplished in the passage of the Fair Labor Standards Act in 1938 (University of Iowa Labor Center, 2004).

Beginning in the early and middle 19<sup>th</sup> century and emerging mainly after World War II a predominant theme was the transition from private charity to public responsibility (Kamerman, 2006). Government involvement moved from child labor protection to child care to support working families. Kiesling (2019) stated that during the Depression, the Works Progress Administration ran a federal and state collaborative program of nursery schools to create jobs. World War II moved women to the workforce, although women with young children were discouraged from participating in the war effort. Kiesling (2019) noted, "The official position of many people in government was that women with young children should be the last people brought into the labor force" (para. 9). Ertman (2019) reported that the U.S. Senate passed the first and only national child care program on June 29, 1943. \$20,000,000 was allocated to provide public child care for employed women. With the end of World War II, government funding of child care services ended (Ertman, 2019).

According to the National Research Council and Institute of Medicine (2000), the structured study of infant behavior appeared in the early to the mid-20<sup>th</sup> century. At that time, researchers in both embryology and evolution posed critical questions about the origins and development during the lifespan of humans. By the 1920s, pediatricians, educators, and social workers were deepening their interaction with psychologists in the field of child studies which resulted in the creation of a dynamic, multidisciplinary, scientific discipline (Zeanah & Zeanah, 2019). Following World War II, technological advances marked the dawn of the Technological Revolution

(Doncaster, 2021). However, in 1958 few psychologists intentionally undertook research with children younger than three years old. Although somewhat mindful of infant care, society remained unfamiliar with any unique educational relevance of the early years (White, 1975).

Alleyne (2016) and the Center for Substance Abuse Treatment (2014) described the 1960s as an era of social revolution. African Americans, women, and individuals with disabilities advocated for rights and services. President Lyndon Johnson declared the War on Poverty. White (1975) summarized the role of early care and education through the 1960s. He stated that daycare was extensively established to meet the needs of parents instead of the child. The early care and education field weathered the roller coaster of government legislation proposed, ignored, passed, expired, and not reauthorized (First Five Years Fund, 2021a). Labor market policy, public (social) assistance policy, education policy, child welfare policy, and child development research expanded ECEC policies and programs (Kamerman & Gatenio-Gabel, 2007). Rhodes and Huston (2012) described numerous intervention programs targeting children of families living in poverty initiated through federal funding. Head Start, Sesame Street, and research demonstration models were created. All initiatives had a similar goal - to break the cycle of poverty by educating adults and children and improving living standards in lower socio-economic communities (Rhodes & Huston, 2012). In addition, Parents as Teachers (1981) emerged providing a home visiting program to young children and their parents regardless of socio economic status. The program goals included increase parent knowledge of early childhood development, enhance parenting skills,

identify developmental delays, aid in a child's school readiness and success, and prevent child abuse and neglect (Hackett, 2011; Parents as Teachers, 2021).

The 1990s brought a renewed focus on young children. Reid (2019) noted that in 1989 the United Nations ratified the Convention on the Rights of the Child, which guaranteed the protection of children's rights to grow and thrive. The movement gained momentum with the World Conference on Education for All in Jomtien, Thailand. According to Kamerman (2006), the Jomtien Declaration (1990) and Dakar Framework for Action (2000) launched a dynamic initiative to educate all. At the forefront was the tenet that access to preschool education was a lawful right.

Scientific advances in neuroimaging elevated the importance of the early years of child development (Shore, 1996). As a result, brain development became a new field of study. Hustedt and Barnett (2011) reported that with the growing recognition that poverty's disadvantages begin well before age 3 or 4, Early Head Start (EHS) was introduced in 1995. In addition, state pre-kindergarten programs were created. The goal was to prepare four-year-old children at risk of low achievement (e.g., from families with low incomes, children with special needs, or limited English proficiency,) for K -12 formal schooling (Hustedt & Barnett, 2011).

As the new millennium began, Congress passed Public Law 107 – 110, the No Child Left Behind (NCLB) Act of 2001. In addition, the added focus was placed on early care and education with the Bush administration's initiative *Good Start, Grow Smart* (The White House, 2002). Three major areas were addressed in the law and Bush initiative: strengthening Head Start, partnering with states to improve ECCE, and providing information to caregivers, teachers, and parents. Kamerman and Gatenio-Gabel (2007) cited multiple factors in the expansion of ECCE programs. These included the increase of single and married mothers in the labor market, the welfare reforms that required mothers to work, and the focus on the impact of early learning and future school success. The National Academies of Sciences, Engineering, and Medicine (2018) noted, "Each purpose has been reflected in the evolution of early care and education over the past century and has been prioritized differently in various early childhood education policies over time" (p. 2). According to Kamerman (2006), the result has been a fragmented ECCE system. Multifaceted societal beliefs and explicit funding reinforced the differences. The last 25 years substantiated the need for a cohesive early care and education system. ECCE authorities and proponents are steadily more convinced of the need to incorporate all these program types (Kamerman, 2006).

Professional and consumer organizations and associations have struggled to raise public consciousness for quality ECCE and educational and financial resources for parents and providers. The efforts by NAEYC, the Children's Defense Fund, and Zero to Three have been critical in impacting public awareness and government policy. Moreover, as the world's population copes with the impact of the global pandemic, greater focus has been placed on the essential role of ECCE not only on the lives of children but the socio-economic well-being of the nation (Guarino, 2021).

#### **Description of ECCE Workforce and Programs**

A joint statement by NAEYC and NACCRRA (2011) described the ECCE workforce as individuals:
working with young children (infants, toddlers, preschoolers, and school-age children in centers, homes, and schools) and their families or on their behalf (in agencies, organizations, institutions of higher education, etc.), with a primary mission of supporting children's development and learning. (p. 5)

According to the Institute of Medicine and National Research Council (2012), "Terminology can be problematic with such a wide spectrum of work represented, even among the members of the workforce themselves" (p. 6). Rhodes and Huston (2012) stated, "Labels for people who work in this field include 'teacher,' 'child care worker,' 'daycare provider,' and 'babysitter,' among many others" (p. 11). NAEYC (2009) stated that the term *teacher* is always intended to refer to any adult responsible for the direct care and education of a group of children in any early childhood setting. Teachers are not only individuals in a classroom but also include family providers, infant/toddler caregivers, and specialists in other disciplines who fulfill the teacher's role.

Maroto and Brandon (2011) estimated that the employed ECCE workforce includes 2.2 million individuals who comprise 31% of the total U.S. teaching workforce for all age groups (infants and toddlers through college-level students). Fifty-one percent work in center-based programs, 27% are paid relatives, 12% are nonrelatives in family child care settings, and 11% are individuals working in the child's home (Maroto & Brandon, 2011). The Institute of Medicine and the National Research Council (2012) noted that researchers in the field have indicated that the definitions of child care workers and preschool teachers do not reflect the reality of their work, especially the overlap in their respective roles. Erdman et al. (2020) acknowledged, "All educators who work with children and their families are professionals" (p. 1). Administrative organization and the funding stream determine the setting or function of an early care and education program (Rhodes & Huston, 2012). Vandell (2004) stated that early care and education settings might be grouped into three widerange categories: center-based, family child care, and in-home care. Multiple funding streams finance early care and education services. They include family fees, federal and state allocations, and independent financial resources (National Academies of Sciences, Engineering, & Medicine, 2018).

Cui and Natzke (2020) reported that young children participate in multiple kinds of early care and education settings before they enroll in kindergarten. Several children attend center-based environments such as Head Start programs, child care centers, or preschools. Others are looked after in relatives' or nonrelatives' homes or are regularly taken care of only by their parents (Cui & Natzke, 2020). DeBord (n.d.) stated that group care for two or more children is usually provided in child care centers. A program may be established in a facility planned explicitly for the care of multiple children or a church, school, or home. ECCE centers may be identified as community-based, Early Head Start, faith-based, Head Start, military child care, nursery school, preschool, public preschool, therapeutic preschool, and corporate care (Child Care Aware of Kansas, 2019; Kamerman & Gatenio-Gabel, 2007). The faith-based programs are part-time or full-time early childhood programs affiliated with or supported by a church or other religious organization (DeBord, n.d.). Military Child Care is ECCE available to families serving in the armed services. Programs were regulated and supported by the Department of Defense (Research Connections, 2020).

According to Research Connections (2020) and Rhodes and Huston (2012), Head Start and Early Head Start are federally funded programs that provide comprehensive services to low-income families. Services included early childhood education, health and nutrition, mental health, and parent engagement. In addition, early Head Start targets pregnant women and families with infants and toddlers to support optimal child development while helping parents/families move toward economic independence. Federal grants that support ECCE are awarded to local public or private agencies (Research Connections, 2020).

Before attending kindergarten, children 2 ½ to 5 years of age participate in early care and education in preschool. DeBord (n.d.) reported that originally preschools were half-day programs. As early care and education services were expanded, long-standing programs offered full-day activities and retained 'preschool' or 'nursey' in their business name. Preschools may be publicly or privately operated and may receive public funds (DeBord, n.d; Research Connections, 2020). Publicly-funded preschool programs typically serve children from disadvantaged families. In contrast, private preschool programs supported by parent fees are more likely to serve children from all backgrounds, and the focus is more on the child than on providing support to the family (Kamerman & Gatenio-Gabel, 2007). Therapeutic preschools and special needs services meet the unique needs of children who have identified disabilities and learning challenges as determined by the Individualized instruction, and support services occur in the classroom, at home, in medical facilities, or other settings (Research Connections, 2020).

Family child care is offered in a home environment for a small group of children in the provider's home (Research Connections, 2020). Per state regulations, programs may be licensed, regulated, or not licensed or regulated. The caregiver may or may not be trained in child care and development. In rural areas, the primary setting of available child care is provided by relatives or family child care homes (DeBord, n.d.). In-home providers (identified as babysitters, nannies, and extended family) provide care for children in the child's place of residence and may or may not be trained in child development or child care (DeBord, n.d.).

The National Academies of Sciences, Engineering, and Medicine (2018) stated that early care and education fulfilled numerous functions in the United States: facilitating child development, supporting parental employment, and encouraging investment in the future workforce. According to Kamerman and Gatenio-Gabel (2007), in the United States, excluding the government, there are many foundations, private advocacy, public policy, research, think-tank, and university organizations-focused on early childhood education and care policies. With the vast number of stakeholders and multiple objectives of ECCE programs it may be challenging to succinctly describe the ECCE workforce.

## **Definition of High-Quality ECCE**

High-quality ECCE dramatically impacts a child's development and well-being (First Five Years Fund, 2021b). One of the more constant and prevalent findings in the research connects the quality of child care that children are provided to nearly every assessment of development that has been investigated (National Research Council and Institute of Medicine, 2000). Zeanah and Zeanah (2019) reported that language and cognitive outcomes in young children, along with behavior and social outcomes, are linked with the quality of care. Van der Kolk (2014) stated that the quality of early caregiving is highly critical in thwarting mental health problems, distinct from other traumas.

All ECCE programs, nonprofit and for-profit child care centers, public schools, and family care providers can provide high-quality early childhood education for children. In the United States, there are considerable differences between early education settings. The approach to quality is not-one size fits all, and quality is not limited to one type of program (Workman & Ullrich, 2017). Kamerman and Gatenio-Gabel (2007) stated that across school-based pre-kindergarten, Head Start, and center-based programs, there is no consensus on the definition of or standards regarding the quality of the ECCC settings.

The National Research Council and The Institute of Medicine (2000) provided a joint statement that indicated that when studies of child care quality are reviewed, three categories emerge: the child-provider relationship, the established nuances and procedures for care, and the local policy framework. Kamerman and Gatenio-Gabel (2007) stated that subjective and quantifiable variables are used when assessing quality in centers. Subjective variables require direct observation of indicators such as staff-child interactions and relationships. Quantifiable variables include adult-child ratios, group size, staff qualifications (education and training), salaries, and turnover rates. These dimensions of quality can be regulated and counted.

NAEYC (2020), Research Connections (2020), and Early Childhood Education (2020) identified additional characteristics of high-quality ECCE programs, which included:

- A clear statement of goals and philosophy that is comprehensive and addresses child development areas.
- A safe, nurturing, and stimulating environment, with the supervision and guidance of competent, caring adults.
- A strong foundation in language development, early literacy, early math, and each child's family background.
- Developmentally appropriate practices include environments, activities, and adult-child interactions adapted to meet a specific group of children's age, characteristics, and developmental needs.

Burchinal, Kainz, and Cai (2011) described factors of quality child care that included close teacher-child relationships, frequent sensitive interactions between the child and the teacher, respectful and effective behavior management, well-designed instruction, and a rich physical learning environment. Boothe Trigg and Keyes (2019) included an additional factor, "Quality child care demands individualized care of infants and young children, which translates into low teacher-student ratios" (p. 604). Meeting the physical and emotional needs of very young children, such as feeding, diapering, comforting, and carrying a non-mobile child, requires engaging one child at a time rather than small or large group activities.

Research Connections (2020) stated that consistency and limiting transitions in a child's early years promote the profound human connections that young children require

for optimum early brain development, self-regulation, and learning. The caliber of care fundamentally depends on the quality of the relationship between the early care and education staff member and the child (The National Research Council and The Institute of Medicine, 2000). An enticing environment and a phenomenal curriculum can be compelling. However, positive development will not be fostered without adept and consistent child care providers (The National Research Council and The Institute of Medicine, 2000).

# **ECCE Teacher Professional Development**

The National Research Council and The Institute of Medicine (2000) jointly stated that a teacher's level of preparation and currently focused education in child development had been repeatedly linked with high-quality interactions and children's development in all early care and education settings. Professional development refers to a range of educational and resource activities developed to prepare individuals to care for, work with, and advocate for young children and their families, as well as improve practices in the field. Professional development includes education, training, and technical assistance (TA), which results in better knowledge, skills, strategies, and qualities for early education professionals (Research Connections, 2020).

Rhodes and Huston (2012) stated, "Professional development and institutional supports are needed to foster a strong ECCE profession and high-quality ECCE" (p. 4). However, there are discrepancies in the education requirements for ECCE teachers across programs in the United States. Rhodes and Huston (2012) indicated: "The ECCE workforce is highly diverse and some characteristics, particularly educational attainment and qualifications, vary by sector" (p. 5). The Institute of Medicine and The National Research Council (2012) jointly identified the ECCE workforce as a range of people who, on one end of the continuum, have minimal training and primarily provide child care devoid of a focus on educational aims. On the other end, individuals with concentrated advanced degrees provide well-designed learning activities, with many others in the middle. Kamerman and Gatenio-Gabel (2007) reported that in 10 states, no special training is required for ECCE teachers, and in 20 states, a bachelor's degree is not required of pre-kindergarten teachers.

State licensing, regulatory bodies, and organization administrators determine the employment and education requirements for ECCE teachers. For example, the Kansas Department of Health and Environment (KDHE, 2020) identified the education and training requirements for all staff working in licensed preschool and child care programs in the state (see Appendix A). The minimum educational requirement for staff members working with children is a high school diploma or the equivalent. Within Kansas licensing, there are variations in education and experience that allow teachers to be lead teachers in classrooms of infants, toddlers, two-year-olds, and preschoolers (KDHE, 2020).

According to the KDHE (2020), regardless of the new staff member's background, the individual must participate in orientation or pre-service training. Research Connections (2020) described pre-service training as the educational experiences that an ECCE "staff member may undergo prior to assuming a particular role or position within a child care program" (para. 135). KDHE (2020) child care licensing requires that each employee or volunteer working with children shall complete orientation within seven calendar days of the start date. Orientation topics include licensing regulations, the policies and practices of the early care and education program, including emergency procedures, behavior management and discipline, the schedule of daily activities, care and supervision of children in care (including any special needs and known allergies), health and safety practices, and confidentiality. Pre-service training or orientation is usually provided by the employer or hiring body (Child Care Aware® of America, 2016; KDHE, 2020)

When early education programs were part of the public school system, the criteria for educational requirements and compensation for the preschool teacher were similar to K-12 teachers (Rhodes & Huston, 2012). Unlike the requirements for teachers in public education systems, state teacher certification is not a requirement for most ECCE teachers (Institute of Medicine and National Research Council, 2015). In addition, many positions do not require a bachelor's degree. Schonert-Reichl, Kitil, and Hanson-Peterson (2017) indicated, "Courses on child and adolescent development can be found in the majority of colleges of education in almost all U.S. states" (p. 12). According to Schonert-Reichl et al. (2017), research suggested a modicum of evidence that college students in teacher preparation programs are informed about children's and adolescents' social and emotional development.

Federal funding often carries requirements for qualifications and credentials. For example, in the reauthorization of Head Start (Improving Head Start for School Readiness Act of 2007), provisions were included to require higher qualifications for the Head Start teaching workforce (Rhodes & Huston, 2012). In addition, Kamerman and Gatenio-Gabel (2007) reported that the adaptation of comprehensive learning standards in many states had improved preschool quality. However, variation in early education quality continues across the country.

ECCE professional development occurs through various venues in a variety of environments. Educational opportunities are provided through two-year and four-year higher education institutions to earn bachelor's or associate degrees and specialty certificates. In addition, professional organizations and child care resource and referral agencies offer credentials, certifications, and continuing education. In-service or ongoing training is professional development that early childhood professionals participate in to gain up-to-date knowledge and practices to augment their skills in the field. An ECCE program may require in-service training hours to continue employment (Research Connections, 2020). KDHE (2020) identified that 16 clock hours were needed to meet annual in-service requirements for each licensure year following 2020.

Child Care Resource and Referral Agencies (CCR&R) are local and statewide organizations, usually funded by the Child Care and Development Fund (CCDF). The agencies provide many services to the families and the ECCE professionals that serve them. Resources offered include referrals and guidance to parents seeking child care, professional development and training for early childhood professionals, mental health services, and additional educational resources for families (Research Connections, 2020). The Family Conservancy is the local resource and referral agency serving 18 countries across eastern Kansas and western Missouri. The agency's mission is to help children and families achieve a lifetime of success (The Family Conservancy, 2021). It offers families resources in early childhood care and education resources, individual and family counseling, and community-based family support services. The early care and education workforce is supported with professional development opportunities and infant-toddler mental health consultation,

Although many individuals in the ECCE workforce are skilled and committed, their ability to provide high-quality experiences for young children is hindered by a lack of shared objectives and identity, lack of institutional support, insufficient or ineffective training, and inadequate compensation (Rhodes & Huston, 2012). The National Research Council and The Institute of Medicine (2000) jointly indicated that as the quickly emerging science of early childhood development expands, its complexity will rise, and the gap between the working knowledge of service providers and state-of-the-art science will be startling. The professional dilemmas that this creates for the early childhood field are daunting.

### Young Children's' Social and Emotional Development

The Eunice Kennedy Shriver National Institute of Child Health and Human Development (2009) stated, "Development is a window into the life of a child and lays the foundation for the life course of the mature adult" (p. 4). Human development is contingent on and stimulated by the quality of our connections with one another (Bailey, 2017). According to the National Research Council and The Institute of Medicine (2000), the foundation of healthy development is human relationships and the outcome of social connections on relationships.

Child development is defined as the *stages* by which children gain skills in the domains of physical, social-emotional, speech and language, and cognitive (Research Connections, 2020). Zeanah and Zeanah (2019) stated that among the most significant factors of the cultural framework of infant development are family beliefs,

interpretations, and explanations of a child's behavior. According to the National Research Council and The Institute of Medicine (2000), early growth involves the steady progression from total dependence on others to handling the world for us to gain the abilities required to succeed independently.

The domains of child development are not clearly defined and often overlap. The number of developmental domains identified and described is based on the field of study in which a professional is engaged (medicine, psychology, education, or mental health). Research Connections (2020) noted that developmental domains are-intertwined. What occurs in one domain impacts growth in the other domains. For this study, the researcher grouped the domains of child development into the following categories:

- Physical: the growth of the child's body, muscles, and senses and development of gross motor and fine motor skills.
- Cognitive: the ability to remember, focus, learn, problem-solve, reason, understand, and adapt.
- Receptive Language: the ability to listen and respond.
- Expressive Language: the ability to communicate non-verbally and verbally.
- Social-emotional: the expression of feelings, relationships with adults and children, and the ability to self-regulate.

Rosenblum, Dayton, and Muzik (2019) recognized that learning domains are frequently investigated separately, yet social and emotional development are inherently interwoven with all other domains. Social and emotional development is a growth process whereby children gain the competencies to recognize and identify their feelings/emotions; express and manage their own positive and negative emotions in a constructive way; develop relationships and empathy with other children and adults; discover their environment; and self-regulate (Bailey, 2011; Niemi, 2020; Research Connections, 2020; Zero to Three, 2005). The cornerstone of early childhood development is self-regulation. It transects all domains of learning (The National Research Council and The Institute of Medicine, 2000). Conscious Discipline (2021) indicated that more than any other indicator, social-emotional skills, such as selfregulation, are a greater predicator of life success.

Healthy social-emotional development is equivalent to infant mental health (Zero to Three Infant Mental Health Task Force Steering Committee, 2001). Zeanah and Zeanah (2019) stated that behavioral indicators of infant mental health include the ability to communicate emotions to parents and teachers, independent engagement with the environment, and self-regulation. The National Research Council and The Institute of Medicine (2000) noted that the extensive array of individual characteristics among young children frequently makes it challenging to discern normal variations and developmental delays from temporary disorders and ongoing disabilities. Boothe Trigg and Keyes (2019) cited Egger and Angold (2006) who reported that approximately 12% of preschool-age children, ages 2 to 5 have social-emotional or mental health problems that contribute to perilously challenging behaviors.

Developmental milestones are age-specific tasks or a set of functional skills that experts recognize that most children should demonstrate within a certain age range (Moodie et al., 2014; NAEYC, 2020). Professionals and families may use milestones to track a child's development, behavior, and learning. Not demonstrating a specific milestone may identify a reason for concern or a possible developmental delay (Research Connections, 2020). According to Godoy et al. (2019) a considerable number of parents of young children reported high incidences of disruptive behavior but they are more concerned about their child's delays in social-emotional development. According to Zeanah and Zeanah (2019), early child care providers are one example of a worldwide setting for enhancing social-emotional and cognitive development. Social and emotional learning is an integral part of human development and education.

# **Trauma Experiences in Young Children**

According to Van der Kolk (2014), trauma impacts those who are closely involved and the people associated with them. Furthermore, the American Psychological Association Presidential Task Force on Posttraumatic Stress Disorder and Trauma in Children and Adolescents (2008) explained that a traumatic event is a situation that threatens injury, the physical integrity of self or others, or death. In addition, it may cause helplessness, terror, or horror at the time of the occurrence.

Significant sources of childhood trauma are adverse childhood experiences (ACE). Bartlett and Sacks (2019) noted that the term ACE was conceived by researchers Vincent Felitti, Robert Anda, and their colleagues in their influential study undertaken from 1995 to 1997. Adverse childhood experiences are described as all types of abuse, neglect, and possibly other trauma experienced by children younger than 18 years of age (Bartlett & Sacks, 2019). Specific examples of ACEs include exposure to physical, emotional, or sexual abuse; physical and emotional neglect; and witnessing violence, severe mental illness, or substance misuse in the home (Centers for Disease Control and Prevention, 2020b; Houry, 2019). In addition, the National Child Traumatic Stress Network (2010) identified traumatic events that included suicides and other deaths or

losses, domestic or sexual violence, community violence, medical trauma, vehicle accidents, war experiences, and natural and human-made disasters.

Van der Kolk (2014) noted that trauma is specifically an event that overwhelms the central nervous system altering how individuals process and recall memories. High levels of chronic stress may rewire the brain's developing architecture. "Trauma is not the story of something that happened back then. It's the current imprint of that pain, horror, and fear living inside people" (Psychotherapy Networker, 2014). According to Bartlett and Sacks (2019) toxic stress can occur when a child encounters persistent and intense adversity (e.g., domestic violence, chronic neglect, severe economic hardship) without sufficient support from a care-providing adult. Van der Kolk (2014) indicated that being traumatized means chronically managing life as if the trauma were ongoing, enduring, and inflexible. The past taints each new experience or situation. A traumatic experience may be triggered by upsetting emotions or abrupt or aggressive actions. At the inkling of threats, the activated disturbed brain networks release vast amounts of cortisol which generates distressing reactions, acute physical sensations, and unpredictable and hostile behaviors (Van der Kolk, 2014).

According to Van der Kolk (2014), the feeling of being safe with other people is probably the most critical aspect of mental health. Safe connections are fundamental to meaningful and satisfying lives. Furthermore, numerous researchers have demonstrated that having a positive support system establishes the sole most robust safeguards against becoming traumatized (Van der Kolk, 2014).

# History and Development of Mental Health Services

Van der Kolk (2014) indicated that society frames the interpretation of traumatic stress. Mental health treatment requires examining the cultural perspective of mental illness and the social evolution of mental health. During the nineteenth century, individuals experiencing mental illness were viewed as possessed (Doncaster, 2021). The Greek physician Hippocrates recommended that people change environments. Persons afflicted with mental illness were isolated from the community. In 1407 the first mental health facility opened in Spain (Doncaster, 2021).

Doncaster (2021) indicated that Burton wrote *The Anatomy of Melancholy*, the first medical discourse on mental welfare, in 1621. Asylums to house the mentally insane were opened in the 18<sup>th</sup> century. As early as 1780, Pinel advocated for the humane treatment of individuals with mental illness. Writings by Dorothea Dix and Nelly Bly in the mid to late 1800s created public awareness of the mistreatment and squalor found in institutions caring for the mentally ill. Treatment of patients was tortuous or nonexistent, and many practices continued through the 1960s (Doncaster, 2021; Foundation Recovery Network, 2021; Samels & MacLowry, 2002).

In the 20<sup>th</sup> century, the first scientific studies of mental illness were undertaken by Emil Kraspelin (Samels & MacLowry, 2002; Tracy, 2019). After being a patient and subjected to abuse in state mental hospitals, Clifford Beers founded Mental Health America in 1909 (Mental Health America, 2020; Samels & MacLowry, 2002). The organization proved to be a driving force in care and advocacy for mental health services. Freud (1856-1939) and Jung (1875-1961) contributed psychoanalytic theories and

42

therapies (Samels & MacLowry, 2002). The 1930s introduced electroshock and cold water therapies (Foundation Recovery Network, 2021).

President Truman signed the Mental Health Act in 1947, which created the National Institute of Mental Health. Treatment of the mentally ill changed in the 1950s (Foundation Recovery Network, 2021). State hospitals were deinstitutionalized, and the first antipsychotic drugs were available. The care and treatment of individuals in need of mental health services moved to community mental health centers, psychiatric hospitals, correctional facilities, and homelessness (Foundation Recovery Network, 2021; Lumen Learning, n.d.).

The experiences of WWI and WWII veterans brought to light the impact of trauma on human behavior. Psychologists and psychiatrists studied and recorded the accounts of soldiers returning from wars. Van der Kolk (2014) reported that in 1941 Kardliner-wrote *The Traumatic Neuroses of War*, describing the *shell shock* experienced by WWI veterans. Today, veterans' traumatic experiences are referred to as post-traumatic stress disorder (PTSD). Before the 1960s, studies and publications reporting PTSD were not frequently published (Van der Kolk, 2014).

Mental Health America (2020) indicated that Congress and advocacy groups paved the way for mental health treatment and services offered in the 21<sup>st</sup> century. In 1955 Congress created the Commission on Mental Illness and Mental Health, followed by the Community Mental Health Act in 1963 (Mental Health America, 2020). Consumers founded the National Alliance for the Mentally Ill in 1979 (Samels & MacLowry, 2002). Through the 1970s and 1980s, the mental health needs of men and women returning from the Vietnam War precipitated professional and public acknowledgment of the behaviors associated with trauma (Magruder et al., 2015; Tull, 2020). According to Van der Kolk (2014), 1980 was a turning point when a group of Vietnam veterans, assisted by the New York psychotherapists Robert J. Lifton and Chaim Shatan, successfully lobbied the American Psychiatric Association to establish a new diagnosis: post-traumatic stress disorder (PTSD). PTSD was defined as common characteristics experienced, to some degree, by all veterans. Concurrently, health professionals, psychiatrists, and therapists provided therapy and counseling to women who revealed they were sexually abused as a child (Van der Kolk, 2014). Individuals having experienced physical, emotional, and sexual abuse as children were heard. Their behaviors were recognized and identified, and efforts were made to provide support and therapy (Center for Substance Abuse Treatment, 2014; Van der Kolk, 2014). Congress passed the Protection and Advocacy for the Mentally Ill Act in 1985, and in 1990 the Americans with Disabilities Act (Mental Health America, 2020).

Van der Kolk (2014) reported that since the 1990's brain imaging tools have shown what happens inside the brains of traumatized people. These scans proved essential in understanding the damage inflected by trauma and influenced the development of entirely new avenues of recovery and healing. De Young et al. (2011) found that despite decades of statistical data, counselors generally have limited knowledge of the impact of traumatic events on younger children compared to older children and adolescents. Zeanah and Zeanah (2019) reported that the DSM-5 (American Psychiatric Association, 2013) has made a definitive endeavor to be more developmentally responsive and included a sub classification of post-traumatic disorder defined for preschool children. Generally, the revisions are comparatively small. In comparison, Zero to Three (2016) published  $DC:0-5^{TM}$  Diagnostic Classification of Mental Health and Developmental Disorders of Infancy and Early Childhood, and in 2021 released  $DC:0-5^{TM}$  Version 2.0. These publications focused significant attention on the mental health and disorders of children from birth through 5 years of age. Over the past 20 years, research in early childhood mental health has advanced expeditiously. Researchers and practitioners who work with young children offer insights into trauma and early childhood mental health (Buss et al., 2015)

### **Mental Health Services for Young Children**

Throughout the ages, little attention has been given to children's mental health needs (Zeanah & Zeanah, 2019). In 1910, Mental Health America facilitated the opening of over 100 Child Guidance Clinics aimed at preventing mental health issues, early intervention, and treating children (Mental Health America, 2020). In the years that followed, limited information on infant mental health was available. In 1975 Dr. Burton White wrote that he could not recommend anyone for children's mental health issues. Instead, White recommended Dr. Spock or a pediatrician or personal family physician when it was a matter of physical health issues. White explained that child psychiatrists usually do not work with children younger than three years of age, although programs may be identified for the treatment of emotional disabilities. According to White, sometimes a program for the early treatment of emotional disabilities or potential emotional handicaps may be found. Regrettably, there is little to offer parents beyond emotional support and recommendations in areas of young children's mental health. Woefully, little was available regarding well-versed or skilled professionals in managing behavior disturbances (White, 1975).

Van der Kolk (2014) stated, "Until 2001, there was no comprehensive organization dedicated to the research and treatment of traumatized children" (p. 157). Congress established the NCTSN as part of the Children's Health Act (2000). Its mission was to raise the standard of care and improve access to services for children and families who experience or witness traumatic events throughout the United States (NCTSN, 2010). Barwick and Urajnik (2021) reported that socio-demographic research corroborated that many children with mental health disorders were not provided the therapy they require. Approximations gleaned from an amalgamation of epidemiological studies indicated that in the U.S only 20% of children with mental health problems receive specialized services (Barwick & Urajnik, 2021; Brinkman, 2022; Centers for Disease and Control Prevention, 2022). According to Buss et al. (2015), Scheeringa and Haslett (2010), and Zeanah and Zeanah (2019), there are several reasons for the lack of early childhood treatment options, including:

- Historically, researchers have spent little time and effort studying the impact of trauma on young children.
- Resistance to the idea that early childhood mental health is critical.
- Apprehensions in diagnosing young children with mental disorders.
- Awareness of the stigma that a trauma-related mental illness diagnosis may plague young children.
- The misconception that young children lack the cognitive skills and social maturity to understand or remember traumatic events.
- Young children may not comprehend the vocabulary or the directions used in specific assessment tools.

• Diagnostic criteria, tools, and assessments for older children and adults were not developmentally appropriate for young children.

Van der Kolk (2014) indicated the advancement of mental health services for children and adults has emerged from three branches of science. Each has led to increased knowledge about the effects of abuse, neglect, and psychological trauma. These new fields are neuroscience, the study of how the brain supports mental processes, developmental psychopathology, the study of the impact of adverse experiences on the development of the mind and brain, and interpersonal neurobiology, the study of how our behavior influences the emotions, biology, and mindset of those around us (Van der Kolk, 2014).

Infant mental health is a multidisciplinary and aggregate field of study, services, and policies regarding emotional and psychological wellness of young children from before birth to 5 years of age (Lieberman et al., 2011). The U.S. Department of Health & Human Services, Administration for Children and Families, Administration on Children, Youth and Families, Children's Bureau (2020) identified mental health services as activities that address matters negatively influencing development, learning, or socialization or involve emotional disorders or dysfunctional behavior. Mental health services are usually provided by private or public mental health agencies and include residential and non-residential activities. Zeanah and Zeanah (2019) indicated that the goals of infant mental health treatment are to decrease or remove-hardship, to impede adverse outcomes (developmental delays and deviance, delinquency, school failure, psychiatric morbidity, interpersonal, isolation or conflict), and to develop healthy outcomes by improving resilience and social competence. Children younger than five typically experience rapid developmental changes that are often misinterpreted or not fully accounted for, which hinders proper diagnosis and intervention (Zero to Three, 2005). Diagnosing young children who have experienced trauma is a complicated process. Lieberman et al. (2011) identified multiple challenges:

- the transitory developmental growth in the first 5 years of a child's life,
- the inabilities in preverbal and nonverbal children to self-report,
- the impact of empirical experience on the young child's behavior,
- the extensive amount of time required to observe current unhindered appropriate developmental milestones,
- the inaccuracies of parents and caregivers as-observers and reporters,
- and the required time and cost in observing young children in various environments.

Zeanah and Zeanah (2019) indicated that the most unique characteristic of the first years is the distinct significance of the numerous interconnected contexts (e.g., family caregiver-infant relationship, social, cultural, historical) within which young children grow. According to Boothe Trigg and Keyes (2019), many infants experience care for extended periods outside of their homes. With that in mind, out-of-home child care for infants and young children must be examined when contemplating the mental health of infants and young children. Essential topics to be considered are the quality of the care provided, the attachment to nonparent caregivers, and the focus on the quintessential social-emotional development of young children. ECCE teachers and administrators observe red flags in development or challenging behaviors (Sorrels, 2015). Kaiser and Rasminsky (2021) stated that the phrase *challenging behaviors* can refer to

several actions. Challenging behavior is any action that impedes a child's socialemotional or cognitive development; is detrimental to the child, other children, or adults; or puts a child in jeopardy for later social struggles or academic failure.

Researchers have corroborated that traumatic events can affect children. Yet, the wide-ranging counselor community appears unfamiliar with infant mental health which deters referrals for this at-risk group of young children. A counselor may not offer support services for a child in a family where domestic violence is witnessed because the clinician lacks knowledge and training to determine trauma-related mental illness in young children (Buss et al. (2015). Van der Kolk (2014) indicated that the intellectual, psychological, biological, or ethical development of children who have experienced trauma was not systematically taught to pediatricians, child-care workers, or part of graduate social work or psychology schools.

Young children access mental health services through a referral from social services, pediatricians, and ECCE professionals. Infant and Early Childhood Mental Health Consultation (IECMHC) is a strategic intervention that partners mental health professionals with early childhood professionals and families to support a young child's social and emotional development and addresses behavior and mental health concerns in ECCE programs (Raine Group, 2014; Research Connections, 2020). Support and services may include an individualized education and care plan for the child, classroom resources, coaching/mentoring with the teacher for all children, or consultation with administrators to develop policies and procedures to benefit children, families, and staff.

Prevent Child Abuse America (2020) offered strategies to reduce or prevent traumatic or adverse experiences that included:

- Providing environments for safe, stable, nurturing relationships for children, families, and communities.
- Providing high-quality early care and education and home visiting programs.
- Assisting adults and children in gaining skills to manage and respond to stress and challenges.
- Accessing screening, referrals, and support for early interventions.

Many of the recommendations may be implemented without direct services of mental health clinicians and are included in the array of services provided by ECCE programs. Within the last five years, the need for mental health services for young children began to be seriously addressed by policymakers and mental health clinicians (Zero to Three (2021). The National Center for Children in Poverty, the Substance Abuse and Mental Health Services Administration (SAMHSA), and Zero to Three worked to administer infant mental health policy initiatives to support mental health professionals and clinicians, families, and young children (Zero to Three, 2021).

Van der Kolk (2014) pointed out that, unfortunately, the educational system and several methods that proclaim to care for trauma are inclined to circumvent the emotional-engagement system and concentrate instead on employing the brain's cognitive capacities. According to Lieberman et al. (2011), the impact of trauma on infant mental health problems is starting to be focused on the care strategies that demonstrate practical evidence of the effectiveness, such as cognitive behavioral therapy and child–parent psychotherapy.

Zeanah and Zeanah (2019) identified multiple challenges in providing infant mental health services. The challenges include the shortage of trained therapists and professionals, the lack of optimal continuity of service to focus on the complete spectrum of needs of young children and their families, and funding. The parent, as the primary caregiver, and the young child are the primary recipients of professional and community mental health efforts. Early care and education personnel are viewed as allied professionals and not central to the plans (Zero to Three, 2021). According to Van der Kolk (2014), social support is not a choice. It is biologically essential, and this truth should be the foundation of all prevention and treatment.

# Summary

Chapter 2 provided an overview of literature related to the definition of ECCE; the history and development of ECCE; the description of the ECCE workforce and programs; the definition of quality ECCE; professional development for ECCE professionals; young children's social and emotional development; a description of trauma experiences in young children; the history and development of mental health services; and mental health services for young children. Chapter 3 describes the methods used to conduct the current study. This chapter includes the research design, setting, sampling procedures, instrument, data collection procedures, data analysis and synthesis, reliability and trustworthiness, researcher's role, and study limitations.

# Chapter 3

# Methods

Early childhood educators regularly encounter young children who have experienced trauma. The purpose of this qualitative phenomenological study was to examine ECCE teacher perceptions of how trauma impacts young children in ECCE center-based programs and the adequacy of the training ECCE teachers receive to work with young children who have experienced trauma. Four purposes guided the current study. The first purpose was to investigate ECCE teachers' perceptions about how trauma has impacted young children in early childhood center-based programs. A second purpose was to explore ECCE teachers' perceptions about the information and skills needed to meet the needs of young children who have experienced trauma. The third purpose was to identify ECCE teachers' perceptions about professional development opportunities available to them that provide preparation to meet the needs of young children who have experienced trauma. The fourth purpose was to examine ECCE teachers' perceptions about challenges they face in accessing professional development to aid them in working with young children who have experienced trauma. Chapter 3 describes the research design, setting, sampling procedures, instrument, data collection procedures, data analysis and synthesis, reliability and trustworthiness, researcher's role, and limitations of the current study.

### **Research Design**

A qualitative phenomenological research design was selected for this study. In qualitative studies, the researcher "is considered by many within the field of qualitative inquiry to be the primary *instrument* of the endeavor" (Saldana, 2011, p. 22).

Creswell (2014) described qualitative research as "an approach for exploring and understanding the meaning individuals or groups ascribe to a social or human problem" (p. 4). According to Bloomberg and Volpe (2012), phenomenological research occurs within natural contexts. Therefore, real-world situations are studied as they naturally unfold (Bloomberg & Volpe, 2012). The phenomenon investigated in the current study was the perceptions of ECCE teachers about how trauma impacts young children in ECCE center-based programs and the adequacy and accessibility of training by ECCE teachers in centered-based programs who are teaching young children who have experienced trauma.

# Setting

The setting for this qualitative study was center-based ECCE programs serving young children in Johnson County, Kansas. The NAEYC (1993) noted,

Early childhood professionals who work directly with young children typically are employed in a variety of settings, including public schools; part-day and fullday centers, whether for-profit or nonprofit; public and private prekindergarten programs, including Head Start; before- and after-school programs; and family child care. (p. 9)

Center-based early learning programs in Johnson County, Kansas were selected as the current study setting because of the availability of numerous ECCE programs located in that geographic region of the state. In addition, center-based programs must meet the same Kansas Department of Health and Environment licensing requirements for teacher qualifications and professional development. The center-based ECCE classrooms in the current study were licensed to serve children birth to 5 ½ years of age.

### **Sampling Procedures**

The population for the current study included all early childhood teachers in the U.S. A purposive criterion sampling procedure was employed to identify the sample for the current study. "The logic of purposeful sampling lies in selecting information rich cases, with the objective of yielding insight and understanding of the phenomenon under investigation" (Bloomberg & Volpe, 2012, p. 104).

The researcher selected 10 participants to participate in the study using criterion sampling. Bloomberg and Volpe (2019) defined criterion sampling as a purposeful strategy in which participants are selected based on specific criteria identified by the researcher. The criteria for inclusion in the sample for the current study included individuals who had two or more years of experience working directly with young children and were employed in a center-based ECCE classroom in Johnson County, Kansas from January 2020 through December 2020. In addition, participants had contact with one or more young children in the center-based ECCE classroom who had experienced trauma.

### Instrument

An original semi-structured interview protocol was developed for the current study. The instrument was developed based upon the researcher's education and experience in ECCE and infant mental health. In addition, three subject matter experts (SMEs) also provided expertise in developing the instrument for the current study. One SME was an Early Care & Education Program Manager from the area child care resource and referral agency. The second SME was a Master Instructor of Conscious Discipline. The third SME was an early childhood faculty member at a university in the Kansas City region. The interview protocol included eight descriptive and demographic and 12 semistructured interview questions aligned with the research questions. The descriptive and demographic questions included the following:

1. If you are willing, please identify your age.

2. What is your highest level of education?

3. How long have you worked in the early childhood care and education field?

4. During the time you have worked in ECCE, what positions have you held?

5. How long have you been employed in your current ECCE teaching role?

6. What is the early childhood care and education program's licensed capacity where you currently work?

- a) 60 or fewer children
- b) 61 99 children
- c) 100 159 children

7. What is the age range of the children in your classroom?

8. Please describe other classrooms where you have worked and the age range of the children in those classrooms.

The semi-structured interview questions aligned with four research questions

included the following:

**RQ1.** What are the perceptions of ECCE teachers about how trauma has impacted young children in early childhood center-based programs?

**IQ9.** This research study looks at behaviors of young children who may have experienced trauma and the teachers' perceptions of being adequately prepared to work with those children who may have experienced trauma. How do you define trauma?

**IQ10.** What characteristics of a child who has experienced trauma in your classroom have you observed?

**IQ11.** How have the behaviors of the traumatized child impacted other children in your classroom or the early education program?

**IQ12.** How have the behaviors of the traumatized child impacted your ability to manage classroom experiences and learning activities?

**RQ2.** What information and skills do ECCE teachers perceive they need to meet the challenges and needs of the young children who have experienced trauma?

**IQ13.** What are the most significant challenges early childhood teachers face when working with young children who have experienced trauma?

**IQ14.** As you reflect on your experiences working with young children who express challenging or problem behaviors associated with trauma, what information do you need to provide quality early care and education?

**IQ15.** As you reflect on your experiences working with young children who express challenging or problem behaviors associated with trauma, what skills do you need to provide quality early care and education?

**RQ3.** What professional development opportunities do ECCE teachers perceive are available to them that provide knowledge and skills to work with young children who have experienced trauma?

**IQ16.** Please describe how you have gained information about young children's social-emotional development and the characteristics of young children who have experienced trauma.

**IQ17.** Please describe how you have gained information about instructional strategies for effectively teaching young children who have experienced trauma.

**IQ18.** How has the professional development you have participated in been delivered in-person, online, or in some other format?

**RQ4.** What challenges do early childhood teachers perceive they face in accessing professional development to aid them in working with young children who have experienced trauma?

**IQ19.** What challenges do you face in accessing professional development that would help you effectively teach young children who have experienced trauma?

**IQ20.** Do you have any additional thoughts or concerns regarding accessing professional development that would assist you in effectively teaching young children who have experienced trauma?

Two pilot interviews were conducted before using the interview protocol with study participants. The two individuals participating in the pilot interviews were teachers working in two different ECCE programs. The pilot interviews offered the researcher of the current study the opportunity to refine interview and observation techniques. As a result of the pilot interviews, an additional question was included in the interview that asked the participant to define the term trauma.

# **Data Collection Procedures**

Prior to conducting research, the researcher submitted a proposal to conduct the study to the Baker University Institutional Review Board on June 7, 2021. Approval to conduct the study was received on June 9, 2021 (see Appendix B). Once approval to conduct the study was received, the researcher recruited participants using professional

and social messaging boards (see Appendix C). Requests for posting the social messaging information (Appendix C) about the study were sent to administrators of Facebook groups: the Family Conservancy, Kansas Infant Toddler Mental Health, and ECCE center-based programs in Johnson County, Kansas. The social media posts were followed by individual emails to 40 ECCE center directors that included an invitation to participate (see Appendix D). In addition, packets of invitation to participate letters (see Appendix D) were personally delivered to center directors at 20 ECCE programs in Johnson County, Kansas, for distribution to program staff. The invitation to prospective interviewees (see Appendix D) provided an overview of the study. It was stated that participation would be voluntary and that there were no risks, discomfort, or compensation associated with the study, the option to withdraw from the study at any time or not answer any question without penalty, the assurance that the interviewee's anonymity and confidentiality would be maintained, the time required for the interview, and the opportunity to review the transcript for accuracy. Within 24 hours of an indication of interest in participating in the study, the researcher contacted the individual by phone or email to answer questions about the study and confirm participation in the study. Following the initial contact, the researcher sent an email to the willing participant, including a confirmation letter (see Appendix E), research questions, tentative date for the Zoom interview, and the consent form (Appendix F). The confirmation letter included similar information to that included in the invitation to participate (Appendix D). The consent form (see Appendix F) included the same information as that included in the invitation to participate (see Appendix D). In addition, the consent form (see Appendix E) explained the participant was providing

permission to audio-video record the interview and indicated the researcher would be taking notes during the interview. Using criterion sampling (Bloomberg & Volpe, 2012), 10 participants were selected to participate in interviews for the study.

The researcher asked each participant to sign and return a consent form (see Appendix F) prior to participation in the study. Twenty-four hours before the scheduled interview, the researcher sent a confirmation email containing the date and time and the instructions to access Zoom to each participant. The researcher used Zoom video conferencing to conduct the semi-structured interviews. Video conferencing was used because of the interviewees' geographic locations and to follow the Centers for Disease Control and Prevention (2020a) safety recommendation during the COVID-19 pandemic.

Rubin and Rubin (2005) indicated that in a qualitative research study the interviewee is not a test subject but rather is a partner. Valenzuela and Shrivastava (2002) described the general interview guide approach. According to these researchers, the interview should be designed to guarantee that the same broad range of data are collected from each interviewee. It offers greater focus than the conversational approach yet provides an amount of flexibility and versatility in capturing-the participant's information.

As the Zoom interview began, the researcher welcomed the participant and thanked her for participation in the current study. The researcher reiterated that the interviewee could withdraw from the study or indicate a desire not to answer any question during the interview. The participant was reminded that the consent form permitted the interview to be audio-video recorded and for notetaking to occur. Participants were reassured that their identity and the interview would be kept confidential using an anonymizing code (e.g., Participant 1, Participant 2, etc.) known only to the researcher.

According to Kvale and Brinkman (2009), the first few minutes of an interview are critical. A positive connection is initiated by attentive listening, with the interviewer demonstrating empathy, and respect for what the participant says. The interviewer is relaxed and is certain about what information he or she will seek. During the interview the researcher was aware of the pacing of the questions, her body position and breathing, and remained focused on the questions being asked.

The interview questions began with eight descriptive and demographic questions, which allowed the interviewees and researcher to engage in conversation and become comfortable sharing information. Descriptive and demographic questions were followed by 12 semi-structured interview protocol questions aligned with the research questions. All participants were asked the same interview protocol questions in the same order. Follow-up questions and probes were used as needed to encourage participants to be as thorough in their responses as possible. This procedure expedites interviews that can be efficiently analyzed and compared (McNamara, n.d). The researcher listened carefully, observed and took notes about nonverbal cues, attended to the conversation flow, asked probing questions as needed, and allowed the participants to speak without interrupting. When necessary, the researcher asked a follow-up question to fully understand the interviewee's response. At the conclusion of each interview, the researcher thanked the participant and ended the Zoom meeting. The researcher emailed a thank you note to the interviewee for their participation the day following the interview.

Interviews were scheduled from July 25 to October 5, 2021. Each interview lasted 30-45 minutes. Participant interviews were scheduled at the interviewee's convenience to provide adequate time for the Zoom meeting without distractions and interruptions. Once each interview was completed, the researcher assigned each recording an identification code known only to the researcher (e.g., Participant 1, Participant 2, etc.) to maintain confidentiality and protect anonymity. This identification code was also used to identify interview transcripts during the data analysis, the presentation of results, and the summary of results. Data were kept for five years on a secure drive accessible only to the researcher and then destroyed.

### **Data Analysis and Synthesis**

Qualitative data analysis is the mechanism of creating structure, organization, and significance to the volumes of data compiled (Bloomberg & Volpe, 2012). Rubin and Rubin (2005) described qualitative data analysis as the process of moving from raw interviews to evidence-based interpretations that are the groundwork for published documents. Creswell (2014) stated that analysis occurs in several overlapping stages. According to Creswell and Plano Clark (2011) qualitative data analysis entails coding the data, subdividing the text into phrases, sentences, or paragraphs, designating a label to each grouping, and then organizing the codes into themes. The researcher applied Creswell's (2014) five general steps for data analysis in qualitative research: (a) organize and prepare the data for analysis; (b) read or look at all the data; (c) code the data; (d) generate descriptions, categories, and themes; and (e) interpret the findings.

Applying Creswell's (2014) first step of data analysis, the researcher organized and prepared the data collected during each interview. The researcher reviewed each audio-video recording after each interview was completed to confirm that the recording was intact. The researcher labeled the interview and audio-video-recording with an identification code (e.g., Participant 1, Participant 2, etc.) to ensure anonymity. All data and notes the researcher had written during each interview were secured in the corresponding participant's file and stored chronologically.

The researcher initiated the transcription of the Zoom interviews by using the online app, Otter.ai (Fu & Sam Liang, 2020). The app provided efficient and precise documentation of the interview transcripts. First, each Zoom interview was uploaded to Otter.ai. After completing the Otter.ai processing, the researcher downloaded the interview transcript. Next, the researcher reviewed the copy while watching and listening to the recording to ensure that the document was transcribed verbatim. In addition, any errors or omissions were corrected.

Once the interview transcript was reviewed, the researcher emailed the appropriate transcript to each participant. Each participant was asked to review the transcript and note clarifications, corrections, omissions, and additional insights. The researcher instructed the participant to accept or return the amended transcript within a week. The participant's review of the transcript ensured the accuracy of the transcription. Bloomberg and Volpe (2012) referred to this process as member checking. After the participants' responses were received or the deadline had passed, the researcher highlighted any participant corrections to the transcript in a specific colored font.

The researcher in this study recognized that stages in data analyses overlap. Creswell (2014) included the use of computers as a step in data analysis. The use of computer software in qualitative analysis is now standard acceptable practice. It
increases the efficiency in handling large amounts of data and supports transparency in the research (Bloomberg & Volpe, 2019; Ho & Limpaecherm, 2020). It is important to note that software supports analysis, but the work remains in the investigator's hands (Bloomberg & Volpe, 2019; Creswell, 2014; Ho & Limpaecherm, 2020; Saldana, 2016).

To apply Creswell's (2014) second data analysis step, read or look at all the data, the researcher read each transcript several times to get an overall impression of the data. After the first readings, the researcher included impressions about the interviews in a research journal. Next, the researcher selected one transcript during the second step and contemplated the question, "What is this about?"(Creswell, 2014, p. 198). Finally, the researcher added notes related to the question in the research journal. Saldana (2016) referred to this process as preliminary jottings. These include anecdotal notes, phrases for codes, memos made on documents, transcripts, or written in a research journal which are all available for future reference.

The next step taken by the researcher of the current study was Creswell's (2014) third data analysis step, coding the data. Coding requires methodically labeling concepts, events, themes, and topical markers so that the researcher can easily locate and review all of the data that relate to the same topic across all the study interviews (Creswell, (2014). According to Saldana (2016) coding is a cyclic undertaking. The researcher began coding with the initial reading of transcripts. Throughout the analysis, the researcher reexamined transcripts to refine, filter, or delete codes. The researcher utilized a streamlined code to theory model, moving from raw data to codes, categories and subcategories, themes, and assertions (Saldana, 2016).

Five transcripts were selected to begin the researcher's first efforts at coding. Initial codes were identified from the first grouping of transcripts. The researcher highlighted common words and phrases across all transcripts and made notes in the page margins. In the data analysis, the researcher transferred manually coded files to a Word Document file. Saldana (2016) recommended that when the researcher is satisfied with initial codes, the codes are moved to a digital file. The right-hand margin was expanded to provide room for coding and researcher comments. Different colored fonts were used to denote commonalities, differences, and unique responses provided by interviewees. The initial codes from the first grouping of transcripts were applied to the second grouping. The researcher found that additional codes emerged. The researcher drafted a codebook added to the researcher's journal from these codes.

Continuing the coding process, the researcher designated the most meaningful wording for the codes and transferred the terms to categories. Next, the researcher merged codes and categories to form themes. This action reduced the entire list of categories. Next, the researcher created abbreviations for each category, and the topics were alphabetized. The researcher then generated a comprehensive list of the codes and categories.

The researcher then conducted a preliminary analysis of the data. The purposes of initial data analysis are to refine the data and organize it for further analysis, define the major characteristics of the data, and summarize the findings (Blischke, Rezaul, & Prabhakar Murthy, 2011). Analyses entailed classifying, comparing, weighing, and combining interview material to glean the intention to reveal patterns or intertwine descriptions of events into a meaningful narrative (Rubin & Rubin, 2005). During this

step, the researcher of the current study drafted figures that illustrated the relationships between the categories across all participants. Next, the researcher created multiple spreadsheets that detailed all keywords and phrases for a specific research question for each respondent to be viewed. The researcher examined the composite of participants' responses to identify common themes, Creswell's (2014) fourth step. With preliminary analysis completed, the research proceeded with the final step of data analysis, interpreting the findings (Creswell, 2014). According to Creswell and Plano Clark (2011) the interpretation of findings entails carefully contemplating specific results and expanding the context in view of the study problems, research questions, the extant literature, and possibly personal history. The researcher recorded personal insights and reflections in a research journal throughout the current study. In addition, the descriptions of how the coding process evolved and themes were developed were documented, providing an audit trail (Bloomberg & Volpe, 2019). The researcher conducted a coding process audit to affirm the accuracy and credibility of the findings for this study. The researcher enlisted one external reviewer to review audio-video recordings and transcripts for accuracy and coding for common themes. The external reviewer holds a doctorate in Educational Leadership in Higher Education and conducted a qualitative research study to complete doctoral requirements. The external reviewer concurred with the themes identified by the researcher conducting the current study.

### **Reliability and Trustworthiness**

Creswell (2014) indicated that reliability and trustworthiness could be established as the researcher reviews the accuracy of the results by implementing distinct protocols. The criteria identified by Lincoln and Guba (1985) to establish trustworthiness are accepted and recognized by many qualitative researchers. These criteria include credibility, dependability, confirmability, and transferability (Bloomberg & Volpe, 2014; Creswell, 2014; Creswell & Plano Clark, 2011; Lincoln & Guba, 1985). Bloomberg and Volpe (2014) described trustworthiness criteria with corresponding research strategies. The researcher used five techniques to ensure the reliability and trustworthiness of the findings in the current study: peer examination of the interview protocol, two pilot interviews, member checking, journaling or narrative analysis, and an audit trail.

During the interview process the significance of the research's integrity is intensified because the interviewer is the primary instrument for gaining information (Kvale & Brinkman, 2009). Peer review of the interview questions was conducted by the researcher's major advisor, research analyst, and subject matter experts (SME) to ensure alignment with the purposes of the study, clarity, and determination that the questions were mutually exclusive and exhaustive. Three subject matter experts (SMEs) provided expertise in developing the instrument for the current study. The SMEs were asked to comment on the language of the interview questions, the understandability of the wording, and the relationship of the interview questions to the research questions. As a result, two changes were made in the wording of interview questions. The researcher conducted two pilot interviews to rehearse and ascertain the interview process's efficacy and questions and improve the researcher's interviewing skills.

After the interviews were conducted, the researcher used member checking to allow study participants to verify the truthfulness and accuracy of interview transcripts. Member checking is a method in which the investigator sends the narrative back to the interview participants requesting they review the information for accurate documentation of their experiences (Creswell & Plano Clark, 2011). Participants reported that the transcriptions were accurate.

According to Creswell (2014) the function of the researcher as the principal data collection instrument requires the identification of personal beliefs, biases, and assumptions at the beginning of the research. Therefore, the researcher chose to journal and write a narrative analysis to address those matters. The narrative analysis included the researcher's description of the research process, synopsis of interview notes, explanation of the coding process, and interpretation of the results of the qualitative research. In addition, the narrative analysis incorporated personal reflections of what the researcher learned and interpretation of the research findings. Throughout all aspects of this study, the researcher engaged in self-reflection, noting impressions, and recording activities in a research journal.

The researcher conducted an audit trail to affirm the accuracy and credibility of the findings for this study. The researcher enlisted one external reviewer to review audio-video recordings and transcripts for accuracy and coding for common themes. According to Creswell (2014) the overall validity of qualitative research is strengthened when a nonpartisan examiner reviews multiple aspects of the study such as the veracity of the transcription, the links between the research questions and the data, the depth of analysis from the raw data through interpretation.

#### **Researcher's Role**

The researcher is the data-gathering instrument. The investigator's ability to observe, listen, and understand are crucial in the study's exploration (Rubin & Rubin,

2005). According to Creswell (2014), the researcher is also responsible for collecting and analyzing data obtained in a qualitative study.

The researcher's role in the current qualitative study was to produce unbiased research. Throughout qualitative research attention must be paid to identify possible prejudices that might exist in the design, implementation, and analysis of the study (Bloomberg & Volpe, 2012). The researcher in the current study acknowledged that biases might have been brought to this study due to her role as an administrator, educator, and consultant in ECCE programs. In addition, the researcher knew and had encountered participants who have worked with young children with challenging behaviors or experienced trauma.

The researcher for this study had experienced educational preparation at the University of Kansas. She earned a Master's Degree in Human Development and Family Life, with an emphasis in ECCE. The researcher was employed as a director and teacher in various ECCE programs and, at the time of the current study, was employed in ECCE as a consultant, educator, and substitute teacher across the Kansas City metropolitan area. The researcher was a foster parent and was the grandmother of an adopted young child who had experienced trauma. The researcher had the experience and background to have a solid foundation to discern the educational needs and challenges of early childhood teachers working with young children who have experienced trauma.

It was imperative to the researcher that the study be credible and trustworthy. These elements and roles required the investigator to recognize the potential for bias and continually exercise awareness, intentionality, and objectivity to maintain the research integrity throughout the study. The researcher of the current study kept a research journal

68

that provided a narrative of analysis of critical steps and insights that occurred throughout the study. The researcher employed attentive listening, careful observation, conscientious notetaking, and objective reflection on participant responses during the data collection phase. The interviews were recorded and transcribed. Member checking and independent audits of the results from interview transcripts assured the integrity of the data analysis. Throughout the current study, the researcher engaged in the process of reflexivity. Reflexivity necessitates the researcher to contemplate and clearly state their point of view, biases, and perspective so that readers are better able to comprehend the filters through which questions were asked, data were collected and analyzed, and results were reported (Sutton & Austin, 2015).

### Limitations

Limitations are elements that may influence the analysis of the research or the determination of the outcome (Lunenburg & Irby, 2008). This study had the following limitations:

- The interview participants may not have provided a representative sample of ECCE teachers resulting in limited validity.
- 2. The researcher did not verify the training opportunities described by the interview participants. As a result, recollections about training related to trauma in young children may have been inaccurate.
- 3. All participants were encouraged to be forthcoming and to go into detail when answering questions. However, Creswell (2014) stated, "Not all people are equally articulate and perceptive" (p. 190). Some participants may not have

clearly understood the interview questions or been able to articulate their responses verbally.

# **Summary**

Using a qualitative phenomenological research design, the current study explored the perceptions of ECCE teachers about how trauma has impacted young children in early childhood centered-based programs and the adequacy of preparation to teach young children who have experienced trauma. Chapter 3 described the qualitative methodology employed in this research. The research design, setting, sampling procedures, instrument, data collection procedures, data collection procedures, data analysis and synthesis, reliability and trustworthiness, researcher's role, and limitations were explained. Chapter 4 presents the results of the data analysis.

### Chapter 4

## Results

This qualitative study was conducted to explore ECCE teachers' perceptions of how trauma impacts young children in ECCE-center-based programs and the adequacy of the training ECCE teachers receive to work with young children who have experienced trauma. Four purposes guided the current study. The first purpose was to investigate ECCE teachers' perceptions about how trauma has impacted young children in early childhood center-based programs. A second purpose was to explore ECCE teachers' perceptions about the information and skills needed to meet the needs of young children who have experienced trauma. The third purpose was to identify ECCE teachers' perceptions about professional development opportunities available to them that provide preparation to meet the needs of young children who have experienced trauma. The fourth purpose was to examine the perceptions of ECCE teachers about challenges they face in accessing professional development to aid them in working with young children who have experienced trauma. Chapter 4 includes the results of the qualitative analysis of transcripts of interviews with 10 early childhood teachers.

## **Descriptive Demographics and Participant Background**

Eight demographic questions were asked to obtain an understanding of the backgrounds of the participants. Ten ECCE teachers employed in early childhood centerbased programs in Johnson County, Kansas, with two or more years of experience working directly with young children who may have experienced trauma participated in the study. Each participant engaged in a face-to-face Zoom interview. The interview responses of the 10 participants were analyzed for this study. All 10 study participants were teachers in KDHE licensed early care and education centers in Johnson County, Kansas. Five of the respondents had worked as float or assistant teachers before taking on the responsibilities of a teacher. Participant ages ranged from 23 to 59 years of age. One participant was younger than 25. Six study respondents were in their late twenties and thirties. Three of the teachers were 45 years of age or older. All 10 participants had graduated from high school. Four of the teachers had attended some college, and two of them were working on their Child Development Associate (CDA) credentials when they were interviewed. Four of the participants had earned college degrees, two held associate degrees, one had obtained a bachelor's degree in elementary education, and one had acquired a master's degree in psychology.

Study respondents had been employed in their current teaching role for 6 months to 11 years. One participant who had recently changed jobs, was previously employed in a child development center for 3 years. Four participants worked for 1 to 4 years in the program where they were currently employed. Four teachers had been at the present site for 5 to 6 years. One participant had worked in the same center for 11 years.

Eight participants were working in centers with a licensed capacity for 100 to 159 children. One participant was employed in a program with 60 to 99 children. One participant was teaching in a center with more than 160 children.

At the time the interviews were conducted, three teachers worked with infants and seven teachers worked in preschool classrooms as defined by KDHE. While employed in the ECCE field, eight participants had experience working with children in multiple age groups from infancy through school age. Participant 8 worked with infants and toddlers only. Participant 1 worked only with preschoolers, including children with special needs. The respondents reported between 3 to 5 years of teaching experience in early care and education. The average length of time of service was 15.5 years. Collectively, the interview participants have nurtured and educated young children for a total of 154.5 years.

Participants were asked 12 interview questions organized around four research questions. Four major themes were identified from the analysis of the data: teachers' perceptions about trauma in early care and education; teachers' professional development needs working with young children who may have experienced trauma; resources and professional development opportunities available to ECCE teachers; and challenges working with young children who may have experienced trauma. The following sections provide a summary of the data and an explanation of the findings with regard to the four themes. The themes generated from responses to interview questions are identified. Direct quotations are included to highlight and demonstrate interview participants' perceptions about their experiences working in ECCE classrooms with young children who may have experienced trauma and the adequacy and accessibility of training to work with young children who have experienced trauma.

### **Teachers' Perceptions About Trauma in Early Care and Education**

Study participants were asked to describe their classroom experiences in the ECCE center with young children who may have experienced trauma. Within this theme, four subthemes emerged in participants' responses: a definition of trauma, the behaviors of children who have experienced trauma, the responses of other children to a child who has experienced trauma, and teachers' responses in the classroom to the behaviors of a child who has experienced trauma. The following four sections report the participants' comments related to each of these four subthemes.

**Teachers' definition of trauma.** Study participants were asked to provide their definition of or understanding of trauma. All respondents agreed that trauma is an event that impacts a child's future experiences. Two respondents mentioned the uniqueness of trauma. Participant 1 stated, "It has to be defined by the child, how they respond to whatever is happening then in their world." Participant 4 stated, "Trauma differs. What an individual perceives is completely different and unique."

Three respondents described trauma as a disruption in a young child's life. Participant 4 noted, "[Trauma is] something that has happened that's big enough to disrupt their feelings in the way that they see the world." Participant 5 said, "[Trauma is] something that has caused them in their past to change their behaviors and change their actions." Participant 8 stated, "I see trauma as something that emotionally impacts a child for a longer period of time. If it continues to affect them, then they were traumatized."

Three participants identified trauma as a negative experience. Participant 2 said, "[Trauma is] an experience that has impacted a child, usually in a negative way, and has some effects on their learning or their abilities, or just their overall academic experience." Participant 10 remarked, "Trauma is "something that affects a child in a negative way, a hurtful way." Participant 6 commented, "Trauma is something that sets a core memory. So, I would define it as a negative core experience."

Teachers offered numerous examples of traumatic events that children may experience. Participant 1 stated, "Trauma can be different for different people." When offering examples of trauma, Participant I said, "[It's] kids that have been abused and neglected." Participant 1 noted that children of divorce "may just not have had their basic needs met. They may be bounced back and forth between mom and dad in fights." Participant 3 shared a personal experience,

Trauma is a family dynamic, divorce, or even military life where a spouse has to leave for a long period of time. My kids don't understand that. So, that's traumatizing for them. Even moving suddenly and not being prepared could be traumatizing.

Participant 7 stated, "Traumatic events may range from mom and dad not being together or a death in the family."

Two teachers identified the current time we are living in as trauma. Participant 1 declared, "We're all living in trauma now with a pandemic." Participant 9 pointed out,

With COVID the children have to start wearing masks and that disrupted their routine for the past four years of never having to wear a mask, and now all of a sudden, having to wear a mask. So that's trauma for them.

**Behaviors of children who have experienced trauma.** All participants offered examples of behaviors exhibited by children when asked what characteristics they had observed in a child in their classroom who has experienced trauma. Respondents described a range of behaviors observed in children who may have experienced trauma. For example, Participant 2 said, "Sometimes it can be those fight or flight responses, depending on the type of trauma." She went on to say, "There's a continuum. They could be either shy and not really verbal, and not really wanting to express anything, or they could be super aggressive to peers or themselves or their teacher." Participant 4 remarked, "They just shy away from the room, they don't want to do anything, and they're mostly in their safe space half the time that they're in the classroom." Participant 3 shared her thoughts, "I think the biggest thing that I have observed is their socialemotional skills are just not there. They struggle with simple things." She continued, "I've also had a kiddo that had severe attachment issues and actually formed an attachment to me to the point where he always had to be next to me, or he didn't feel safe."

Participant 8 reported,

It's different for each kid. When certain kids are having meltdowns, they need you to leave them alone for a minute, but some of them may need you to hold them and hug them. When they're already upset, it's really hard to find out from them what they need.

Participant 10 said, "[There] may be some anxiety separation issues." Participant 7 offered a personal experience,

I had a little girl who was in foster care. Her mom was on a drug bender. And the little girl was with her at the time in the hotel. She was very attached to certain people in the classroom. So, if somebody new walked in, she was very skittish. She didn't know exactly what to do. If I would walk out of the room, she would scream, she would throw things. She was not happy.

Participant 6 recalled, "I had a little girl who was extremely anxious. Anytime that she felt anxious, she would pull out her hair. She would get so nervous, like when she had to leave her mom, she would pull her hair." Participant 5 shared a personal experience, "I had a student last year that was sold by her mother into prostitution at age four. When

she came back, she had scars and was very quiet and withdrawn, her behavior changed because of her trauma." Participant 4 reported,

It seems to either go one of two ways. They [traumatized children] withdraw and do not-engage with anybody, students or teachers, or become aggressive. They seem to lash out a lot at everyone around them and are very, very quick to anger. Participant 9 stated, "I would have to say the children act out. They don't listen. They decide to be defiant, throw things at the teachers, and hit other students. Things like that." Participant 8 described a child's behavior, "He would just go off the handle. Actually, we ended up having to ask him to leave the center because he trashed one of our classrooms breaking shelves during one of his tantrums." Participant 10 recalled, "When I was in Head Start there were a lot of anger issues. Kids hurting other kids and throwing computers and tipping over furniture. Just the anger issues because they're so angry inside." Participant 1 commented, "They may hit another child. They may just not know how to ask for attention appropriately."

Three participants reported they had seen regression in children's development. Regression was seen most often in toileting skills. Participant 4 stated, "A big one that I've seen is regression in a lot of things, so if we're potty training, the child regresses back to having multiple accidents." Participant 3 noted, "Kids also sometimes tend to have more accidents, potty accidents, when they experience trauma depending on what type of trauma it is." Participant 8 identified behaviors of two children who had experienced trauma. "Both of them were four at the time and were still in pull-ups due to their situations. One of them wouldn't go in the bathroom alone without screaming." Six teachers described food and eating issues demonstrated by children who may have experienced trauma. Participant 9 mentioned, "It might affect their eating habits." Participant 5 explained, "If they've been denied food and they start to take food in their pockets or stash it away in their cubby, they've obviously had some kind of a traumatic event to cause them to feel they need to store food." Participant 4 reported, "They also have eating problems. Sometimes they won't eat, or sometimes they eat really fast." Participant 2 commented, "Someone who was neglected with food or something like that might be overly hungry or think they're overly hungry and [they] gravitate [to] and pull things out of the trash or steal people's food." Participant 1 reported she had "seen kids that have double-fisted food, telling me that they may not be getting their needs met for meals at home." Participant 7 described a child's response, "When it was time to eat, she would hurry up and scarf everything down as if it was going to be her last meal."

One teacher shared a child's behavior that the other respondents had not noted. Participant 6 recalled an experience of a child in her care,

His parents were divorced. They went through the divorce when he was younger. He would dress up, like princess gowns and everything, but that was him acting out his family. I saw that as him just projecting how he wanted his mother or his father to be around when he was at the other home. So, if he was at his dad's house, he would want to wear wigs that week. But if he was at his mom's house, he wanted to bring his tractors to school.

She continued, "It was strange to see how he would flip from week to week. It was strange to see how the dynamics flipped when he was switching homes."

Two participants reported children were expelled from the early care and education program due to behaviors that may have been impacted by trauma. For example, Participant 2 commented, "Sometimes those kids might get kicked out of one program because the teachers or staff there don't know how to or can't work with them. It was too challenging for them." Participant 8 shared the example of the little boy trashing a classroom multiple times and then being asked to leave. She explained, "We were unable to help him in our facility due to not having the proper tools and training to deal with someone of that nature."

**Responses of other children to a child who has experienced trauma.** All respondents shared multiple ways the behaviors of a child who may have experienced trauma impacted the other children in the ECCE program. Teachers described occasions when classmates avoided the child or were frightened by the behaviors displayed, or engaged in similar behaviors. Participant 2 pointed out, "It can be a safety concern for other children." Participant 4 stated, "When those behaviors arise it complicates the mood, the tone, [the] activities and such in the classroom."

Five respondents recognized that other children avoided the child with the challenging behaviors in the classroom. Participant 3 reported, "Some children don't seem to be bothered by it. But I think that they just kind of shy away from it." Participant 1 recalled,

I feel like in our little center, I've noticed other kids will sort of avoid them, especially if they're lashing out with behavior, or shyness might be another one where they kind of just go inward and keep to themselves. The other kids tend to not go toward them socially. I feel like they're already at a deficit. Participant 9 stated, "The other children would be more hesitant to want to play with those kids or to want to go into that classroom knowing that that student will be in there." Participant 5 shared,

The aggressive students they will avoid. Those older kiddos really remember, 'Oh, you hit me with the block yesterday, and I don't want to play with you today.' And two-year-olds obviously would let it go, water under the bridge, new day. Not my kiddos. They figure out who they want to play with and who they don't really quickly.

Two teachers stated that children were frightened by the challenging behaviors displayed by a child that may have experienced trauma. Participant 8 recalled, "I did have children telling me that they were scared when the child was screaming." Participant 3 reported, "Sometimes when kiddos are crying a lot, the other kids in the classroom sometimes get scared and cover their ears and ask "Why is this happening? Why is so and so, so upset?"

Three respondents described the feelings of the other children in the classroom. Participant 8 stated,

You're focusing on that child that needs you in that moment. And then these other kids, they feel like they feel ignored because they're trying to come up to you and have conversations with you during all of this. You literally have to say, 'I can't do this right now.'

Participant 4 remarked, "I'd say that they just require a lot more of the teachers who are present. I would say the other kids don't get the needed focus." The teacher continued, "It could also harbor bad feelings among the students about this child, if it's like aggressive stuff, or if they see the teacher is favoring this kid a little more or something similar to that." Participant 8 noted,

They just seem to not understand that other children don't deal with things the same way they do. It makes it really hard to communicate with them. 'I'm sorry, I know you need me right now. But they need me more right now.'

Two participants reported that other children might engage in similar behaviors exhibited by the children who may have experienced trauma. Participant 3 offered her opinion, "Other kids, I think, seek attention from the teacher more because they feel like that child is getting more attention obviously because that child is struggling." Participant 3 commented, "Honestly, one of the biggest things I think I've experienced is there's a lot of aggression in that child, then you have some behaviors that seem to start to be difficult with other kids because they're learning that behavior."

Two respondents reported that other children in the classroom made an effort to engage the child who may have experienced trauma. Participant 5 said, "Depending on which behavior they're exhibiting, children will try to bring the quieter students into play and encourage them to join them." Participant 6 described a situation in her classroom where a child moved between his divorced parents' households, and he would dress in clothing of the parent he was not living with that week. She revealed, "It was actually just kind of positive because the other little boys were like, "Oh, well, we can dress up and be whatever we want to be too." She went on to say,

They were like, 'We could just all play together. It doesn't matter if I dress up in a princess dress, because I like it. Or if we play with blocks doesn't matter.' So that was kind of fun, just to see the gender roles kind of melt away. And everybody just embraced that fun learning experience.

Participant 6 offered an additional positive experience among the children in her classroom,

With the little girl, her friends realize, 'Oh, she's having a hard time, we need to give her space.' In that class, there were a lot of little girls. So, it was very nurturing, very calming, 'We just need to give her space. All she needs is some time away.'

**Teachers' responses in the classroom to the behaviors of a child who has experienced trauma.** All participants described how the behaviors of a child who has experienced trauma impacted their ability to manage classroom experiences and learning activities. Four study participants disclosed teachable moments. Participant 1 reflected,

It can certainly be challenging. I guess for me it's kind of a trial and error. When you get kids at the beginning of the year, you don't always know their background or their history. And I may try to set those boundaries right away and realize, well, that's making it worse. So, then I have to rethink and adjust, maybe giving them a little bit more positive feedback, a little gentle touch here and there.

Whatever it takes to find what works for that child.

Three teachers talked about offering children reassurance and feelings of being safe. Participant 10 stated,

I would say, me personally learning to help them to feel safe and let the whole class know. Include the whole class in not singling the child out. But to talk to everybody about the issue. Like it was a class issue, not just the child's issue. Participant 5 fondly recalled,

I tell them every day that they are safe and it's my job to keep them safe. And I need them to help me do that, and they tell their parents that when they go home and their parents come back and say, 'so, it's your job to keep them safe.' Yup, it's my job. I teach them too, I promise. Gotta keep them safe first and foremost.

Participant 7 shared,

A lot of children don't fully understand some of the emotions. So that's why we do emotion cards and talk about emotions with the children. When a child is having one of those moments, we sit with them and talk to them. And also, let the other children in the classroom know that we don't do certain things like this. And if we need to have a moment, we need to breathe, and that's going to be okay.

Six teachers described the adjustments that may be made in the classroom routine and activities when working with a child who may have experienced trauma. Respondents described a balancing act between the child needing one-on-one attention and engaging the rest of the class. Participant 1 said,

It would depend on whatever the needs of that child are, and then trying to be creative. A lot of times, it's a child that maybe can't focus in the group. So maybe name it with a fidget toy or something like that. I give the directions to the whole group, and then I go and give them their own direction, just to make sure you know, eye contact, closeness, just to make sure that they've understood what I've expected of them. Participant 2 stated,

Yeah, that can be hard because depending on what it is, you may need to give more attention to that child or really stop whatever you're doing, leading the classroom or teaching in the classroom, to work through this with this child because you don't want to just leave them when they are having a meltdown or whatever the case may be.

Teachers described modifications in the schedule or planned activities. Participant 4 explained,

I just don't have time following our schedule that we normally follow because B is having trouble. And they need more attention from the teacher who wants to help them through their situation, and they don't want to make it worse. So instead of having to focus on this one small group that may or may not be working at the time, I usually just fall back to having them all do their centers, and that way, I can work through each child individually as the other ones are more occupied doing what they would like to do at that time.

Participant 8 recalled,

But there's also days where I had to cancel circle time because it was too intense of a moment. I couldn't focus on teaching all the kids and helping one student who is acting out at the same time. It disrupts everything so much that you can't always get back to it. You have to move forward and move on with your day.

Participant 9 reiterated,

It makes it a little bit harder to work with the other children when you have to focus one-on-one on that child who's acting out. The other children get left to do

whatever they want instead of actually teaching them their letters or how to write their names and stuff because you now have to focus on one child who is throwing things at the teachers. You have to be one on one with that child. Participant 8 summed up classroom activities, "I mean, there are good days and bad days. There have been days where I was able to have my normal class."

Two participants expressed their frustrations with being the only teacher in the classroom when challenging behaviors occurred. Participant 3 disclosed,

Honestly, it makes it very difficult. You need an extra set of hands. You cannot teach all of your children and get your routine done and everybody gets the attention and education that you want to give them when you are constantly having to help this child that has experienced trauma that cannot function more than 10 - 15 minutes at a time some days.

Participant 3 further stated,

And we also need to be listened to when we say that we need help, meaning that when two teachers cannot do a room of 12 kids when you have two children who have trauma. With the ratios, you think you should be able to handle that. But it's not feasible because both of you are still dealing with the children that have trauma. And neither one of you can teach.

Participant 4 replied, "I feel like we have too many kids in the classroom and too few teachers to have to deal with the behaviors that arise. So, I definitely feel like maybe I need extra help managing the classroom." Participant 2 stated, "They [teachers] really require some additional hands in the classroom. So that way, we can keep the rest of the group going while we're working through this issue."

Two participants identified the possibility of children leaving the classroom or the ECCE program. Participant 2 reported, "They [the child] may be pulled out of the classroom for some special services to help work through that trauma." She continued describing the alternative,

Sometimes those kids might get kicked out of one program because the teachers there, the staff there, don't know how to or can't work with them. You know, it was too challenging for them. So, then they get bounced to another place, and it is just kind of a cycle that goes down. So, trying to break that cycle in early childhood is a challenge. And obviously, not one person can be the person that solves it by themselves.

Two respondents recalled there was little change in the classroom experiences. Participant 6 commented, "It was never really crazy things to manage." Participant 7 explained,

I wouldn't say it impacted me or didn't impact me. It was kind of the opposite. They were willing to do activities in the classroom with the other kids. They didn't stray away from the class of what we were doing. They were very adamant if the other kids in the class are having fun, they want to have fun as well and learn and use their fine motors and working on pincer grasp and everything like that in between.

## **Professional Development Needs of ECCE Teachers**

During the interviews participants were asked what information and skills they perceived they needed to work with young children who may have experienced trauma. All 10 respondents identified multiple needs. Within this theme, three subthemes were identified in participants' responses: information needed by ECCE teachers, skills needed by ECCE teachers, and attributes desired for ECCE teachers. Each of the subthemes is described in the next sections.

**Information needed by ECCE teachers.** Five participants described a need for general information about trauma. Participant 10 offered, "I just think it's [trauma] something that every teacher should maybe be trained on. At least get an insight or a little taste of some training for it and how to recognize it." Participant 7 advocated,

I believe that all teachers in any child care should have a trauma class that they attend so they know how to deal with children who have had any sort of trauma in their life, from mom and dad not being together or a death in the family.

Participant 3 commented, "I think the biggest thing about children that do have trauma is understanding what their trauma is. Once we understand what they have gone through, then we can have the empathy that we need to have as human beings." Participant 1 remarked,

That's a broad question. [We need] constant education as far as defining trauma. Some people don't recognize trauma as trauma. I think that we have a lot of that. I feel like we need a lot more knowledge of the ever-changing, and I know the pandemic is also still basically new.

Seven teachers wanted a child's background information regarding trauma and the triggers that may impact the child's behavior. Participant 5 reported, "The first thing that came to my mind is more information. If there's background information that we know, it'd be really nice to be able to have it from the get-go." Participant 5 continued,

And a lot of times, we don't have that information. Sometimes we get students, and we have no idea what they've been through. [It would] help us [if] they all came with the book, and you just read it, "Oh, look, their grandma died six months ago, [or] their uncle's really sick in the hospital right now.

Participant 9 acknowledged,

I would need to know a little bit about [the child's] family background. Did the parents lose their jobs? Are the parents working from home? Are the parents out? Are the parents traveling? So, information about their family and stuff to see what kind of care that I could [provide to] help them. Because if a child doesn't have their mom around all the time, then I would be able to give that child a little extra love and stuff. Cuz not everybody has a normal mom and dad all the time at home. Some children only have a mom. We got one child whose father was not in the picture and then all of a sudden came back in the picture, and now that child started acting up. We'd like to know these kinds of things so we could be able to prepare ourselves for the way the child may behave.

Participant 2 commented,

I always like to have as much information as I'm able to have. It really helps you to understand more clearly what's going on and what might be triggers or how to help. So, the more information, the better. Obviously, sometimes you don't get all of that [information]. And then you just have to work with what you can get. Participant 10 remarked,

Learning how to help them cope with what they've been through. Recognizing the signs of what they might need, whether it be something to hold on to, help

them feel safe, or just some hugs. Learning, just learning through these classes. Participant 8 said, "[I need] communication between parents or foster parents or grandparent, whatever the situation may be. Just information and communication with them would definitely help." Participant 7 stated, "There are certain children that have struggled with more traumatic experiences than other children. I think that we, as individuals, as teachers, we need to know what and how to resolve an issue." Participant 3 indicated,

If the kiddo is on an IEP having that information as teachers in a childcare setting is the number one thing if their trauma is related through that. Because if we don't know what's going on at school and why they are on that individual education plan, then we can't help them when they're not at school. So being in those meetings, being in contact with those teachers is a huge part of the communication that needs to happen. And it really fails to happen a lot of the time.

**Skills needed by ECCE teachers.** Respondents were asked to reflect on the skills needed to provide quality ECCE while working with young children who have expressed challenging or problem behaviors that may be associated with trauma. Among the skills participants identified were communication techniques and strategies to aid in calming and self-regulation for the child and themselves. Participant 1 stated, "You can have the information, but you have to be able to apply it." The interviewee went on to say, "If you

have an educator in there that can kind of give us guidance on how to respond when a kid does different behaviors. So, you can actually see it, match it." Participant 9 responded, "Just trial and error. Just trying to figure out what works best. And if that doesn't work, then try something else." Respondent 9 noted, "We're starting to learn different ways to help kids that are in trauma. So that would be a really good skill too, so I can know other ways to help people."

Five teachers specifically mentioned skills included in the practices of Conscious Discipline<sup>®</sup>. Participant 8 admitted,

I need to learn, personally, myself, how to help children get out of their brainstem when they're in that fight or flight instinct when their defenses are up and you just can't get through to them. I need to learn how to refocus them. And I need to find ways to get them out of that moment.

Participant 3 emphasized,

Definitely that's where my Conscious Discipline<sup>®</sup> comes in to affect being able to keep myself calm and to help that child calm themselves and get back and in their brainstem and not be in that fight or flight mode. I think it is a really big tool, a skill, but I think everyone needs it. Because if we can't stay calm and not be frustrated with the situation, then we can't help that child.

Participant 4 commented,

I'm not sure how to explain it. But I would like the things that they teach in Conscious Discipline. I'm not 100% sure what they're called, but the steps that they take to kind of get inside the child's head in how they're reacting and how they're behaving. I don't think it's empathy but more something along the lines of that.

Participant 5 described the skills needed in her classroom,

Oh my gosh! Conscious Discipline<sup>®</sup> is valuable here with the I Love You rituals to build those connections. And the breathing. I cannot tell you how many times people will hear me down the hall saying smell the flower, blow out the candle. If there's a blow-up, smell the flower and blow out the candle. And then they're doing that as they hear me say it. I know it works really great to calm yourself and the entire room, including myself. The Conscious Discipline<sup>®</sup> training has absolutely changed everything that I have done in my classroom in the last decade at least.

Participant 6 advocated,

I think you definitely need to learn some sort of Conscious Discipline<sup>®</sup> practice. I know I keep saying that a lot. Over quite a few years that I've been at **second**, we've had a lot of talk about Conscious Discipline<sup>®</sup>. And I think it's the best method of communication because it makes a child, I feel, makes them feel heard. So, I really think that those kind of practices are good. And making sure that you learn how to talk to a child, getting down on their level, using phrases that they understand, and making sure they understand what they're saying.

Two participants did not specifically mention Conscious Discipline<sup>®</sup> strategies but described similar skills and additional practices. Participant 10 commented,

Learning to remain calm is a big thing through any rough patches that one of these children might have. And learning how to help them cope with what they've been through. Recognizing the signs of what they might need, whether it be something to hold to help them feel safe or just some hugs. Learning, just learning through these classes.

Participant 7 remarked,

You need to set yourself up to a standard of learning more from other teachers. So, I did classes through KCCTO [Kansas Child Care Training Opportunities], and I learned a lot of information from them and doing my own research to have better skills of writing things down and taking notes if this happened again. I would know the time of day, and when it occurred, and what happened if there was another child involved. I'm just being extra cautious to certain things that might trigger a child. We never know if the child may have certain triggers or smells that might cause them to go back if they did have trauma. Just making sure that we have a sheet in our classroom that would remind us, this is what you need to do. This is how you need to calm down. And this is what we can work on with the other children in the classroom.

Attributes desired for ECCE teachers. All 10 study participants identified one or more attributes desired for teachers of young children who may have experienced trauma. Attributes included patience, self-confidence, listening skills, and the desire to be a leader in the ECCE field. Six out of 10 respondents acknowledged patience was an attribute required by ECCE teachers. Participant 9 recognized, "I'm definitely going to need to be patient for a child." She went on to say,

The teachers just need to be more patient and ask for help instead of trying to think that they can do it all on their own. Teachers definitely need to just ask for help and say, Hey, I'm not sure what I need to do to make this work. Can you help me?

Participant 4 commented, "I would say patience, for one. I feel like we have too many kids in the classroom and too many teachers to have to deal with the behaviors that arise." Participant 2 noted,

Patience and understanding even if you don't completely understand what they're going through. I guess it kind of wraps back into patience and sympathy or empathy - to try to relate to that kid and just let them know that you're there and caring. You have to care that they've went through something and that you need to work through it.

Participant 5 agreed with Participant 2, "Patience and empathy, you don't know what they've gone through. You don't know where they've been. You just have to be patient and give them a chance to trust you." Participant 7 replied, "I think you need to have patience." Participant 6 recalled,

Prior to being an early childhood education teacher, I was a retail manager for a very long time, which has made me acquire a lot of patience. So, I feel like as long as you have the education and patience, it's not easier by any means, but I feel more equipped, just having the education along with patience.

Two participants described self-confidence as an attribute needed by ECCE teachers. Participant 1 shared, "I think there's something to say about doing the same thing for 30 years. You feel a little more confident, and I know how to handle this. This is the problem; we can take care of it." Participant 4 stated, I guess self-confidence while they are dealing with the trauma. Sometimes they [teachers] might lack the confidence in doing and think that they're not doing the right thing or doubt themselves and how they're supposed to handle the situation and how they've handled other children in their care. So, I feel like that's just the greatest challenge of all, at least in my experience.

Two respondents noted that listening skills were attributes needed by teachers working with young children who may have experienced trauma. Participant 6 explained,

I think just listening to the kids, and if you think that something is happening, just listening and letting them know that you're there, and they're in a safe place. You're a person that they trust. In my household, we use the term 'trusted adults.' As a mother, I like to use the word 'trusted adult' so that they know that I'm somebody they can talk to.

Participant 10 identified several skills teachers might need, "Nurturing and listening skills. Just to be there for them, just to recognize the signs that they might need extra care. Empathy."

Two participants voiced a desire to be leader in the ECCE field. Participant 2 stated,

I have kind of, in the last few years, really learned how to be an advocate for them. And you know, not be afraid to push some buttons to ask them questions to try to get them the help that they need and that they deserve.

Participant 2 later added, "Being proactive. There's been some times that I've taken some online courses about things that I need to know more about." Finally, Participant 4

stated," I have also done my fair share of research in trauma care just because I enjoy doing research on how to better myself as a teacher."

## **Resources and Professional Development Available to ECCE Teachers**

Participants were asked to describe how they gained information and skills about social-emotional development and instructional strategies for effectively working with young children who may have experienced trauma. Answers to questions resulted in identifying three subthemes: resources needed by ECCE teachers, professional development available to ECCE teachers, and networks and organizations accessed by ECCE teachers. The following three sections provide additional detail for each of the three subthemes.

**Resources needed by ECCE teachers.** Respondents acknowledged the need for additional help inside the classroom, communication with the family, and external resources. Participant 10 stated,

I would say maybe resources like another person to help me to help them [students]. Resources to help me to back the child or give me insight about how to help them. Any extra information or resources to help me help them. Participant 3 offered,

Getting the right resources to be able to help that child not just at school but when they're not at school. To help the behavior of that child in the whole child aspect. Taking that approach, I feel like it not only helps that kiddo but also helps me balance my classroom. So, we know what to do when Joey's crying for this or that. We have tools in place to help that child. **Professional development available to ECCE teachers.** When ECCE teachers were asked how they gained information and strategies for effectively teaching young children who may have experienced trauma they described a variety of venues. Examples included post-secondary courses, face to face classes, online sources, and personal research. All participants had obtained professional development through inperson classes. Participant 4 commented,

[I need] lots of training in what to do. I'm more of a hands-on kind of person. I appreciate someone coming in and showing me how to go through the motions of how to deal with a situation instead of just watching it on online training or having a PowerPoint shown to us.

Participant 1 stated, "It's all been in person." Nine respondents had also obtained training online. Participant 8 responded, "Most of our yearly trainings like our modules and stuff we did online." Participant 6 mentioned, "Courses through school are online." Participant 9 identified specific online classes, "I've done a few of the courses in the KCCTO program." In addition to classes, Participant 3 said, "I've done a lot of just sitting back and observing what's going on and taking notes." Participant 2 noted, "There's been some in-person trainings, and there's been some online trainings, especially about the COVID and those sorts of things. Some of its just word of mouth, you know, like meeting with other teachers."

Regarding information on social-emotional development, Participant 5 noted, "I've not had any formal training that I can tell you exactly. I've gotten the information that I have from coworkers and management and trainings over the years. I took Child Development in college too." Participant 6 explained, Learning about social-emotional education is fascinating for me. I think children who have experienced trauma might have bigger emotions you can see more quickly. So, I think me having at least some sort of education, a teacher's inservice or my child development classes, has helped me always learn as an educator.

Participant 5 stated,

Once you've gotten your child abuse and neglect, your first aid, and safety, then what? That's it? We're done. Twenty-nine years ago, I did that? I don't remember much from it. I did it so long ago, decades ago, but there's nothing after that. And in-service may or may not touch on it [trauma] during professional development days. There's just not a lot of training out there that, at least from my point of view, that we're exposed to or know of.

Participant 3 shared, "I've had very little training on it [trauma] through a few websites. But besides that, that's really all the information and tools that I've been given or have been able to find on dealing with kids with trauma in the classroom." Participant 6 acknowledged, "We do have our in-service days where we learn about the trauma." Participant 9 mentioned, "Just taking classes and stuff." Participant 6 recalled,

One of my [CDA] classes is actually pretty much geared toward classroom management and dealing with children who have experienced trauma. It's based on understanding how to deal with them and making sure they have a safe space, making sure that they can talk to you. And they just have a space to deal with those emotions. Participant 7 explained, "I would definitely say training classes, only because there are certain children that have struggled with more traumatic experiences than other children. I think we, as individuals, as teachers, we need to know what and how to resolve an issue." Participant 9 reported, "My center is doing a trauma care training that we're starting the school year with presented by someone. We're starting to learn different ways to help kids that are in trauma."

Four teachers reported that Conscious Discipline<sup>®</sup> is a resource for professional development. Participant 2 commented, "We built in a lot of training with Conscious Discipline<sup>®</sup>. So that was kind of an underlying curriculum or strategy that we used there." Participant 4 stated,

We've also had a couple of Conscious Discipline<sup>®</sup> trainings with people who have been certified in Conscious Discipline<sup>®</sup> throughout the years. We also have a couple of resources. One of our directors is trained in Conscious Discipline<sup>®</sup> so we could always ask her questions and talk to her about different things. And I think that might be all.

Participant 8 reported, "We've done Conscious Discipline<sup>®</sup> training. And we've done several yearly trainings, but nothing specifically for the kids that need you more than others." Participant 3 said, "I've really researched on my Conscious Discipline<sup>®</sup> website for tools and things through Becky

Participant 4 described another professional development resource, "We have had a lot of meetings. She [the director] definitely wants us to have basic information before we walk into the classroom. So, we do modules from KCCTO. We do a bunch of things
through there." Participant 2 recalled, "At one center I worked in we did use some social-emotional curriculums, and things like that to try to help. We used the Second Steps, social and emotional curriculum."

Five respondents revealed they gain information through reading. Participant 5 remarked, "I know that I read everything that comes across my newsfeed or my news apps. They all know I'm interested in childcare-related things. So, all those things pop up." Participant 2 shared, "I've also read a couple books. Mostly like parenting books, but sometimes those can relate into the classroom as well." Participant 9 mentioned, "Just learning kind of from a book." Participant 7 reported, "I'm always reading stuff on the internet of new things that are coming out in child care and new things about children having emotions and people not understanding them." Participant 1 said, "I also read a lot online. And I will seek out information if I have a specific behavior or something that I feel is a little out there."

Networks and organizations accessed by ECCE teachers. All 10 respondents reported they had contact with organizations and agencies that provide support for children that may have experienced trauma or an informal network of colleagues and peers. Four of 10 participants reported personal connections to professional agencies and organizations that support children who may have experienced trauma. Participant 10 shared she received support through the foster care system, "I'm fostering my niece right now who has gone through trauma, and I've taken [foster parent] classes that have helped me." Participant 5 disclosed,

I've been sending people to the school district for decades to get screened for that 3 - 5-year-old Early Childhood Developmental Screening the Shawnee Mission School District has there once a month. One of my students and her caseworker came by today from Johnson County Mental Health. This child doesn't have any obvious behavioral challenges. She fits right in. And so, I asked the lady what I can do to get help for the kids that I really could use assistance with. [She said] you just have to refer them to Johnson County Mental Health. She said that they are teaming up with the school district. Evidently, the mental health clinic is now working with the school districts for those intake sessions. That was exciting news that they were teaming up.

Participant 1 recounted,

I was feeling a little burned out. I took one year off. During that time, I was volunteering. I learned so much about the systems that are in place. I completed my CASA volunteer training. And then, I went on to the LeAnn Britton Center, which is now the BC center. I worked as a para there. I learned so much from watching those teachers that have been doing that for years. I was trauma-informed trained through that.

Participant 9 recalled, "I have experience with children that have behaviors and are on the spectrum. In my 10 years of early childhood, I was a para-educator in the school district for three years." Participant 2 reported,

The only center that I've worked in that really had a partnership with someone was the Family Conservancy. The Family Conservancy aided in all of those trainings. Had we not had that partnership, I don't know that we would have gotten those trainings. I don't know that we would have known about the trainings about trauma-informed care.

Four participants noted that they rely on an informal network of peers and colleagues to gain information and support when working with children who may have experienced trauma. Participant 9 commented,

Just asking veteran teachers, 'Hey, you know, I'm not quite sure about this child, you know, behavior and the way they're acting, is that normal?' Somebody who may have a little bit more experience than me maybe will say, 'Yeah, pretty normal for the age.'

Participant 1 stated, "I can also reach out to the people that I worked with prior to my current setting and get some tips. I also troubleshoot with my other teachers." Participant 2 explained,

I depend on my peers and coworkers a lot. 'Hey, I have a kid doing XYZ. What has worked for you?' Kind of bounce those ideas off or calling upon more professionals in the field like the school social worker, psychologist, or whoever might have some resources that could help in those ways.

Participant 7 emotionally described,

I also have a friend. Her and her kids have gone through a lot of trauma in their life. She lost a son at three months old from SIDs. Her older kids witnessed the ambulance and everything. So, I've dealt with a lot of that with her and her children. So, I learned a lot from her. Only because she has so much going on, I take a lot of her advice and a lot of her information. I always have it in the back of my head. Participant 7 continued, "There's also some Facebook groups with parents that talk about their children and their emotions and types of therapy or stuff that they've read or learned about. I get a lot of information from them as well."

# **Challenges Working With Young Children Who Have Experienced Trauma**

Participants identified numerous challenges facing ECCE teachers working with young children who have experienced trauma. Within this theme, six subthemes emerged from the participant's responses: lack of awareness of trauma, needs of children who may have experienced trauma, the lack of preparedness of ECCE teachers, lack of relevant training opportunities for ECCE teachers, lack of research regarding young children who may have experienced trauma, and attitudes toward ECCE and trauma. Each of the six subthemes is described in additional detail in the next sections.

Lack of awareness of trauma. Six participants commented about the lack of awareness about trauma, noting that people have never heard of trauma in young children and do not understand or recognize it. Participant 2 acknowledged, "[I] never heard of trauma-informed care 10 years ago." Participant 1 stated, "People don't recognize what it [trauma] is." The teacher continued,

[I] feel like it's still fairly new. I remember being trained back with CASA, and they were saying that they are getting ready to do all of the trauma-informed training to schools. And I don't know if that, in fact, happened.

Participant 3 commented, "There's a lot of trauma that is out there that is related to children that people don't understand. That's trauma as well." Respondent 4 revealed, "They don't really read or observe the signs of trauma or challenging behaviors in children."

Participant 7 expressed her concern, "I think teachers need to be more aware of what they need to look for - signs and symptoms. Kids who may just not be there and just be super upset." Participant 5 shared the confusion among her coworkers,

One of my coworkers had a baby with a blue bottom. It was actually a greenish bluish tint. And they immediately jumped to 'Oh, my gosh! He's been spanked so hard that there's a bruise.' They didn't think about the fact that the baby just had a birthday and that it ate a bunch of icing that had discolored its poop. They immediately wanted to redline it and call it [abuse].

Participant 7 stated,

As a teacher, I think we have other members of our staff that do not recognize it [trauma] or seek it [training] out. So sometimes, if you have to have an assistant in your classroom, they may handle something totally different than you would have handled it. That is my frustration.

**Needs of children who may have experienced trauma.** Four participants described young children's needs: the need for special attention, the skills to release anger and get into a calm state, and someone present to support them. Participant 10 recognized,

They [children] need resources to help release that anger and help to get them in a calm state, and to know how to bring their emotions down. Kids have such big feelings and so many feelings that they don't understand. And so we have to help teach them about the feelings and the emotions that they feel because they don't know how to recognize them.

Participant 7 commented,

Basically, just making sure that each child in the classroom knows where they stand. And I think that some teachers in the classrooms don't know how to deal with children that are having a blowout of being very upset and not knowing how to calm down. And I think that's a key importance for not just the children, but I think it's more key for the teachers to know how to make that child feel a little bit better, supporting them through whatever they need for their schooling or their friends.

Participant 10 remarked, "We have to keep reminding ourselves that this child has been through way too much. And they need special attention. They need special things in their life." She also offered,

We've got so many children coming through our class that we forget sometimes, 'Oh, this child needs me. I need to make special arrangements in my class for the needs of this child so that the other children don't really notice who takes away from them'.

Participant 7 said, "Any child that has had any trauma in their life, definitely [needs] having somebody there for them, to show them emotions and show them how to calm down."

Three participants noted the developmental needs of children. Participant 1 stated,

We look at our little three-year-olds who have not really remembered anything outside of living in a pandemic. So, we were talking about, for example, a lot of teaching and learning goes on through facial expressions with the masks. You know, for one year, you can probably get it. We are now starting basically our third year affected by this. So, like one thing that I'm doing, I'm wearing a mask lanyard. So, I can pull my mask down and stand far away because a "B" and a "V' sound the same inside there. I want them to see my face more. We're just trying to be creative with how to maybe get some of the little ones with what they just missed out on. But it will be interesting to see how that's going to impact them in the long run.

Participant 3 expressed her concern,

It's important to me, with this whole mask thing, different people feel different about it. Kids are traumatized by that unfortunately. And my kiddos that are special needs in my room go to school every day and have to wear a mask. I know it's for their safety, and it's for the safety of others, but it's not helping develop their developmental needs. These kiddos have speech issues. One has speech, hearing, physical, I mean, he's got a long list of things. The past two years, it's been nothing but a damper on his education and physical development and mental development, too, because he can't hear you, and then he can't see your face. So, it's just, it's been really hard this year with that, and I've struggled a lot with that. I really hope that it gets changed soon, but I know it's not going to.

Participant 7 commented, "I just think we need to educate, constantly question, what can we do to get them [children] up to speed?"

Four participants were mindful of the support needed by children and teachers. Participant 8 stated, "I don't want to feel like they are getting forgotten or not getting the things they need. But I honestly don't think that we are able to give them the things they need, at least not on a large scale." Participant 2 expressed her concern,

They get bounced to another place, and then it's just kind of a cycle that goes down. So, trying to break that cycle in early childhood is a challenge. Obviously, not one person can be the person that solves it by themselves.

Participant 3 remarked,

Just knowing what to do and how to do it and having a good support system is the biggest thing that we need as early educators so we can help these kiddos. Otherwise, I don't know how they can stay in a program that doesn't work for the teachers and the kid.

Participant 7 shared her thoughts,

Children, you know, their brain is a sponge, and they remember. They don't remember everything, but it absorbs so much. Teachers do need to understand that as well. So, when the child is at home, and they're seeing these things, it's absorbing into their brain. They might have that experience. They might take it to school.

Lack of preparedness of ECCE teachers. Six respondents reflected on the lack of preparedness of ECCE teachers which included unrealistic expectations, inexperience or no experience, and insufficient skills and tools. Participant 8 stated," Honestly, I just don't think we're trained for it [trauma]. I don't think we're equipped for it [trauma]. And I don't think we're given the right tools to help those children." Participant 3 noted,

I think more than anything; we are not educated enough on how to deal with trauma. So not having the resources provided for us, we get burned out.

Unfortunately, places start losing teachers because we don't know what to do. We're not trained in this area of expertise. On top of the fact that we're already shorthanded in the early education field as it is. So, it just really makes it a challenge and a struggle to go to work every day and do your job.

Participant 4 reflected,

In my personal experience, I think it's just the kind of people that enter the early child care community. Most of them don't really know what they're getting into. I mean, as a high schooler, I never imagined that I'd be a preschool teacher, you know, and that I'd be dealing with all of this.

Participant 7 voiced concern regarding inexperienced teacher hires,

I do think that if a teacher has not had any childcare experience at all, or experience with children, definitely classes for them to take or read about something. Not all teachers are around children or have ever worked in childcare. So, they may not experience what we experience as teachers and what we know. I definitely think that they should have like a little film, or [information] in our pamphlet or booklet that we have for employment of how to deal with these things [trauma]. And if they haven't dealt with trauma, then we can all share stories of what we have gone through in our classrooms. Or if we had our own personal experience with children, then we can let them know of signs to look for. I would hate for a teacher to go into a classroom and not know anything about trauma. Participant 1 shared an observation,

As a teacher, I think we have other members of our staff that do not recognize trauma or seek it [training] out. So sometimes, if you have to have an assistant in your classroom, they may handle something totally different than you would have handled it.

Participant 4 said,

Sometimes they might lack confidence and think that they're not doing the right thing or doubt themselves, and how they're supposed to handle the situation and how they're handling other children in their care as well. So, I feel like that's just the greatest challenge of all.

Participant 8 added a reflection, "The whole situation just opened my eyes up to how unprepared we are to help kids like that if we don't know the situation. And a lot of times we can't ask the situation."

Lack of relevant training opportunities for ECCE teachers. All participants described challenges and limitations experienced in obtaining training related to the needs and challenging behaviors of children who may have experienced trauma. Participant 2 shared,

Depending on where you work, it kind of depends on how much training you get. If you work at somewhere that I've worked at, you know, places that don't give you a whole lot of training. Thankfully, I have my bachelor's degree and have quite a bit of experience under my belt. But not everybody has that. And so, they don't know how to handle it. They may just think that somebody is acting out for attention or something like that and not really know how to figure out what the root of it is.

Participant 3 commented, "I think the biggest thing is there needs to be more training out there." Participant 4 remarked,

I think the biggest one is, in fact, training. Teachers don't really have the sufficient training to deal with this [trauma] appropriately, and sometimes they think that they're not doing the right thing or doubt themselves and how they're supposed to handle the situation and how they're handling other children in their care as well. So, I feel like that's just the greatest challenge of all.

Participant 5 shared an observation,

I don't know that they [ECCE teachers] have enough training other than the first aid and safety and child abuse and neglect that we do when we first start in this field. I see a lot of teachers struggling with a student or a child and not understanding why that child is struggling. They get angry and frustrated about their child not being compliant. And then it's aggravating because you don't know what they've gone through or where they're coming from.

Two participants stated they did not have challenges in obtaining ECCE

professional development. Participant 6 revealed,

Our director, I feel like she really wants us to learn, and she wants to have a good staff. So anytime we've experienced interest in classes or asked her for help, she's really great about putting us in those courses or making sure she gives us the tools so that we can have a better education to teach the children. So, I personally haven't had any challenges with that.

Participant 1 disclosed, "I just feel like that's a matter of educating them [ECCE teachers], you know, getting the in-services in and hoping that they all seek it out to learn."

Among the challenges in accessing professional development for ECCE teachers, five participants identified the cost of the training. Participant 9 stated, "Sometimes the cost. Sometimes if the cost of a class is \$20, and I may not have that extra \$20 right then and there to take that class." Participant 2 reported, "I think cost is one of the challenges. Especially when you're in a private center it can be very expensive to have somebody who is an expert in those fields provide quality training." She continued, "I think that it's hard for centers to find good quality resources and training. And then if they [teachers] do ask for the training, maybe it's too costly." Participant 5 noted, "Sometimes, the school will help pay for training. Often, the school won't or can't cover the cost involved in additional training. And they just give you the amount that you need for an annual compliance check."

Three participants acknowledged that the center pays for their professional development. Participant 4 admitted,

I think that if [the director] wasn't providing it [training] for us, that it would be more of a money problem. I know that this training definitely cost a lot to take. She just paid for all of it. She does the same with other trainings that we've had throughout the years. And then we do our own research to try to find some free trainings throughout the year.

Participant 8 recalled, "At first, I didn't know that work would pay for it. And it was a money situation." On the other hand, Participant 6 stated, "Our center pays for it all."

Three participants talked about the availability of ECCE training. Participant 3 provided insight into the current availability of professional development,

Right now, because of COVID, I think we're very limited to anything that is faceto-face as far as training for any topic, but trauma as well. We had a really hard time finding anything out there that was trauma-related, and I can't remember what the site was. So, we finally did find one [training session], but we really have a difficult time finding any training.

Participant 5 reported, "An absence of the knowledge obviously, is frustrating. Sometimes we don't know when there's trainings available that could help that would benefit us." Participant 2 reported, "I think that it's hard for centers to find good quality resources and training. And then if they [teachers] do ask for the training, maybe it doesn't work with the center schedule." She added,

The Family Conservancy aided in all of those trainings. Had we not had that partnership, I don't know that we would have gotten those trainings from there. I don't know that we would have known about the trainings about trauma-informed care.

Three participants described the role the ECCE program took in providing professional development. Participant 7 disclosed, "I've never really thought it was hard to get training because [the director] did provide a lot of us with some trainings." She continued, "Sometimes it just might not be on the day that is best for you. But it always seems to work out." Participant 5 stated, "Another factor would be where they have the trainings. It is usually at the center".

Participant 2 described the problems in finding training resources, "I think that it's hard for centers to find good quality resources and training." Participant 2 further explained,

I think it's more of an access standpoint. When you work for a school district, you know people. It's kind of that social network in place that you can get to a school psychologist quicker or social worker quicker to help you get in contact with places like The Family Conservancy or whatever. At a center I worked at, we just happened to have that partnership. But there's several centers that I've worked at that didn't have any partnerships like that. So, there was not really any help that they could get besides just googling, finding somebody that might be able to help, if that makes sense.

Participant 10 responded, "Challenges? Time - finding the time between just juggling life. It's been hard. But we've made it through."

Four teachers discussed the relevancy and quality of course content. Participant 8 spoke honestly, "It's also not knowing what I should be taking. I can always ask to take a class, but I don't know what classes I should take." Participant 2 pointed out trainers may not deliver the content expected,

Sometimes the trainings that we've had just don't address the topic. You talk to them, and you're like, we need training on this [topic], but then when they show up, they kind of go off on a different tangent. And so, it didn't really necessarily help. In defense of the trainer, Participant 2 said, "I think in that same breath, it can also be hard for the trainer to scaffold to the teachers and the program too." Participant 5 described her frustration regarding the relevancy of the training content,

We have a professional development day, a half of that is dedicated towards training. But not necessarily trauma victims and how to deal with them. It's so boring because, you know, you've been through 1000 of them at this point in time, and you've heard all of it. It's just it's frustrating to listen to the other teachers. Participant 2 reiterated, "Sometimes for the experienced teachers, it can just be a repetitive thing and not any new information for them."

Two teachers reported little continued use of the training obtained. Participant 8 commented, "I just feel like anything that gets trained or taught to us, we use it for a little while, and then we revert back to what's natural for us, what we've always done. Participant 8 recalled,

I was given an opportunity to take an online course. It taught me about classroom setup and how placing your shelves will stop [children] running in your classroom and having it be like a safe space that also doesn't allow them too much free space at the same time. I did learn some of that. But it didn't really stick.

# Lack of research regarding young children who may have experienced

**trauma.** Two respondents expressed the need for more research. Participant 2 noted, "There's just been more research in the last 10 years, more studies. More knowledge base around it [trauma]." Participant 3 agreed, "I think the biggest thing is there needs to be more training out there. There needs to be more research out there." Attitudes toward ECCE and trauma. Three participants revealed attitudes held by adults toward children who may have experienced trauma. Participant 1 observed among her peers, "They just think that kids are behaving that way. Because they just need to be more disciplined or whatever it is, they may not understand that the child is kind of screaming out for more guidance and love." Participant 7 replied, "A lot of people and teachers say, "Well they are kids. It's gonna be okay. They don't remember a lot". But honestly, they do remember a lot." Participant 4 stated, "I think that most people don't take it [behavior] as seriously as they might need to." The comments offered by participants demonstrated a lack of understanding of trauma and its relationship to behavior.

Seven of 10 participants shared their expectations for continuing work in the early care and education profession and working with young children who may have experienced trauma. Three respondents described their expectations for the ECCE community. Participant 4 commented, "There should be more training involved before coming into the field or more knowledge of what the field is going to be like before hiring people." Participant 2 noted, "I wish that there was more training for those private sectors. Unfortunately, a lot of times, those are the ones that need it the most, or that have a lot of the trauma kids." Participant 10 declared, "Every teacher should be trained on it [trauma]." Three participants expressed their hopes for the future. Respondent 1 stated, "Hoping that they [ECCE teachers] all seek it [training on trauma]." Participant 5 fondly recalled, "I tell everybody I work with what I tell my students, so they will also tell all their kids how much they love them and that their class is the best class in school. Every teacher can say that to their own kiddos." Participant 7 expressed her goals, 'Being

the best teacher that I can be for all these children. I want to know everything that I can because I don't want to let these children down." Participant 6 shared her aspirations, "Hope other centers have someone who is as knowledgeable as my director." She added, "Hope everybody is passionate about it [early care and education]."

# Summary

Chapter 4 provided the results of the analysis of responses to interview questions provided by 10 ECCE teachers working in KDHE licensed ECCE centers in Johnson County, Kansas. Questions focused on the ECCE teacher's experience working with children who may have experienced trauma and its impact on classroom experience. Participants reflected on the information, skills, and resources needed to provide quality ECCE for young children who may have experienced trauma. Lastly, participants shared perceptions about the challenges of accessing professional development and the networks needed to work with young children who may have experienced trauma.

Chapter 5 provides an interpretation and recommendations related to the study. This last chapter provides a study summary that includes an overview of the problem, purpose statement and research questions, methodology review, and major findings. The major findings related to the literature are reported. Finally, chapter 5 provides conclusions that include implications for action, recommendations for future research, and concluding remarks.

#### Chapter 5

## **Interpretation and Recommendations**

The current study investigated ECCE teachers' perceptions about how trauma has impacted young children in early childhood center-based programs; the information and skills needed to meet the needs of young children who have experienced trauma; professional development opportunities available to them that provide preparation to meet the needs of young children who have experienced trauma; and the challenges ECCE teachers face in accessing professional development to aid them in working with young children who have experienced trauma. Chapter 5 is organized into three major sections. The first section presents an overview of the research problem, the purpose of the study and research questions, a review of the methodology, and major findings. The second section relates study findings to the current literature. Finally, the third section concludes with the researcher's implications for action, recommendations for future research, and closing remarks.

### **Study Summary**

This section summarizes the study, including an overview of the problem. Next, the purpose statement and research questions utilized in the study are identified. Finally, this section concludes with a review of the methodology and the major findings.

**Overview of the problem.** The National Association for the Education of Young Children (NAEYC) (1993) reported that it is increasingly evident that teachers and child care providers who provide specialized knowledge and skills to their job are best able to offer high-quality services to young children and their families. Early care and education teachers play a significant role in young children's social and emotional development. They are often on the front lines observing children's behaviors and interactions with peers, other adults, and the environment. ECCE staff frequently recognize or identify a challenge, a need, or a red flag in a young child's development (Sorrels, 2015). Erdman et al. (2020) reported that nearly every teacher would have contact with a child or family who has been impacted by trauma and indicated that the intervention and support a young child receives can make a critical difference in lifelong development. Slightly over onequarter of the children ages 0-4 have experienced trauma (Bailey & Krause, 2018). Trauma in young children is a public health crisis (Erdman et al., 2020; Van der Kolk, 2014). The American Psychological Association Presidential Task Force on Posttraumatic Stress Disorder and Trauma in Children and Adolescents (2008) indicated that most children and adolescents with traumatic exposure or trauma-related psychological symptoms are not identified and do not receive any help. Until recently, the efforts and funding to provide mental health training were focused on mental health professionals and clinicians, with limited training being offered to ECCE providers (Zero to Three, 2021). While there is compelling research about the development of young children and the quality of early childhood education experiences, few studies have focused on the perceptions of ECCE teachers about the impact of trauma on young children in center-based programs, the knowledge and skills needed to meet the challenges of these young students, and the availability and challenges of accessing professional development to work with young children who have experienced trauma.

**Purpose statement and research questions.** The purpose of this qualitative phenomenological study was to explore the perceptions of ECCE teachers about how trauma impacts young children in ECCE center-based programs and the adequacy of

professional development they receive to work with young children who have experienced trauma. Four purposes guided the current study. In addition, four research questions were formulated to address the purposes of this study.

**Review of the methodology.** A qualitative phenomenological research design was selected for this study. Phenomenological research focuses on the interpretation of the lived experiences of those interviewed. The phenomenon investigated in the current study was the perceptions of ECCE teachers in centered-based programs about the impact of trauma on young children, the knowledge and skills needed to work with traumatized young children, and the availability and challenges associated with accessing professional development focused on educating young children who have experienced trauma. This research method required the researcher in the current study to refrain from inserting personal experiences and focus exclusively on the participants' experiences. Constructivist-focused research allows the researcher to ask questions that lead to the inductive development of meaning from the data (Creswell, 2014).

Upon approval to conduct the study from the Baker University IRB committee (See Appendix B), 25 Facebook posts (see Appendix C) were sent to ECCE organizations, infant-toddler mental health professional groups, and ECCE programs. The social media posts were followed by individual emails inviting participation in the study (see Appendix D) to 40 ECCE centers. In addition, the invitation to participate letters were personally delivered to center directors at 20 ECCE programs in Johnson County, Kansas, for distribution to program staff. The efforts resulted in 12 ECCE teachers responding by email or text message expressing their interest in participating in the study. An email was sent to the 12 teachers confirming their interest in participating in the study (see Appendix E). As a result, 10 individuals consented to participate in the study. The participants in this study were individuals who had two or more years of experience working directly with young children who may have experienced trauma. Participants were employed in a center-based ECCE classroom in Johnson County, Kansas, from January 2020 through July 2021. All study participants signed an informed consent form (see Appendix F) prior to engaging in the interview.

The interview protocol was reviewed by the researcher's major advisor, research analyst, and three subject matter experts (SMEs), who provided feedback on the validity, relevance, significance, and clarity of the questions of the research protocol. Two pilot interviews were conducted. Both pilot interview reviewers were teachers who had worked in ECCE centers with young children who had experienced trauma. The pilot interview reviewers examined the interview questions, provided potential responses, and offered recommendations for future interviews. An interview protocol that included eight descriptive and demographic questions and 12 open-ended interview questions aligned with the research questions were designed for the study. Interviews were scheduled from July 2021 thru October 2021 and lasted 30 to 45 minutes. The interviews were conducted using Zoom video conferencing and were recorded. The video-audio recordings were transcribed using Otter.ai, an online transcriptions service. Each transcript was assigned a number (e.g., Participant 1, Participant 2, etc.) to maintain anonymity. After each interview, the researcher reviewed each audio-video recording to confirm that the recording was intact. Each respondent participated in member checking the interview transcript by reviewing its accuracy. No modifications were received. The researcher conducted in vivo coding for each transcript and then compiled the data by each

interview question into a spreadsheet. Throughout the analysis, the researcher reexamined transcripts to refine, filter, or delete codes. The researcher utilized a streamlined code to theory model (Saldana, 2016), moving from raw data to codes, categories and sub-categories, themes, and assertions. The researcher then reviewed the categories and sub-categories to determine themes and sub-themes. One coding auditor reviewed and supported the researcher's coding procedures. The auditor agreed with the data coding and findings. One recommendation was made to merge the five themes into four themes. Further analysis was conducted. As a result, four themes were identified that reflected the respondents' perceptions.

**Major findings.** Participant responses to interview questions offered varied experiences in center based ECCE programs working with young children who may have experienced trauma. Data analysis identified four themes: teachers' perceptions about trauma in early care and education, professional development needs of ECCE teachers, resources and professional development available to ECCE teachers, and challenges working with young children who may have experienced trauma. In addition, multiple sub-themes were noted with each theme.

All participants reported their perceptions about trauma in the early care and education field. Four subthemes were identified: the teachers' definitions of trauma; the behaviors of children who have experienced trauma; responses of other children to a child who has experienced trauma; and teachers' responses to the behaviors of a child who has experienced trauma. All 10 study participants indicated that trauma is an event that impacts a child's future. Examples of traumatic events included abuse and neglect, divorce, absence of a parent, death, and the pandemic. All participants described a range of behaviors observed in children who may have experienced trauma. Examples of children's behavior included aggression, regression, separation anxiety, and eating issues. Two participants commented that children were expelled from the child care facility due to the severity of a child's behavior in the early childhood setting. All participants shared multiple ways the behaviors of a child who may have experienced trauma impacted the other children in the ECCE program. Teachers described occasions when classmates avoided the child with the challenging behaviors or were frightened by the challenging behaviors displayed, or engaged in similar behaviors exhibited by the children. Participants reported how a child's challenging behaviors impacted their ability to manage classroom experiences and learning activities. Participant 8 summed up classroom activities, "There are good days and bad days. There have been days where I was able to have my class just normal." Four study participants disclosed teachable moments, the adjustments made in classroom routine and activities, and expressed their frustrations in being the only teacher in the classroom.

All participants voiced multiple professional development needs when working with young children who experienced trauma. Three subthemes were identified: information needed, skills needed, and attributes desired for ECCE teachers. Five respondents noted a need for general information about trauma. Seven participants indicated it would be helpful to know a child's history regarding trauma and the triggers that may impact the child's behavior. Five teachers specifically described the skills necessary in the practices of Conscious Discipline<sup>®</sup>. All teachers identified one or more attributes desired for teachers of young children who may have experienced trauma.

Attributes included patience, self-confidence, listening skills, and the desire to be a leader in the ECCE field.

All participants identified the resources and professional development-available to them when working with young children who experienced trauma. Three subthemes were identified: the resources needed by ECCE teachers, professional development available to teachers, and the networks and organizations accessed by teachers. Respondents acknowledged the need for additional help inside the classroom, communication with the family, and external resources. All participants had accessed professional development through in-person classes. Nine out of 10 respondents had obtained training through online providers. Participant 2 summarized, "There's been some in-person trainings. There's been some online trainings, especially [during] the COVID." Sources of professional development included college courses, CDA classes, center in-service days and staff meetings, reading, Conscious Discipline® offerings, and KCCTO online classes. Four participants described personal connections to professional agencies and organizations that support children who have experienced trauma. Five individuals had direct contact with Kansas Infant/Toddler Services, Johnson County CASA, a school district, a foster care agency, or The Family Conservancy. In addition, four participants stated they relied on an informal network of peers and colleagues, including veteran teachers, current and former coworkers, a friend, and Facebook groups.

All participants revealed numerous challenges facing ECCE teachers who work with young children who experience trauma. Six subthemes were identified through analysis of the data: lack of awareness of trauma; needs of children who may have experienced trauma; lack of preparedness of ECCE teachers; lack of relevant training opportunities for ECCE teachers; lack of research in the area of trauma and young children; and attitudes toward ECCE and trauma. Six participants commented about the lack of awareness about trauma, noting that people have never heard of trauma in young children and do not understand or recognize it. Four participants described young children's needs: the need for special attention, the skills to release anger and get into a calm state, and someone present to support them. Three participants described the needs and delays in language and social development of young children. Six respondents reflected on the lack of preparedness of ECCE teachers. For example, Participant 8 stated, "We're not trained for it [trauma]. We're not equipped for it. And I don't think we're given the right tools to help those children."

All participants described the challenges and the limitations experienced in obtaining professional development related to the needs and challenging behaviors of children who may have experienced trauma. Teachers revealed difficulties in locating available professional development opportunities, the cost of training, the scheduling of training, and the relevance and quality of the training provided. Participant 2 commented, "Depending on where you work, it kind of depends on how much training you get." Participant 3 provided insight into the current availability of professional development, "Right now, because of COVID, I think we're very limited to anything that is face-to-face as far as training for any topic, but trauma as well." She added, "We really we have a difficult time finding any training." Five participants identified the cost of the training as a barrier to accessing professional development. Three respondents described difficulty in the scheduling of training. Four teachers discussed the relevancy and quality of course content. Participant 5 shared an observation, "Once you've gotten your child

abuse and neglect, your first aid, and safety, then what?" Participant 8 spoke honestly, "It's also not knowing what I should be taking." Two participants stated the need for more research regarding young children who may have experienced trauma.

Although faced with numerous challenges working with young child who may have experienced trauma all ten participants offered optimistic closing thoughts. Seven of 10 participants shared their expectations for continuing work in the ECCE profession and working with young children who experienced trauma. Three respondents described their expectations for the ECCE community. Three participants expressed their hopes for the future. Participant 7 stated, "[I want to be] the best teacher that I can be for all these children. I want to know everything that I can because I don't want to let these children down."

#### **Findings Related to the Literature**

Trauma is a negative experience that chemically impacts the brain, mind, and body for a lifetime (Van der Kolk, 2014). Reactions to traumatic events can vary from child to child and may be presented as a variety of behaviors (American Psychological Association Presidential Task Force on Posttraumatic Stress Disorder and Trauma in Children and Adolescents, 2008; Buss et al., 2015; Erdman et al., 2020; Nicholson et al., 2019; Zero to Three, 2020). A range of social-emotional, communication, and developmental needs can be observed in young children who have experienced trauma (Alexander, 2019; Nicholson et al., 2019; Sorrels, 2015). Because of the trauma experienced by young children, ECCE teachers may observe tantrums, aggressive behaviors, increased irritability and frustration, hyper-alert responses, increased separation anxiety, detachment, lack of interest or ability to concentrate, regression, limited communication, or difficulty engaging adults and other children (American Psychological Association Presidential Task Force on Posttraumatic Stress Disorder and Trauma in Children and Adolescents, 2008; Buss et al., 2015; Erdman et al., 2020; Nicholson et al., 2019; Stone & Bray, 2015; Zero to Three, 2020). Stone and Bray (2015) reported that children who have experienced trauma might interact negatively or aggressively with classmates or adults or withdraw and not engage with others. Respondents in the current study articulated many of the same ramifications of trauma in the young children and early childhood educational settings noted by the researchers. There was consensus among all participants that trauma is an event that impacts a young child's life and may have far-reaching effects. Each participant reported one or more behaviors they had observed in children who may have experienced trauma. Seven of 10 respondents in the current study described the social-emotional and language developmental needs of trauma experiencing young children. Two participants commented that children were expelled from the child care facility because of the severity of their behaviors. Substance Abuse and Mental Health Services Administration (2020) and Tokarz (2017) reported that young children in early learning settings are expelled more than three times the number of children in elementary and secondary schools due to challenging behaviors.

Erdman et al. (2020) and Kaiser and Rasminisky (2021) indicated that early care and education programs lack the professional resources to manage the needs and challenging behaviors of children who may have experienced trauma. Bailey (2011) noted that teachers do not know what to do and are overwhelmed in their efforts to support young children with challenging behavior. According to Alexander (2019), there is no 'how-to' manual to prescribe a teacher's most effective response when interacting with a child who may have experienced trauma. Participants in the current study voiced multiple needs related to working with young children who experienced trauma. Respondents identified information, skills, resources, and personal attributes needed by teachers of young children who may have experienced trauma. Five respondents indicated the need for general information about trauma. According to Alexander (2019), Erdman et al. (2020), and Sorrels (2015) teachers will better understand the impact trauma has on young children and be better able to respond to their needs when they possess a definition of trauma and the possible resulting behaviors. Seven current study participants wanted a child's background information regarding trauma and the triggers that may impact the child's behavior. Nicholson et al. (2019) identified several skills needed by teachers, recognizing children's observable reactions to stress, minimizing the traumatic triggers in their classroom environments, and strategies to aid children in calming and self-regulation. Five participants in the current study described the skills and practices of Conscious Discipline<sup>®</sup>. Conscious Discipline<sup>®</sup> is a social-emotional learning program that is trauma-based and trauma-informed. In addition, the program is scientifically and research-based (Bailey, 2011, 2017). Bailey and Krause (2018) stated that establishing relationships with children having experienced trauma is difficult. It requires patience, empathy, and persistence. All participants in the current study identified one or more attributes desired for teachers working with young children who experienced trauma. Attributes included patience, self-confidence, listening skills, and the desire to be a pathfinder in the ECCE field.

According to Erdman et al. (2020) and Stone and Bray (2015) teachers cannot singlehandedly meet the needs of young children. However, they are critical advocates for better resources and support services for young children. Participants in the current study acknowledged the need for additional help in the classroom, support from outside of the classroom, and communication with family. Stone and Bray (2015) stated that teachers must remember that they do not work in isolation in schools. Young children grow and develop with a vast array of needs met through a network of individuals and services (Zeanah & Zeanah, 2019). Within the state of Kansas, access to infant-toddler mental health services varies. In addition, the availability of services differs in regions due to the funding, the number of trained mental health professionals, and the community focus and priorities (Kansas Health Institute, 2012; The Family Conservancy, 2021; Zeanah & Zeanah, 2019). Four participants in the current study described personal connections to professional agencies and organizations that support children who have experienced trauma. Five respondents had direct contact with at least one agency: the foster care system, Johnson County Mental Health, Johnson County CASA, a school district, or The Family Conservancy. Four participants stated they rely on an informal network of peers and colleagues, including veteran teachers, current and former coworkers, friends, and Facebook groups.

According to LeeKeenan and Chin Ponate (2018), professional development refers to training and educational experiences that support professional growth, online coursework, workshops, conferences, onsite technical training and assistance, books, professional journals, and professional learning communities. All 10 current study participants had accessed professional development through in-person classes. Participant 2 indicated, "There's been some in-person training. There's been some online trainings, especially during the COVID." Nine out of the 10 respondents had obtained training through online sources. Respondents indicated training occurred through college courses, CDA classes, center in-service days and staff meetings, Conscious Discipline<sup>®</sup> offerings, KCCTO online classes, and personal reading. Four of the 10 respondents in the current study discussed the relevance and quality of professional development content.

Early childhood teachers are receptive and searching for knowledge and skills, yet current professional development fails to provide theoretical knowledge of child development or the relevant application of the information to their work with children and families (Alexander, 2019; Hamre, Partee, & Mulcahy, 2017; Rhodes & Huston, 2012; Zaslow, Tout, Halle, Vick Whittacker, & Lavelle, 2010). Bailey and Krause (2018) indicated that no matter a teacher's educational or professional background, no one enters work with children who may have experienced trauma with the skills they need. Several researchers cited that the early childhood field has limited resources about trauma-informed practice (Alexander, 2019; Nicholson et al., 2019; Van der Kolk, 2014; Zaslow et al., 2010). Nicholson et al. (2019) observed that educators of young children have significantly few professional development opportunities compared to professions such as social work, nursing, or K-12 teachers. Dickinsen and Brady (2006) stated that funding ECCE programs, inclusive of professional development and training, is a critical challenge across the country. Most ECCE centers and family care providers are inadequately financed and lack the federal and state funds supporting Head Start/Early Head Start to provide training for teachers (Hamre et al., 2017; National Academies of

Sciences of Engineering and Medicine, 2022). Sim et al. (2019) noted that limited resources are available to an ECCE teacher, including time and funding. Three participants in the current study reflected on the lack of preparedness of ECCE teachers. Participant 2 disclosed, "Depending on where you work, it kind of depends on how much training you get." Participant 3 provided insight into the current availability of professional development. "Right now, because of COVID, I think we're very limited to anything that is face-to-face as far as training for any topic, but trauma as well." She added, "We really have a difficult time finding any training." All participants in the current study described challenges and limitations experienced in obtaining professional development related to children's needs and challenging behaviors who may have experienced trauma. Participants described difficulties in locating available professional development opportunities, the cost of training, the scheduling of training, and the relevance and quality of the training provided.

# Conclusions

The current study investigated ECCE teachers' perceptions about how trauma has impacted young children in early childhood center-based programs; the information and skills needed to meet the needs of young children who have experienced trauma; and the adequacy professional development opportunities available to them that provide preparation to meet the needs of young children who have experienced trauma; and the challenges ECCE teachers face in accessing professional development to aid them in working with young children who have experienced trauma. Ten ECCE teachers participated in the qualitative interviews. This section includes the researcher's recommendations for action, recommendations for future research, and closing remarks. **Implications for action.** Participant responses to interview questions in the current study provided insight into the needs of ECCE teachers and the resources ECCE teachers accessed when working with young children who may have experienced trauma. Participants reported that they and their colleagues possess limited knowledge about trauma and the skills needed to work with traumatized young children effectively. The results of this study have implications for additional support, resources, and professional development that is relevant and accessible.

ECCE teachers indicated they have limited resources and time to seek meaningful training opportunities from program administrators, the local resource and referral agency, and post-secondary institutions to fill in the gaps in trauma awareness and effective classroom management strategies. Early childhood administrators must consider steps to embrace and implement trauma-informed/trauma-sensitive programs to provide safe, positive relationships among children, staff, and families. To foster success for teachers and children, administrators of early care and education programs must develop a comprehensive professional development plan to support teachers' understanding of trauma awareness and the skills to employ trauma-sensitive strategies. A plan may require including a social and emotional learning (SEL) program such as Conscious Discipline<sup>®</sup>, the HighScope<sup>®</sup> Preschool Curriculum, or the Pyramid Model for Promoting Social Emotional Competence in Infants and Young Children. To bring a comprehensive professional development plan to fruition, ECCE owners, administrators, and center directors must communicate with resource and referral agencies about the need for relevant, frequent trauma-informed training for ECCE staff. In addition, they must communicate with post-secondary institutions voicing the need to provide in-depth

content on trauma in social and emotional development classes and offer accessible continuing education classes. ECCE owners, administrators, and directors must seek opportunities to collaborate and participate in cross-disciplinary partnerships in the community and region and engage in a search for infant- toddler mental health professionals and organizations to build a partnership for referrals and consultation and training for staff and families.

With the awareness that ECCE professionals seek and require relevant information and skills to work with young children who have experienced trauma, individuals and organizations must answer the call to provide accessible and affordable professional development to the private sector ECCE professionals. A review of the literature for this study revealed that research and professional development are currently available to infant-toddler mental health clinicians and service providers. Infant-toddler mental health professionals must reach out and develop partnerships beyond Head Start/Early Head Start and school districts to meet the needs of private sector ECCE programs and family care providers. Efforts must be made to expand the training opportunities usually provided to mental health professionals working with young children who have experienced trauma to ECCE teachers and family care providers. Assistance and support must be offered to ECCE owners and administrators to create and maintain trauma-responsive early care and education programs.

In order for ECCE professional and infant-toddler mental health professionals to best support young children who may have experienced trauma and their families, individuals and organizations must become collaborators and advocates. Time and effort must become a priority to alert policymakers and elected officials at the local, state, and national levels about the mental health needs of young children and their families. In addition, policies and funding are needed to support professional development. All individuals caring for and educating young children must consider joining the advocacy efforts of NAEYC, Power to the Profession, and Zero to Three.

Parents are a child's first teacher. Families cannot be ignored or left out of partnerships with ECCE professionals and infant-toddler mental health specialists. Communication and efforts must be made to support young children who have experienced trauma. Families also have a role in advocating for their children's health and emotional well-being. Parents of young children must trust their early care and education providers and be willing to courageously share their traumatic histories, voice their concerns, and develop lasting partnerships that will benefit their children.

**Recommendations for future research.** The experiences described in the current study may provide insight into increasing the relevant professional development opportunities focused on information and skills to aid ECCE teachers in working with young children and their families who may have experienced trauma. The current study involved 10 ECCE teachers. The participants in this study were individuals who had two or more years of experience working directly with young children who may have experienced trauma. Participants were employed in a center-based ECCE classroom in Johnson County, Kansas, from January 2020 through July 2021.

The current study included only ECCE teachers in center-based early learning programs. Researchers have reported that over 3 million home-based ECCE providers may work with young children who have experienced trauma. The first recommendation based on the findings of the current study is to examine the perceptions of family care providers about how trauma impacts young children in family care programs and the adequacy of the training family care teachers receive to work with young children who have experienced trauma. A future study could investigate similarities and differences between ECCE family care teachers and providers and center-based teachers and providers.

The sample was limited by geographic location where participants were employed in a center-based ECCE classroom in Johnson County, Kansas. A second recommendation for future research is to conduct a study similar to the current study to include ECCE teachers and providers across Kansas. The current study involved the use of a qualitative research design. A quantitative research design using a survey focusing on elements of the current survey could be used to investigate ramifications of trauma in early care and education, teachers' professional development needs working with young children who may have experienced trauma, resources and professional development accessed by ECCE teachers, and the challenges working with young children who may have experienced trauma. Data could provide more in-depth information about each variable. The current study included only classroom teachers in ECCE centers. Therefore, the third recommendation for future research is to study ECCE directors, administrators, and owners' perceptions of trauma-informed care and their access to professional opportunities to offer them and practices they may provide or refer their staff.

Existing research has indicated that ECCE teachers affiliated or employed with Head Start/Early Head Start programs and school districts have access to professional development to support teachers when working with young children who have experienced trauma. Therefore, a fourth recommendation for future research is to explore the local, regional, and national infant-toddler mental health resources currently available to ECCE not affiliated with Head Start/Early Head Start or a school district. In addition, the study may examine how professional development opportunities may be accessed and collaborations formed with the private sector to support children and families who may have experienced trauma.

The current study was initiated before the COVID-19 pandemic. Therefore, the literature reviewed did not reflect the present circumstances of experiencing a collective traumatic event. The final recommendation for future research is to study the impact of the COVID-19 pandemic on information, resources, and professional development developed and accessible to the private sector ECCE teachers, programs, and family care providers in their support of young children and families who have experienced the trauma of the pandemic.

**Concluding remarks.** Through the experiences of the COVID 19 pandemic, it is apparent that individuals involved in the early care and education workforce are essential to the economy, the well-being of families, and the growth and development of children (Hogan, 2020; Lee & Parolin, 2021). The early care and education workforce includes nearly 2 million individuals in center-based programs and over 3 million in family care settings (Rhodes & Huston, 2012). The available literature suggests that the field of early care and education has been fragmented due to its origination, governance and affiliations, and funding sources. Over the past 30 years, efforts have been made to develop a collaboration among ECCE professionals to create a united workforce, identify professional designations, align professional preparation, advocate for compensation, and
gain government recognition and support (Hogan, 2020; LeeKeenan & Chin Ponte, 2018; Power to the Profession Task Force, 2020). Early childhood development and emotional health are critical to a child's future well-being. Access to quality early care and education and mental health services for young children who need them diminishes the risk of problems later at home, school, and in the community (Kansas Health Institute, 2012; Van der Kolk, 2014; Zeanah, & Zeanah, 2019). The information gained by mental health professionals has been slow to move into the field of ECCE (Buss et al., 2015; Sorrels, 2015). Collaborative efforts must be established and continued through local, state, and national networks comprised of early care and education and infant-toddler mental health professionals, for-profit and nonprofit organizations, and resource and referral sources, including ECCE teachers and administrators (Hogan, 2020; Zaslow & Martinez-Beck, 2006; Zero to Three, 2021). This study contributed to the literature as it is the first to examine ECCE teachers' perceptions about how trauma impacts young children and classroom experience in ECCE center-based programs and the adequacy of the training ECCE teachers receive to work with young children who have experienced trauma. Findings from the interviews and data analysis confirmed existing literature regarding the ramification of trauma on children's behavior and its impact on the ECCE classroom. Participants described the need for information about trauma, the skills, and resources to meet the needs of children in their classroom who have experienced trauma and manage classroom activities. Additionally, each participant reported challenges in accessing resources and professional development to support working with young children who may have experienced trauma. Currently, KDHE licensing requirements and post-secondary coursework do not focus on trauma or trauma-informed practice.

ECCE teachers face numerous challenges in accessing the professional development they seek or perceive that is needed to provide quality early care and education and work successfully with young children who may have experienced trauma. The current study's findings identified the need for collaboration among ECCE professionals and infant-toddler mental health practitioners and organizations to provide relevant professional development for ECCE teachers and providers and advocate for young children and their families who have experienced trauma.

#### References

- Alexander, J. (2019). Building trauma sensitive schools: Your guide to creating safe, supportive learning environments for all students. Baltimore, MD: Paul H. Brookes Publishing Co.
- Alleyne, A. (2016, September 9). 'You say you want a revolution': 9 moments that defined the '60s. Retrieved from https://www.cnn.com/style/article/you-say-youwant-a-revolution-victoria-and-albert-museum/index.html
- American Psychological Association Presidential Task Force on Posttraumatic Stress
   Disorder and Trauma in Children and Adolescents. (2008). *Children and trauma: Update for mental health professionals*. Retrieved from American Psychological
   Association website: http://www.apa.org/pi/families/resources/update.pdf
- Bailey, B. (2011). Managing emotional mayhem: The five steps for self-regulation.Oviedo, FL: Loving Guidance.
- Bailey, B. (2017). *The theoretical and scientific basis of conscious discipline*. Oviedo,FL: Loving Guidance.
- Bailey, B. & Krause, A. (2018, November 30). Real talk for real teachers #32- Conscious leadership: Ways to support teachers who work with trauma [Video file].
  Retrieved from <u>https://www.youtube.com/watch?v=TpTmc3hPFIk</u>

Barnett, W. S. (2011). Effectiveness of early educational intervention. *Science*, *333*(6045), 975-978. doi: 10.1126/science.1204534

- Bartlett, J. D., & Sacks, V. (2019, April 10). Adverse childhood experiences are different than child trauma, and it's critical to understand why [Web blog post]. Retrieved from https://www.childtrends.org/blog/adverse-childhood-experiences-differentthan-child-trauma-critical-to-understand-why
- Barwick, M. & Urajnik, D. (2021, January 27). Child mental health. *Britannica Encyclopedia*. Retrieved from https://www.britannica.com/topic/child-mental-health
- Blischke, W. R., Rezaul, K. M., & Prabhakar Murthy, D. N. (2011). Preliminary data analysis. In Warranty data collection and analysis (pp 159-189). https://doi.org/10.1007/978-0-85729-647-4\_8
- Bloomberg, L. D., & Volpe, M. (2012). *Completing your qualitative dissertation: A road map from beginning to end* (2nd ed.). Thousand Oaks, CA: Sage Publications.
- Bloomberg, L. D., & Volpe, M. (2019). *Completing your qualitative dissertation: A road map from beginning to end* (4th ed.). Thousand Oaks, CA: Sage Publications.
- Boothe Trigg, A. & Keyes, A. W. (2019). Child care and early education as contexts for infant mental health. In C. Zeanah (Ed.), *Handbook of infant mental health*, (pp. 599-609). New York, NY: Guildford Press.
- Bornstein, M. H. (2012). Cultural approaches to parenting. In *Parenting, Science and Practice, 12*(2-3), 212–221. https://doi.org/10.1080/15295192.2012.683359
- Briggs-Gowan, M. J., Ford, J. D., Fraleigh, L., McCarthy, K., & Carter, A. S. (2010).
  Prevalence of exposure to potentially traumatic events in a healthy birth cohort of very young children in the northeastern United States. *Journal of Traumatic Stress*, 23(6), 725–733. https://doi.org/10.1002/jts.20593

Bickman, L. (2022). Mental health services. In *Education Encyclopedia -StateUniversity.com*. Retrieved from https://education.stateuniversity.com/pages/2224/Mental-Health-Services-Children.html

- Burchinal, M., Kainz, K., & Cai, Y. (2011). How well do our measures of quality predict child outcomes? A meta-analysis and coordinated analysis of data from large-scale studies of early childhood settings. *In:* Zaslow, M., Martinez-Beck, I., Tout, K. and Halle, T. (Eds.). *Quality measurement in early childhood settings* (pp. 11–31). Baltimore, MD: Paul H. Brookes Publishing.
- Buss, K. E., Warren, J. M., & Horton, E. (2015). Trauma and treatment in early childhood: A review of the historical and emerging literature for counselors. *The Professional Counselor*, 5(2), 225–237. Retrieved from http://tpcjournal.nbcc.org
- Cahan, E. D. (1989). Past caring: A history of U.S. preschool care and education for the poor 1820-1965. Retrieved from

https://www.researchconnections.org/childcare/resources/2088#

- Centers for Disease Control and Prevention. (2020a). *Considerations for institutions of higher education*. Retrieved from https://www.cdc.gov/coronavirus/2019ncov/community/colleges-universities/considerations.html
- Centers for Disease Control and Prevention. (2020b). *Early brain development and health*. Retrieved from https://www.cdc.gov/ncbddd/childdevelopment/earlybrain-development.html

Centers for Disease Control and Prevention. (2022). *Data and statistics on children's mental health* [Webpage]. Retrieved from

https://www.cdc.gov/childrensmentalhealth/data.html

- Center for Substance Abuse Treatment. (2014). *Appendix C: Historical account of trauma*. (Treatment Improvement Protocol (TIP) Series, No. 57.). Retrieved from: https://www.ncbi.nlm.nih.gov/books/NBK207202/
- Center on the Developing Child. (2007). *The science of early childhood development*. Retrieved from https://developingchild.harvard.edu/resources/inbrief-science-of-ecd/
- Child Care Aware® of America. (2016). *Staff handbook 2*. Retrieved from https://www.childcareaware.org/wp-content/uploads/
- Child Care Aware of Kansas. (2019). 2019 Child care supply demand report. Retrieved from https://ks.childcareaware.org/for-businesses-and-the-community/research-information/
- Child Care Lounge. (2018). *Milestones: A child care history timeline*. Retrieved from https://childcarelounge.com/pages/milestones-a-child-care-history-timeline
- Child labor. (2020, April 17). Retrieved from https://www.history.com/topics/industrialrevolution/child-labor
- Childcare Resource and Research Unit. (2020). *What is quality in early childhood education and care?* Retrieved from https://findingqualitychildcare.ca/high-quality-child-care/what-is-quality

Children's Health Act of 2000, Pub. L. No. 106-310, H.R. 4365, 106th Congress.

- Cleveland Clinic Pediatrics. (2020, April 28). The lasting effects of childhood trauma [Web blog post]. Retrieved from https://health.clevelandclinic.org/childhoodtraumas-lasting-effects-on-mental-and-physical-health/
- Cohen, J., Onunaku, N., Clothier, S., & Poppe, J. (2005). *Helping young children* succeed: Strategies to promote early childhood social and emotional development (Early Childhood Research and Policy Report). Retrieved from National Conference of State Legislatures website: https://www.zerotothree.org/resources/136-helping-young-children-succeed-

strategies-to-promote-early-childhood-social-and-emotional-development

Concordia University St. Paul. (2019, January 23). 7 must-haves for early childhood teachers [Web blog post]. Retrieved from

https://online.csp.edu/blog/education/early-childhood-teachers/

- Conscious Discipline. (2021). *Key outcomes*. Retrieved from https://consciousdiscipline.com/methodology/research/
- Creswell, J. W. (2014). *Research design: Qualitative, quantitative, and mixed methods approaches.* Thousand Oaks, CA: Sage Publications.
- Creswell, J. W., & Plano Clark, V. W. (2011). *Designing and conducting mixed methods research*. Thousand Oaks, CA: Sage Publications.
- Cui, J., & Natzke, L. (2020). *Early childhood program participation: 2019* (NCES 2020-075). Retrieved from http://nces.ed.gov/pubsearch/pubsinfo.asp?pubid=2020075.
- DeBord, K. (n.d.). *Quality child care: What does it really mean?* Retrieved from https://extension.tennessee.edu

- Demause, L. (2002). The evolution of childrearing. In *The emotional life of nations*. Retrieved from https://psychohistory.com/books/the-emotional-life-of-nations/chapter-8-the-evolution-of-childrearing/
- De Young, A. C., Kenardy, J. A., & Cobham, V. E. (2011). Trauma in early childhood: A neglected population. *Clinical Child and Family Psychology Review*, 14(3), 231– 250. doi.org/10.1007/s10567-011-0094-3
- Dickinsen, D. K., & Brady, J. P. (2006). Toward effective support for language and literacy through professional development support. In M. Zaslow & I. Martinez-Beck (Eds.), *Critical issues in early childhood professional development* (pp. 141–170). Baltimore, MD: Paul H Brookes Publishing.
- Doncaster, C. P. (2021). *Timeline of the human condition*. Retrieved from https://www.southampton.ac.uk/~cpd/history.html
- Early Childhood Education. (2020). In *Encyclopedia of Children's Health*. Retrieved from http://www.healthofchildren.com/E-F/Early-Childhood-Education.htm
- Early, D. M., Maxwell, K. L., Burchinal, M., Alva, S., Bender, R. H., Bryant, D.,... Zill, N. (2007). Teachers' education, classroom quality, and young children's academic skills: Results from seven studies of preschool programs. *Child Development*, 78(2), 558-580. doi: 10.1111/j.1467-8624.2007.01014.x

Economic Opportunity Act of 1964, Public Law 88-452, 78 STAT 508.

Erdman, S., Colker, L. J., & Winter, C. (2020). Trauma and young children: Teaching strategies to support and empower children. Washington DC:-National Association for the Education of Young Children. Ertman, T. (2019, June 27). The Lanham Act and universal childcare during World War II [Web blog post]. Retrieved from

https://www.wwiimemorialfriends.org/blog/wwii-universal-childcare/

- Eunice Kennedy Shriver National Institute of Child Health and Human Development, National Institutes of Health, Department of Health and Human Services, (2009). Child Development and Behavior Branch (CDBB), NICHD, *Report to the NACHHD Council, January 2009* (Reference Only) (NA). Retrieved from https://www.nichd.nih.gov/publications/product/316
- First Five Years Fund. (2021a). *Federal legislation impacting early childhood education due for reauthorization*. Retrieved from https://www.ffyf.org/federal-legislationimpacting-early-childhood-education/
- First Five Years Fund. (2021b). *Why-it-matters/lifelong gains*. Retrieved from https://www.ffyf.org/why-it-matters/lifelong-gains/
- Foundation Recovery Network. (2021). *History of mental health treatment: Dual diagnosis*. Retrieved from https://dualdiagnosis.org/mental-health-and-addiction/history/
- Friedman-Krauss, A. H., Barnett, W. S., Garver, K. A., Hodges, K. S., Weisenfeld, G. G.,
  & DiCrecchio, N. (2019). *The state of preschool 2018: State preschool yearbook*.
  Retrieved from https://nieer.org/state-preschool-yearbooks/2018-2
- Fu, Y., & Sam Liang, S. (2020). Otter.ai [Online Software]. Retrieved from https://otter.ai/

- Gebhard, B., Ochshorn, S., & Jones, L. (2010). Toward a bright future for our youngest children: Building a strong infant-toddler workforce. Washington, DC: Zero to Three, Policy Center.
- Gerrity, E. & Folcarelli, C. (2008). Child traumatic stress: What every policymaker should know. Retrieved from https://www.nctsn.org/resources/child-traumaticstress-what-every-policymaker-should-know
- Godoy, L., Davis, A., Heberle, A., Briggs-Gowan, M., & Carter, A. S. (2019).
  Care giver report measures of early childhood social-emotional function. In C.
  Zeanah (Ed.), *Handbook of infant mental health* (pp. 259-278). New York, NY:
  Guildford Press.
- Grille, R. (2006, Dec 11). From horror to hope: The evolution of childrearing. Retrieved from https://www.kindredmedia.org/2006/12/from-horror-to-hope-the-evolution-of-childrearing/
- Guarino, A. (2021). FFYF's blueprint for progress. Retrieved from https://www.ffyf.org/ffyfs-blueprint-for-progress/
- Gwynedd Mercy University. (2020). Early childhood education degree and career overview. Retrieved from https://www.gmercyu.edu/academics/learn/what-isearly-childhood-education
- Hackett, E. A. M. (2011). A longitudinal study on the achievement of the children whose parents participated in a Parent As Teachers program (Doctoral dissertation).Available from ProQuest (UMI 3576683)

- Halfon, N., Shulman, E., & Hochstein, M. (2001). Brain development in early childhood.
  In N. Halfon, E. Shulman & M. Hochstein (Eds.), *Building community systems for young children* (pp. 1–26). Retrieved from ERIC database. (ED467320)
- Hamre, B. K., Partee, A., & Mulcahy, C. (2017). Enhancing the impact of professional development in the context of preschool expansion. *AERA Open*, 3(4). 1– 16.https://doi: 10.1177/2332858417733686

Head Start Expansion and Quality Improvement Act of 1990, S.2229, 101st Congress.

- Heywood, C. (2001). A history of childhood: Children and childhood in the west from medieval to modern times. Malden, MA: Polity Press.
- Ho, L., & Limpaecherm, A. (2020). *Essential guide to coding qualitative data*. Retrieved from https://delvetool.com/guide
- Hogan, L. (2020). *Power to the profession and the unifying framework for the early childhood education profession* [Video file]. Retrieved from https://www.youtube.com/watch?v=hDvVr2hXdLo
- Houry, D. (2019, July 11). *Identifying, preventing, and treating childhood trauma*. Retrieved from https://www.cdc.gov/washington/testimony/2019/t20190711.htm
- Hustedt, J. T., & Barnett, W. S. (2011). Financing early childhood education programs:
  State, federal, and local issues. *Educational Policy*, 25(1), 167–192.
  doi.org/10.1177/0895904810386605
- IGI Global. (2020). *What is young children?* Retrieved from https://www.igi-global.com/dictionary/young-children/33643

- Institute of Medicine and National Research Council. (2012). *The early childhood care and education workforce: Challenges and opportunities: A workshop report.* https://doi.org/10.17226/13238
- Institute of Medicine and National Research Council. (2015). *Transforming the* workforce for children birth through age 8: A unifying foundation https://doi.org/10.17226/19401
- Ippen, G. C., Harris, W. W., Van Horn, P., & Lieberman, A. F. (2011). Traumatic and stressful events in early childhood: can treatment help those at highest risk? *Child Abuse & Neglect*, 35(7), 504–513. https://doi.org/10.1016/j.chiabu.2011.03.009
- Kaiser, B., & Rasminsky, J. S. (2021). Addressing challenging behavior in young children: The leader's role. Washington, D.C.: National Association of the Education of Young Children.
- Kamerman, S. B. (2006). A global history of early childhood education and care. In Strong foundations: Early childhood care and education—EFA global monitoring report 2007. Retrieved from http://lst-iiep.iiep-unesco.org/cgibin/wwwi32.exe/[in=epidoc1.in]/?t2000=023596/(100)
- Kamerman, S. B., & Gatenio-Gabel, S. (2007). Early childhood education and care in the United States: An overview of the current policy picture. *International Journal of Child Care and Education Policy 1*, 23–34. https://doi.org/10.1007/2288-6729-1-1-23
- Kansas Department of Health and Environment. (2020). *Kansas law and regulations for licensing preschools and child care centers*. Topeka, KS: Author.

- Kansas Health Institute. (2012, August 4). Early childhood mental health Services in Kansas [PDF]. Retrieved from https://healthforward.org > uploads > 2012/02 > Childrens\_Mental Health\_KS-\_report\_1.pdf
- Kiesling, L. (2019, October 2). Paid child care for working mothers? All it took was a world war. *New York Times*. Retrieved from https://www.nytimes.com/2019/10/02/us/paid-childcare-working-motherswwii.html
- Kvale, S., & Brinkman, S. (2009). InterViews: Learning the craft of qualitative research interviewing. Thousand Oaks, CA: SAGE Publications.
- Laughlin, L. (2013). *Who's minding the kids? Child care arrangements: Spring 2011* (Current Population Reports, 70-135). Retrieved from U.S. Census Bureau website: https://www.census.gov/prod/2013pubs/p70-135.pdf
- Lee, E., & Parolin, Z. (2021). The care burden during COVID 19: A national database of child care closures in the United States. *Socius: Sociological Research for A Dynamic World* (7), 1-10. doi: 10.1177/2378023121 1032028
- LeeKeenan, D., & Chin Ponate, I. (2018). From survive to thrive: A director's guide for leading an early childhood program. Washington, DC: National Association of the Education of Young Children.
- Lieberman, A., Chu, A., Horn, P., & Harris, W. (2011). Trauma in early childhood:
  Empirical evidence and clinical implications. *Development and Psychopathology*, 23, 397-410. doi:10.1017/S0954579411000137
- Lincoln, Y. S., & Guba, E. G. (1985). *Naturalistic inquiry*. Newbury Park, CA: Sage Publications.

- Lumen Learning. (n.d.). Mental health treatment: Then and now. In *Introduction to Psychology: Module 15: Therapy and Treatment*. Retrieved from https://courses.lumenlearning.com/wmopen-psychology/chapter/introduction-tomental-health/
- Lunenburg, F. C., & Irby, B. J. (2008). Writing a successful thesis or dissertation: Tips and strategies for students in the social and behavioral sciences. Thousand Oaks, CA: Corwin Press.
- Magruder, K., Serpi, T., Kimerling, R., Kilbourne, A. M., Collins, J. F., Cypel, Y., ...
  Kang, H. (2015). Prevalence of posttraumatic stress disorder in Vietnam-era
  women veterans: The health of Vietnam-era women's study (HealthVIEWS).
  JAMA Psychiatry, 72(11), 1127–1134. doi:10.1001/jamapsychiatry.2015.1786
- Maroto, M. L., & Brandon, R. N. (2011). *Commissioned papers The early childhood care and education workforce*. Retrieved from https://www.ncbi.nlm.nih.gov > books > NBK189910
- Martinez-Beck, I., & Zaslow, M. (2006). Introduction: The context for critical issues in early childhood professional development. In M. Zaslow & I. Martinez-Beck (Eds.), *Critical issues in early childhood professional development* (pp. 1–16).
  Baltimore, MD: Paul H Brookes Publishing.
- Maxwell, K. L., Lim, C. I., & Early, D. M. (2006). Early childhood teacher preparation programs in the United States: National report. Chapel Hill, NC: The University of North Carolina, FPG Child Development Institute.

- McLoyd, V. C., Aikens, N. L., & Burton, L. M. (2006). Childhood poverty, policy, and practice. In K. A. Renninger, I. E. Sigel, W. Damon, & R. M. Lerner (Eds.), *Handbook of child psychology: Child psychology in practice* (pp. 700–775). Hoboken, NJ: John Wiley & Sons.
- McNamara, C. (n.d.). *General guidelines for conducting research interviews*. [Power Point presentation]. Retrieved from

http://managementhelp.org/businessresearch/interviews.htm

- Mechling, J. (2021). *Child-rearing advice literature*. Retrieved from https://www.encyclopedia.com/children/encyclopedias-almanacs-transcripts-andmaps/child-rearing-advice-literature
- Mental Health America. (2020). *Our history* [Webpage]. Retrieved from Mental Health America website: https://www.mhanational.org/our-history
- Michel, S. (2011). *The history of child care in the U.S.* Retrieved from http://socialwelfare.library.vcu.edu/programs/child-carethe-american-history/
- Monnat, S., & Chandler, R. (2015). Long-term physical health consequences of adverse childhood experiences. *Sociological Quarterly*, (4): 723-752. Retrieved from https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4617302/

Moodie, S., Daneri, P., Goldhagen, S., Halle, T., Green, K., & LaMonte, L. (2014). Early childhood developmental screening: A compendium of measures for children ages birth to five (OPRE Report 201411). Retrieved from https://files.eric.ed.gov/fulltext/ED561406.pdf

- National Academies of Sciences, Engineering, and Medicine. (2018). *Transforming the financing of early care and education*. https://doi.org/10.17226/24984
- National Academies of Sciences, Engineering, and Medicine. (2022). Addressing the impact of COVID-19 on the early care and education sector. https://doi.org/10.17226/26463.

National Association for the Education of Young Children. (1993). Conceptual framework for early childhood professional development. Retrieved from http://www.naeyc.org/files/naeyc/file/positions/PSCONF98.PDF

- National Association for the Education of Young Children. (2009). *Developmentally appropriate practice in early childhood programs serving children from birth through age 8.* Retrieved from https://www.naeyc.org/resources/positionstatements/dap
- National Association for the Education of Young Children. (2020). Developmentally appropriate practice: A position statement of the national association for the education of young children. Retrieved from

https://www.naeyc.org/resources/position-statements/dap/contents

- National Association for the Education of Young Children, & National Association of Child Care Resource & Referral Agencies. (2011). *Early childhood education professional development training and technical assistance glossary* [PDF file]. Retrieved from https://www.naeyc.org/GlossaryTraining\_TA.pdf
- National Child Traumatic Stress Network. (2010). *What is child trauma?* Retrieved from https://www.nctsn.org/what-is-child-trauma/about-child-trauma

- National Child Traumatic Stress Network Schools Committee. (2008). *Child trauma toolkit for educators*. Retrieved from https://www.nctsn.org/resources/childtrauma-toolkit-educators
- National Research Council and Institute of Medicine. (2000). From neurons to neighborhoods: The science of early childhood development. Washington, DC: The National Academies Press.
- National Scientific Council on the Developing Child. (2007). *The science of early childhood development: Closing the gap between what we know and what we do* [PDF]. Retrieved from https://developingchild.harvard.edu/resources/the-scienceof-early-childhood-development-closing-the-gap-between-what-we-know-andwhat-we-do/
- Nicholson, J., Perez, L., & Kurtz, J. (2019). *Trauma-informed practices for early childhood educators*. New York, NY: Routledge.

Niemi, K. (2020, December). What is SEL? Retrieved from https://casel.org/what-is-sel/

- No Child Left Behind (NCLB) Act of 2001, Pub. L. No. 107-110, § 101, Stat. 1425 (2002).
- Obladen, M. (2016). From right to sin: Laws on infanticide in antiquity. *Neonatology*, *109*(1), 56-61. doi: 10.1159/000440875
- Old wives tale. (2021). In *Collins English Dictionary online*. Retrieved from https://www.collinsdictionary.com/us/dictionary/english/old-wives-tale

Oneill, T. (2013, April 19). Don't think of ugly people: How parenting advice has changed. *The Atlantic*. Retrieved from https://www.theatlantic.com/health/archive/2013/04/dont-think-of-ugly-peoplehow-parenting-advice-has-changed/275108/

Palmer, J. (2021, April 27). The early care and education workforce. *National Conference of State Legislatures*. Retrieved from https://www.ncsl.org/research/human-services/the-early-care-and-education-workforce.aspx

Parents as Teachers. (2021). Our history. Retrieved from parentsasteachers.org

- Paul, C. A. (2017). National child labor committee (NCLC). Retrieved from http://socialwelfare.library.vcu.edu/programs/child-welfarechildlabor/national-child-labor-committee/
- Perry Preschool Project. (1962). *Our history*. Retrieved from https://highscope.org/whowe-are/our-history/

Phillips, D. A., & Lowenstein, A. E. (2011). Early care, education, and child development. *Annual Review of Psychology* 62, 483-500. doi: 10.1146/annurev.psych.031809.130707. PMID: 20822436

Power to the Profession Task Force. (2020). *Unifying framework for the early childhood education profession*. Retrieved from http://powertotheprofession.org/

Prevent Child Abuse America. (2020). *Vital signs/ACEs report resources & talking points* [Web page]. Retrieved from https://preventchildabuse.org/vital-signs-acesreport-talking-points-other-resources/ Psychotherapy Networker. (2014, August 11). When is it trauma? Bessel Van der Kolk explains [Video file]. Retrieved from

https://www.youtube.com/watch?v=6vwNX0Qo1IA

- Raine Group. (2014). Early childhood mental health consultation protects and maximizes our national investment in early care and education. Phoenix, AZ: Southwest Human Development.
- Reid, K. (2019, October 21). *Child labor: Facts, FAQs, and how to help end it.* Retrieved from https://www.worldvision.org/child-protection-news-stories/child-labor-facts
- Research Connections. (2020). *Child care & early education glossary*. Retrieved from https://www.researchconnections.org/childcare/childcare-glossary
- Rhodes, H., & Huston, A. (2012). *Building the workforce our youngest children deserve and commentaries* (Social Policy Report 26). Retrieved from doi/pdf/10.1002/j.2379-3988.2012.tb00070.x
- Rosenblum, K. L., Dayton, C. J., & Muzik, M. (2019). Infant social development: Emerging competence in a relational context. In C. Zeanah (Ed.), *Handbook of infant mental health* (pp. 95-119). New York, NY: Guildford Press
- Rubin, H. J., & Rubin, I. (2005). *Qualitative interviewing: The art of hearing data*.Thousand Oaks, CA: Sage Publications.
- Saldana, J. (2011). *Fundamentals of qualitative research*. New York, NY: Oxford University Press.
- Saldana, J. (2016). *The coding manual for qualitative researchers* (3<sup>rd</sup> ed). Thousand Oaks, CA: Sage Publications.

Samels, M., & MacLowry, R. (2002). Timeline: Treatments for mental illness.
[Supplemental Material]. In R. MacLowry (Producer), & M. Samels, (Director).
Brilliant Madness. Retrieved from

http://www.shoppbs.pbs.org/wgbh/amex/nash/timeline/index.html

Scheeringa, M. S., & Haslett, N. (2010). The reliability and criterion validity of the Diagnostic Infant and Preschool Assessment: A new diagnostic instrument for young children. *Child Psychiatry & Human Development*, 41, 299-312. doi: 10.1007/s10578-009-0169-2

Schonert-Reichl, K. A., Kitil, M. J., & Hanson-Peterson, J. (2017). To reach the students, teach the teachers: A national scan of teacher preparation and social and emotional learning. A report prepared for the Collaborative for Academic, Social, and Emotional Learning (CASEL). Vancouver, B.C.: University of British Columbia.

- Schonfeld, D. J., Demaria, T., & Kumar, S. A. (2020). Supporting young children after crisis events. *Young Children*, 75(3), 8-15. Washington DC: National Association for the Education of Young Children.
- Sesame Street Workshop. (1969). *Our history*. Retrieved from https://www.sesameworkshop.org/who-we-are/our-history

Shahinfar, A., Fox, N. A., & Leavitt, L. A. (2000). Preschool children's exposure to violence: Relation of behavior problems to parent and child reports. *American Journal of Orthopsychiatry*, 70(1), 115–125. https://doi.org/10.1037/h0087690]

Shore, R. (1996). *Rethinking the brain: New insights into early development*. New York, NY: Families and Work Institute.

- Sim, M. P. Y., Belanger, J., Stancel-Piqtak, A., & Karoly, L. (2019). Starting strong teaching and learning international survey 2018 conceptual framework (OCED Education Working Papers, No. 197). https://doi.org/10.1787/106b1c42-en
- Sorrels, B. (2015). *Reaching and teaching children exposed to trauma*. Lewisville, NC: Gryphon House.
- Statman-Weil, K. (2015). Creating trauma-sensitive classrooms. *Young Children*, 70(2), 72-79. Retrieved from

https://www.naeyc.org/resources/pubs/yc/may2015/trauma-sensitive-classrooms

- Stone, J., & Bray, S. (2015). Trauma and young children: How the problem plays out. Advances in Early Education and Day Care (19), 177-211. doi:10.1108/S0270-4022120150000019012
- Substance Abuse and Mental Health Services Administration. (2020). Understanding child trauma. Retrieved from https://www.samhsa.gov/child-trauma/understanding-child-trauma
- Sutton, J., & Austin, Z. (2015). Qualitative research: Data collection, analysis, and management. *The Canadian Journal of Hospital Pharmacy*, 68(3), 226–231. https://doi.org/10.4212/cjhp.v68i3.1456
- The Abecedarian Project. (1972). *The Abecedarian project*. Retrieved from https://abc.fpg.unc.edu/abecedarian-project
- The Family Conservancy. (2021). *About us*. Retrieved from https://www.thefamilyconservancy.org/kansas-city-charity-for-children/

- The Learning Tree Institute. (2021, March 4). 2020 Kansas child care market analysis final report. Retrieved from http://www.dcf.ks.gov/services/ees/Documents/Child\_Care/Provider\_Market\_Rat e\_Study.pdf
- The White House. (2002). *Good start, grow smart: The Bush administration's early childhood initiative executive summary*. Retrieved from https://georgewbush whitehouse.archives.gov/infocus/earlychildhood/earlychildhood.html
- Tokarz, B. (2017). A place where all children can thrive Strategies for the traumainformed classroom. In *Exchange essentials: Trauma-informed practice*. ExchangePress.com
- Tracy, N. (2019, October 23). *The history of mental illness*. Retrieved from https://www.healthyplace.com/other-info/mental-illness-overview/the-history-ofmental-illness
- Tull, M. (2020). *The long-term impact of PTSD in Vietnam war veterans*. Retrieved from https://www.verywellmind.com/ptsd-from-the-vietnam-war-2797449
- University of Iowa Labor Center. (2004). *Child labor in U.S. history*. Retrieved from https://laborcenter.uiowa.edu/special-projects/child-labor-public-education-project
- U.S. Department of Health & Human Services, Administration for Children and Families, Administration on Children, Youth and Families, Children's Bureau. (2020). *Child maltreatment 2018.* Retrieved from https://www.acf.hhs.gov/cb/researchdata-technology /statistics-research/child-maltreatment

- Valenzuela, D., & Shrivastava, P. (2002). *Interview as a method for qualitative research* [Power Point presentation]. Retrieved from https://sdhrc.tbzmed.ac.ir
- Vandell, D. (2004). Early child care: The known and the unknown. *Merrill-Palmer Quarterly*, *50*(3), 387-414. doi:10.1353/mpq.2004.0027
- Van der Kolk, B. A. (2005). Developmental trauma disorder: Toward a rational diagnosis for children with complex trauma histories. *Psychiatric Annals*, 35(5), 401-408. doi:http://dx.doi.org.bakeru.idm.oclc.org/10.3928/00485713-20050501-06
- Van der Kolk, B. A. (2014). *The body keeps score: brain, mind, and body in the healing of trauma*. New York, NY: Penguin Books.

White, B. (1975). The first three years. Englewood Cliffs, NJ: Avon Books.

Workman, S. & Ullrich, R. (2017, February 13). Quality 101: Identifying the core components of a high-quality early childhood program. Retrieved from https://www.americanprogress.org/issues/earlychildhood/reports/2017/02/13/414939/quality-101-identifying-the-corecomponents-of-a-high-quality-early-childhood-program/

- Zaslow, M., & Martinez-Beck, I. (2006). *Critical Issues in Early Childhood Professional Development*. Baltimore, MD: Paul H. Brookes
- Zaslow, M., Tout, K., Halle, T., Vick Whittaker, J., & Lavelle, B. (2010). *Toward the identification of features of effective professional development for early childhood educators* [Literature Review].

http://www.ed.gov/about/offices/list/opepd/ppss/reports.html.

Zeanah, C., & Zeanah, P. (2019). The scope of infant mental health. In C. Zeanah (Ed.), *Handbook of infant mental health* (pp. 1-21). New York, NY: The Guilford Press.

- Zero to Three. (2005). *Helping young children succeed: Strategies to promote early childhood social and emotional development*. Retrieved from https://www.zerotothree.org/resources/136-helping-young-children-succeed-strategies-to-promote-early-childhood-social-and-emotional-development#downloads
- Zero to Three. (2016). *Infant-early childhood mental health*. Retrieved from https://www.zerotothree.org/resources/110-infant-early-childhood-mental-health
- Zero to Three. (2020). *State of babies' yearbook: 2020*. State of Babies webpage Retrieved from https://stateofbabies.org/
- Zero to Three. (2021, February 4). Infant and Early Childhood Mental Health (IECMH) Policy Series State Stories of IECMH Policy Innovation: Awareness, Leadership and Workforce Development. Session 2 [Webinar]. Retrieved from https://www.zerotothree.org/resources/3759-webinar-series-infant-and-earlychildhood-mental-health-policy
- Zero to Three Infant Mental Health Task Force Steering Committee. (2001). *Definition of infant mental health*. Washington, DC: National Center for Clinical Infant Programs. Retrieved from https://ztt.confex.com/ztt/btti14/webprogram/Handout/Session12328/IMH%20De finitions\_%20Brown.pdf

# Appendices

Appendix A: K.A.R. 28-4-428a. Education and Training Requirements.

### K.A.R. 28-4-428a. Education and Training Requirements.

### (a) Orientation.

(1) Each person shall, before applying for a license, complete an orientation program on the requirements for operating a preschool or a child care center. If the person is not an individual, the person shall designate an individual to meet this requirement. The orientation shall be provided by the county health department or the secretary's designee that serves the county in which the preschool or child care center will be located.

(2) Each licensee shall provide orientation to each program director not later than seven calendar days after the date of employment and before the program director is given sole responsibility for implementing and supervising the program.

(3) Each licensee shall ensure that orientation is completed by each staff member who will be counted in the staff-child ratio and by each volunteer who will be counted in the staff-child ratio. Each staff member and each volunteer shall complete the orientation within seven calendar days after the date of employment or volunteering. Each staff member shall complete the orientation before being given sole responsibility for the care and supervision of children.

(4) Each licensee shall ensure that the orientation for each program director, staff member, and volunteer is related to work duties and responsibilities and includes the following:

(A) Licensing regulations;

(B) the policies and practices of the preschool or child care center, including emergency procedures, behavior management, and discipline;

(C) the schedule of daily activities;

(D) care and supervision of children in care, including any special needs and known allergies;

(E) health and safety practices; and

(F) confidentiality.

(b) Health and safety training.

- (1) Each staff member who is counted in the staff-child ratio, each volunteer who is counted in the staff-child ratio, and each program director shall complete health and safety training either before employment or volunteering or not later than 30 calendar days after the date of employment or volunteering. Each staff member shall complete the training before being given sole responsibility for the care and supervision of children.
- (2) The health and safety training shall be approved by the secretary and shall include the following subject areas:

(A) recognizing the signs of child abuse or neglect, including prevention of shaken baby syndrome and abusive head trauma, and the reporting of suspected child abuse or neglect;

(B) basic child development, including supervision of children;
(C) safe sleep practices and sudden infant death syndrome if the individual will be caring for children under 12 months of age;
(D) prevention and control of infectious diseases, including immunizations;
(E) prevention of and response to emergencies due to food and allergic reactions;
(F) building and premises safety, including identification of and protection from hazards that could cause bodily injury, including electrical hazards, bodies of water, and vehicular traffic;
(G) emergency preparedness and response planning for emergencies resulting from a natural disaster or a human-caused event, including violence at a facility;
(H) handling and storage of hazardous materials and the appropriate disposal of bio-contaminants, including blood and other bodily fluids or waste; and
(I) precautions when transporting children, if transportation is provided.

(3) Each staff member counted in the staff-child ratio, each volunteer counted in the staff-child ratio, and each program director who was employed at the facility before July 1, 2017 and who has completed the training in the subject areas specified in paragraphs (b)(2)(A), (B), and (C) shall be exempt from training in the subject areas specified in paragraphs (b)(2)(D) through (I).

(c) Pediatric first aid and cardiopulmonary resuscitation (CPR) certifications.

(1) Each staff member counted in the staff-child ratio, each volunteer counted in the staff-child ratio, and each program director shall obtain certification in pediatric first aid and in pediatric CPR as specified in this subsection either before the date of employment or volunteering or not later than 30 calendar days after the date of employment or volunteering.

(2) Each individual who is required to obtain the certifications shall maintain current certifications.

(3) Each licensee shall ensure that, for each unit in a preschool or child care center, at least one staff member or volunteer counted in the staff-child ratio who has current certification in pediatric first aid and current certification in pediatric CPR is present at all times.

(d) Medication administration training. Each program director and each staff member designated to administer medications shall complete the training in medication administration as specified in this subsection.

(1) The training shall be approved by the secretary.

(2) Each program director and each staff member designated to administer medications who was employed at the facility before July 1, 2017 shall complete the training not later than December 31, 2017. The program director or the staff member

designated to administer medications shall not administer medications afterDecember 31, 2017 unless the individual has completed the training.(3) Each program director and each staff member designated to administermedications who is employed at the facility on or after July 1, 2017 shall completethe training before administering medication to any child.

(e) Education requirements. Each program director shall be a high school graduate or the equivalent. For each unit in a preschool or child care center, there shall be present at all times at least one staff member who has a high school diploma or the equivalent, as required in K.A.R. 28-4-429.

(f) Annual in-service training requirements.

(1) For purposes of this subsection, "licensure year" shall mean the period beginning on the effective date and ending on the expiration date of a license.

(2) In each licensure year, each program director shall assess the training needs of each staff member and each volunteer and shall provide or arrange for annual inservice training as needed.

(3) In each licensure year, each program director shall complete in-service training as follows:

(C) for each licensure year ending during the 2019 calendar year, 12 clock-hours; and

(D) for each licensure year ending during the 2020 calendar year, and for each subsequent licensure year, 16 clock-hours.

(4) In each licensure year, each staff member counted in the staff-child ratio and each volunteer counted in the staff-child ratio shall complete in-service training as follows, based on the staff member's or volunteer's job responsibilities and the training needs identified by the program director:

(A) for each licensure year ending during the 2017 calendar year, 10 clockhours;

(B) for each licensure year ending during the 2018 calendar year, 10 clockhours;

(C) for each licensure year ending during the 2019 calendar year, 12 clockhours; and

(D) for each licensure year ending during the 2020 calendar year, and for each subsequent licensure year, 16 clock-hours.

(5) The training shall be approved by the secretary.

(g) Documentation. Each licensee shall ensure that documentation of all orientation, training, certifications, and education requirements is kept in each individual's file in the preschool or child care center.

## Appendix B: Baker University IRB Approval



Baker University Institutional Review Board

June 9<sup>th</sup>, 2021

Dear Mickie Hodapp and Tes Mehring,

The Baker University IRB has reviewed your project application and approved this project under Expedited Status Review. As described, the project complies with all the requirements and policies established by the University for protection of human subjects in research. Unless renewed, approval lapses one year after approval date.

Please be aware of the following:

- 1. Any significant change in the research protocol as described should be reviewed by this Committee prior to altering the project.
- 2. Notify the IRB about any new investigators not named in original application.
- 3. When signed consent documents are required, the primary investigator must retain the signed consent documents of the research activity.
- 4. If this is a funded project, keep a copy of this approval letter with your proposal/grant file.
- 5. If the results of the research are used to prepare papers for publication or oral presentation at professional conferences, manuscripts or abstracts are requested for IRB as part of the project record.
- 6. If this project is not completed within a year, you must renew IRB approval.

If you have any questions, please contact me at npoell@bakeru.edu or 785.594.4582.

Sincerely,

Nathan D. Par

Nathan Poell, MLS Chair, Baker University IRB

Baker University IRB Committee Sara Crump, PhD Nick Harris, MS Christa Manton, PhD Susan Rogers, PhD

## Appendix C: Facebook and Professional Messaging Board Posting

### Participate in a research study on Early Childhood Teachers Perceptions of Training to ECCE staff who Work with Young Children Who Have Experienced Trauma Traumatic events may include a serious illness that requires significant medical intervention; the loss or life-threatening illness of a close family member; witnessing domestic or community violence or threats of terrorism; homelessness;

and neglect or physical or sexual abuse.

### Seeking Volunteers for Individual Interviews

to provide feedback about your early childhood experience and preparation to work with young children who have experienced trauma

**Who should consider participating?** Eligible early childhood education staff who are currently working with young children who have experienced trauma in a center-based program in Johnson County Kansas who have two or more years of experience working directly with young children

### Date & Time will be scheduled based upon your availability and convenience.

Contact Mickie Hodapp by phone (816-898-0338) or email (michelelhodapp@stu.bakeru.edu) to sign-up or get more information.

Your thoughts and experiences are important in determining the training and support early care and education staff need when working with young children having experienced trauma.

# Appendix D: Invitation to Participate

Ladies and Gentlemen:

I am seeking preschool and child care teachers in Johnson County Kansas to voluntarily participate in a doctoral research study. The study explores the perceptions of early childhood teachers about how trauma impacts young children in the classroom and the adequacy of their preparation to teach young children who have experienced trauma.

I have worked in the early childhood field for over 20 years as an infant, toddler, preschool teacher and center director in a variety of program settings. I am currently undertaking qualitative research to complete the doctoral requirements in Educational Leadership at Baker University.

Your participation in this study is completely voluntary. There are no known or anticipated risks or discomfort associated with your participation in the study. You will not receive any compensation for participation. You may withdraw from the study at any time, or refuse to answer any individual question, or end participation at any time during the interview. Your written consent to participate is required before the interview begins. The consent form is attached to this letter.

Interviews will be conducted August 25 – September 3. One-on-one interviews will be conducted using Zoom video conferencing. It will require approximately 45 to 60 minutes of your time, scheduled at your convenience. The interview includes the 19 questions. Interviews will be audio-recorded. Confidentiality and anonymity will be maintained. Your audio-video recording and transcript will be assigned an anonymous code (e.g. Participant 1, Participant 2, etc.) known only to me. Once your interview has been transcribed, I will send you the transcript of the interview for you to review its accuracy.

The qualifications of the participant include:

- Two or more years of experience working with young children,
- Employed an child care or preschool program in Johnson County Kansas
- Employed in a center-based ECCE classroom sometime during January 2020 through July 2021.
- May have had contact with one or more young children in the center-based ECCE classroom

who had experienced trauma.

Any questions don't hesitate to call or text me, 816-898-0338 or email: michelelhodapp@stu.bakeru.edu or mhodapp.hgk@gmail.com. Thank you for your consideration.

Wishing you well,

mickie Hodopp

Michele "Mickie" Hodapp Doctoral Candidate, Baker University

Phone: 816- 898-0338 Email: michlelelhodapp@stu.bakeru.edu mhodapp.hkg@gmail.com

# Appendix E: Confirmation to Participate

#### **Confirmation to Participate**

[Date]

Dear [Early Education Teacher's Name]:

I am delighted you have agreed to participate in my dissertation research study. This confirmation letter contains information I shared with you during our phone conversation. I am a doctoral student in the Graduate School of Education at Baker University in Kansas. I am seeking early childhood care and education (ECCE) teachers for a qualitative study regarding the perceptions of ECCE teachers about how childhood trauma impacts young children, information and skills ECCE teachers need to effectively work with young children who have experienced trauma, the training ECCE teachers have received to work with young children who have experienced trauma, and challenges ECCE teachers face regarding the availability and accessibility of professional development to aid them in working with young children who have experienced trauma.

Your participation in this study will involve approximately 45 to 60 minutes of your time, scheduled at your convenience. One-on-one interviews will be conducted using Zoom video conferencing. The interview includes 20 questions (provided below). Interviews will be audio-videorecorded.

Participation in this study is completely voluntary. There are no known or anticipated risks or discomfort associated with your participation in the study. You will not receive any compensation for participation. You may withdraw from the study at any time, or refuse to answer any individual question, or end participation at any time during the interview. All of your responses will remain confidential. Your audio-video recording and transcript will be assigned an anonymous code (e.g. Participant 1, Participant 2, etc.) known only to me. Once your interview has been transcribed, I will send you the transcript of the interview for you to review its accuracy.

The Consent Form and interview questions are included with this email. The consent form includes an agreement to participate, permission to audio-video record the interview and

171

allows the researcher to take notes during the interview. If you are willing to participate, please return the signed Consent Form to me at michelelhodapp@stu.bakeru.edu as soon as possible.

We have tentatively scheduled the zoom interview for <u>(date and time)</u> Upon receipt of the Consent Form I will send an email confirming the time of our interview and the zoom link for the interview.

Although this study will not benefit you personally, I hope that the results will generate recommendations and resources for ECCE teachers to gain access to professional development opportunities that will aid and support them in working with young children who have experienced trauma.

Please feel free to contact me at 816-898-0338 or michelelhodapp@stu.bakeru.edu or my major advisor, Dr. Tes Mehring at tmehring@bakeru.edu if you have any questions about the study or interview process.

Many thanks for your time and willingness to participate in this study.

Wishing you well,

Michele "Mickie Hodapp Ed.D. Candidate Baker University 4016 W. 100 Place Overland Park, KS 66207-3737 816-898-0338 michelelhodapp@stu.bakeru.edu Dr. Tes Mehring Major Advisor Baker University 7301 College Boulevard, Suite 120 Overland Park, KS 66210 913-344-1236 tmehring@bakeru.edu

## **Appendix F: Consent Form**

### **Consent Form**

Please carefully consider the following information before deciding whether to participate in this study.

**Purpose of This Study:** The purpose of this qualitative study is to examine the perceptions of early childhood teachers about how trauma impacts young children in early childhood centered-based programs and the training early childhood teachers receive to work with young children who have experienced trauma.

**Interview Questions:** The interview will include the questions provided at the end of the Consent Form.

**Interview Process:** The interview will take approximately 45-60 minutes. The researcher will ask each participant the 20 interview questions provided at the end of the Consent Form. Additional questions may be asked to clarify responses. Each interview will be audio-video recorded and transcribed. Once the interview transcription is completed, the researcher will send the transcript to the interviewee for review of accuracy and additional comments, additions, deletions, or corrections.

**Confidentiality:** All responses to interview questions for this study will be confidential. A neutral code (e.g. Participant 1, Participant 2, etc.) will be assigned to the interview recording and transcript to maintain anonymity. Each interview will be audio-video recorded, transcribed, and saved to a secure database accessible only to the researcher. Your name will not be associated in any way with the research findings. Recordings and transcripts will be kept on a secure jump drive accessible only to the researcher and will be destroyed after five years. **Permission to Audio-Video Record:** The interview will be audio-video recorded to facilitate accuracy in creating a transcription of the interview. Notes will be taken during the interview. Your consent to participate in the interview also indicates consent to audio-video record the interview and the interviewer to take notes.

**Voluntary Participation:** Your participation in this study is completely voluntary. You may choose to not answer any question in this study at any time. If you feel uncomfortable answering any of the questions, we will skip them. You may withdraw from the study at any time without penalty. You may withdraw by informing the experimenter that you no longer wish to participate (no questions will be asked). Your audio-video recording will be destroyed if you decide to withdraw from the study. If you would like to withdraw prior to the interview or after your consent form has been submitted, please contact the researcher at michelelhodapp@stu.bakeru.edu..

**Risks and Benefits:** No risks or discomfort is associated with participation in the study.

There is no penalty for withdrawing from the study. There is no compensation or benefits for participating in this study.

### For Questions Regarding This Study, Contact:

Principle Investigator: Michele L Hodapp 4016 W. 100 Place Overland Park, KS 66207-3737 816-898-0338 michelelhodapp@stu.bakeru.edu Academic Advisor: Tes Mehring, Ph.D. Graduate School of Education Baker University 7301 College Boulevard, Suite 120 Overland Park, KS 66210 913-344-1236 tmehring@bakeru.ed

### Agreement:

The nature and purpose of this research have been sufficiently explained and I agree to participate in this study. I understand that I am free to withdraw at any time without

incurring any penalty. I understand the interview will be audio-video recorded and the researcher may be taking notes.

Date:

Name (Printed):\_\_\_\_\_

Please return the signed consent form via email to Mickie Hodapp at:

Michelelhodapp@stu.bakeru.edu

\_\_\_\_

### **Interview Questions**

**1.** If you are willing please identify your age.

**2.** What is your highest level of education?

**3.** How long have you worked in the early childhood care and education field?

**4.** During that time in ECCE what positions have you held?

5. How long have you been employed in your current employment role?

**6.** What is the licensed capacity of early childhood care and education program you are currently working in?

- a) 60 or fewer children
- b) 61-99 children
- c) 100 159 children

7. What is the age range of the children in your classroom?

**8.** Please describe other classrooms where you have worked and the age range of the children in those classrooms.

**9.** This research study looks at behaviors of young children who may have experienced trauma and the teachers' perceptions of being adequately prepared to work with those children who may have experienced trauma. How do you define trauma?

**10**. What characteristics of a child who has experienced trauma in your classroom have you observed?

**11.** How have the behaviors of the traumatized child impacted other children in your classroom or the early education program?

**12.** How has the behaviors of the traumatized child impacted your ability to manage classroom experiences and learning activities?

13. What are the greatest challenges that face early childhood teachers in the coming year in working with young children who have experienced trauma?14. As you reflect on your experiences working with young children who express

challenging or problem behaviors that may be associated with trauma whatinformation do you perceive you need to provide quality early care and education?15. As you reflect on your experiences working with young children who expresschallenging or problem behaviors that may be associated with trauma what skills do

you perceive you need to provide quality early care and education?

16. Describe how you have gained information about social-emotional development of young children and characteristics of young children who have experienced trauma.

**17.** Describe how you have gained information about instructional strategies for effectively teaching young children who have experienced trauma.

**18.** How has the professional development you have participated in been delivered in-person, online, or some other format?

**19.** What challenges do you face in accessing professional development that would assist you in effectively teaching young children who have experienced trauma?

**20.** Do you have any additional thoughts or concerns regarding early childhood teachers working with young children who have experienced trauma?